ELECTROKINETIC PROCESS IN SOIL REMEDIATION FOR LEAD REMOVAL

NG YEE SERN

THESIS SUBMITTED IN FULFILMENT OF THE REQUIREMENTS FOR THE DEGREE OF DOCTOR OF PHILOSOPHY

FACULTY OF ENGINEERING UNIVERSITY OF MALAYA KUALA LUMPUR

2016

UNIVERSITI MALAYA

Original Literary Work Declaration

Name of Candidate:	NG YEE SERN
Matric No:	KHA120057
Name of Degree:	KHA-DOCTOR OF PHILOSOPHY
Title of Thesis ("this work"):	ELECTROKINETIC PROCESS IN SOIL REMEDIATION FOR LEAD REMOVAL
Field of Study:	CHEMICAL ENGINEERING – SEPARATION PROCESSES

I do solemnly and sincerely declare that:

- (1) I am the sole author/writer of this Work;
- (2) This Work is original;
- (3) Any use of any work in which copyright exists was done by way of fair dealing and for permitted purposes and any excerpt or extract from, or reference to or reproduction of any copyright work has been disclosed expressly and sufficiently and the title of the Work and its authorship have been acknowledged in this Work;
- (4) I do not have any actual knowledge nor do I ought reasonably to know that the making of this work constitutes an infringement of any copyright work;
- (5) I hereby assign all and every rights in the copyright to this Work to the University of Malaya ("UM"), who henceforth shall be owner of the copyright in this Work and that any reproduction or use in any form or by any means whatsoever is prohibited without the written consent of UM having been first had and obtained;
- (6) I am fully aware that if in the course of making this Work I have infringed any copyright whether intentionally or otherwise, I may be subject to legal action or any other action as may be determined by UM.

Candidate's Signature:

Date

Date

Subscribed and solemnly declared before,

Witness's Signature:

Name:

Designation:

ABSTRACT

Soil washing is one of the popular soil remediation methods which is widely studied for treating heavy metal contaminated land. However, this method suffers from high wash solution consumption, which eventually increases treatment cost. In order to solve this problem, a study on the incorporation of electrokinetic process into soil washing as soil remediation method was carried out, as electrokinetic process can transport metal ions using electricity in the absence of hydraulic gradient. In this study, two-stage electrokinetic washing was introduced as a novel soil remediation method in treating lead (Pb) contaminated soil at low wash solution consumption. This process consists of: i) initial soil washing to provide desorption condition in the soil, and ii) electrokinetic process to transport desorbed Pb from the soil. The study revealed that two-stage electrokinetic washing could enhance Pb removal efficiency by 4.98-20.45% in comparison to normal soil washing. Among the wash solutions, 0.1M citric acid emerged as the best wash solution as it not only yielded high removal efficiency at low power consumption compared to 0.01M NaNO₃, 0.1M HNO₃ and 0.01M EDTA but also maintained a stable system, low solution: soil ratio and low effluent generation.

A further study on the effect of operating parameters using citric acid as the wash solution revealed that the increase in electric potential difference and wash solution concentration enhanced the removal efficiency and the interaction between these two parameters was significantly positive whereby low pH and high current density were the most important criteria in the removal process. However, unfavourable high effluent generation and power consumption were also observed under these conditions. An optimisation study on these parameters showed that an optimum removal efficiency of 84.14% with negligible extra effluent generation and low power consumption of 2.27kWh/kg Pb removed could be achieved under 7.58V and 0.057M citric acid concentration. The study proved that two-stage electrokinetic washing

process could enhance the soil remediation efficiency by $\approx 16\%$ than normal soil washing under similar low consumption of wash solution at <0.8mL:1g soil and operating conditions.

Other than incorporating into soil washing, the ability of electrokinetic process as soil pretreatment method in contaminated soil volume reduction was also investigated for single Pb contaminated soil and Pb/Cr co-contaminated soil. The results showed that electrokinetic process could provide soil volume reduction by concentrating Pb and Cr into smaller soil volume in both types of soils via electromigration without hydraulic flow. The study also suggested that the performance of electrokinetic process was strongly dependent on the types of wetting agents, types of contaminants and soil conditions, whereby 0.1M citric acid was suitable for single Pb contaminated soil while 0.1M EDTA showed better performance in treating Pb/Cr cocontaminated soil. Further study on the application of approaching electrode technique in electrokinetic process revealed that this technique did not enhance the electromigration significantly in the present study. Nevertheless, approaching electrode was found to reduce the power consumption by 18% to 42% for single Pb contaminated soil and 22.5% for co-contaminated soil.

ABSTRAK

Pencucian tanah merupakan salah satu teknik rawatan tanah popular di mana ia telah dikaji secara meluas dalam rawatan tanah yang dicemari oleh logam berat. Akan tetapi, kaedah ini mengalami penggunaan larutan pencucian yang tinggi dan ini menyebabkan peningkatan kos rawatan. Untuk menyelesaikan masalah ini, satu kajian dalam penggabungan proses elektrokinetik dalam pencucian tanah telah dijalankan atas sebab keupayaannya dalam pengangkutan ion logam dengan menggunakan daya elektrik tanpa kewujudan cerun hidraulik. Dalam kajian ini, pencucian elektrokinetik dua peringkat telah digunakan sebagai teknik novel dalam rawatan tanah yang dicemari oleh plumbum (Pb) dengan menggunakan larutan pencucian yang rendah. Proses ini terdiri daripada: i) pencucian tanah awal untuk memberi keadaan desorption dalam tanah, dan ii) proses elektrokinetik untuk mengangkut Pb dari tanah. Kajian menunjukkan bahawa pencucian elektrokinetik dua peringkat dapat meningkatkan kecekapan penyingkiran Pb sebanyak 4.98-20.45% berbanding dengan proses pencucian tanah biasa. 0.1M Asid sitrik muncul sebagai larutan pencucian terbaik kerana ia tidak sahaja menunjukkan kecekapan penyingkiran yang tinggi dan penggunakan kuasa yang rendah berbanding dengan 0.01M NaNO₃, 0.1M HNO₃, dan 0.01M EDTA, malah dapat mengekalkan kestabilan sistem, nisbah larutan: tanah yang kecil dan penghasilan efluen yang kecil.

Kajian selanjutnya pada kesan parameter-parameter operasi dengan menggunakan asid sitrik sebagai larutan pencucian menunjukkan bahawa peningkatan perbezaan potensi elektrik dan konsentrasi larutan pencucian dapat meningkatkan kecekapan penyingkiran dan interaksi parameter-parameter tersebut adalah positif dan ketara di mana pH yang rendah dan arus yang tinggi adalah kriteria-kriteria penting dalam proses penyingkiran. Tetapi, ini juga meningkatan penjanaan efluen dan pengunaan kuasa. Pengoptimuman parameter-parameter tersebut menunjukkan bahawa kecekapan penyingkiran optimum pada 84.14% dengan ketiadaan efluen ekstra dan penggunaan kuasa yang kecil pada 2.27kWh/kg Pb disingkir dapat dicapai pada 7.58V dan 0.057M konsentrasi acid sitrik. Kajian membuktikan bahawa proses pencucian elektrokinetik dua peringkat dapat meningkatkan kecekapan penyingkiran sebanyak ≈16% berbanding dengan pencucian tanah biasa dalam keadaan optimum sistem dan penggunaan jumlah larutan pencucian yang sama pada nisbah <0.8mL:1g soil.

Selain pengabungan dalam proses pencucian tanah, keupayaan proses elektrokinetik sebagai kaedah pra-rawatan tanah dalam pengurangan isipadu tanah tercemar juga dikaji untuk tanah yang dicemar oleh Pb dan Pb/Cr. Kajian menunjukkan proses elektrokinetik dapat mengurangkan isipadu tanah tercemar dengan memadatkan kepekatan Pb dan Cr dalam isipadu tanah yang lebih kecil dengan proses elektromigrasi tanpa aliran hidraulik. Kajian juga mencadangkan prestasi proses elektrokinetik banyak bergantung kepada jenis larutan pencucian, bahan cemar serta keadaan tanah, di mana 0.1M asid sitrik lebih sesuai untuk tanah tercemar oleh Pb tunggal manakala 0.1M EDTA memberi prestasi yang lebih bagus dalam tanah yang dicemar oleh Pb dan Cr. Kajian selanjutnya dalam aplikasi teknik penghampiran elektrod dalam proses elektrokinetik menunjukkan teknik ini tidak meningkatkan elektromigrasi secara ketara dalam kajian ini. Akan tetapi, teknik penghampiran elektrod dapat mengurangkan penggunaan kuasa sebanyak 18% ke 42% untuk tanah tercemar oleh Pb tunggal dan 22.5% untuk tanah tercemar oleh Pb dan Cr.

ACKNOWLEDGMENTS

Firstly, I would like to express my deepest gratitude to both of my supervisors, Prof. Ir. Dr. Mohd Ali Hashim and Prof. Dr. Bhaskar Sen Gupta for their helps and supervisions during my candidature period. Their patient guidance in this study is much appreciated.

Besides that, I would like to thank University of Malaya for their financial supports of PPP research grant during my study. My grateful thank is also dedicated to my colleague, Dr. Soumyadeep Mukhopadhyay for his help especially in soil characterisation study.

Lastly, I would like to thank my family and my beloved wife for their courage and support during my study.

TABLE OF CONTENTS

Original Literary Work Declaration	ii
Abstract	iii
Abstrak	v
Acknowledgments	vii
Table of Contents	viii
List of Figures	xiv
List of Tables	xviii
List of Symbols and Abbreviations	xix
List of Appendices	xxii

CHAPTER 1: INTRODUCTION	1
1.1 Background	1
1.2 Research objectives	3
1.3 Scope of study	4
1.4 Thesis outline	5

CHAPTEF	R 2: LITERATURE REVIEW	7
2.1 Lead		7
2.1.1	Production and application of lead	8
2.1.2	Toxicological effect of lead on human	8
2.1.3	Soil contamination by lead	10
2.1.4.	Environmental regulations for lead concentration in soil	11
2.2 Popular	methods for treating heavy metals contaminated soil	12
2.2.1	Soil washing	12
2.2.2	Stabilisation and solidification	15

2.3 Electrokinetic soil remediation		16	
2.3.1	Process description		16
2.3.2	Factors affec	ting remediation efficiency using electrokinetic	
	process		20
	2.3.2.1	Electric potential difference and current	20
	2.3.2.2	pH conditioning	22
	2.3.2.3	Complexing agents	25
	2.3.2.4	Acidic based agents	28
	2.3.2.5	Soil texture	30
	2.3.2.6	Contaminant species	32
2.3.3	Incorporation	n of electrokinetic process with soil washing/	
	flushing		33
	2.3.3.1	Electrokinetic flushing	33
	2.3.3.2	Vertical electrokinetic flushing	34
	2.3.3.3	Electrokinetic washing	37
2.4 Summary of literature review		40	
CHAPTER 3	: METHODO	DLOGY	42
3.1 Materials.			42

3.1 Materia	lls	42
3.1.1	Chemicals	42
3.1.2	Soil	43
3.2 Equipm	nent and apparatus	43
3.2.1	Electrokinetic washing system	43
3.2.2	Electrokinetic soil pretreatment system	45
3.2.3	Analytical instruments	46
3.3 Experir	nental procedures	46

	3.3.1	Degree of so	bil contamination	46
	3.3.2	Soil spiking		
	3.3.3	Screening for	or wash solutions	48
	3.3.4	Two-stage e	electrokinetic washing	49
	3.3.5	Electrokinet	tic soil pretreatment process	51
	3.3.6	Response su	Irface methodology	52
3.4	Analytic	cal methods		53
	3.4.1	Electric curr	rent	54
	3.4.2	Soil propert	ies	54
		3.4.2.1	Soil pH	54
		3.4.2.2	Pb concentration in soil	55
		3.4.2.3	Cr concentration in soil	56
	3.4.3	Pb depositio	n efficiency	57
	3.4.4	Aqueous pro	perties	58
3.5	3.5 Precautious steps for chemicals handling		58	
3.6	3.6 Minimisation of error		59	

3.5 Precaut	tious steps for chemicals handling	58
3.6 Minim	isation of error	59
CHAPTE	R 4: RESULTS AND DISCUSSION	60
4.1 Study of	outlines	60
4.2 Soil ch	aracterisation	61
4.2.1	Soil properties	61
4.2.2	Degree of contamination and the mechanisms	63
4.3 Screen	ing of wash solutions	66
4.4 Two-st	age electrokinetic washing	68
4.4.1	Introduction	68
4.4.2	Efficiency comparison	69

4.4.3	Effect of wa	sh solution on two-stage electrokinetic washing	70
	4.4.3.1	Removal efficiency	70
	4.4.3.2	Effluent generation	75
	4.4.3.3	System stability – anode chamber	78
	4.4.3.4	System stability – cathode chamber	80
	4.4.3.5	Power consumption	85
	4.4.3.6	Selection of best wash solution	86
4.4.4	Effect of ope	erating parameters	88
	4.4.4.1	ANOVA analysis	89
	4.4.4.2	Removal efficiency	94
	4.4.4.3	Effluent generation	97
	4.4.4.4	Power consumption	100
	4.4.4.5	Electric current density and pH in wash solution	
		chambers and soil	101
	4.4.4.6	Pb deposition efficiency	106
	4.4.4.7	Optimisation study	108
4.4.5	Performance	e of two-stage electrokinetic washing under	
	optimum co	nditions	109
	4.4.5.1	Electrokinetic properties	110
	4.4.5.2	Removal efficiency	111
	4.4.5.3	Deposition efficiency	115
	4.4.5.4	Possible transport mechanisms	116
4.4.6	Summary a	nd limitations	119
4.5 Applicat	tion of electro	kinetic process as soil pretreatment method	121
4.5.1	Introduction	1	121
4.5.2	Approachin	g electrode assisted electrokinetic process	122

4.5.3	Applicat	ion of electrokinetic process in treating single	
	contami	nated soil	124
	4.5.3.1	Effect of wetting agents on electric current and soil	
		рН	124
	4.5.3.2	Effect of wetting agents on Pb migration	127
	4.5.3.3	Effect of approaching electrode on electric current	
		and soil pH	129
	4.5.3.4	Effect of approaching electrode on Pb migration	132
	4.5.3.5	Removal efficiency and power consumption	134
	4.5.3.6	Summary of remediation of single contaminated soil.	135
4.5.4	Applica	tion of electrokinetic process in treating co- contaminated	
	soil		136
	4.5.4.1	Effect of wetting agents on electric current and soil	
		рН	136
	4.5.4.2	Effect of wetting agents on Pb migration	139
	4.5.4.3	Effect of wetting agents on Cr migration	141
	4.5.4.4	Effect of approaching electrode on electric current	
		and soil pH	145
	4.5.4.5	Effect of approaching electrode on Pb and Cr	
		migrations	147
	4.5.4.6	Removal efficiency and power consumption	150
	4.5.4.7	Summary of remediation of co-contaminated soil	151
4.5.5	5 Summa	ry	152

CHAPTER 5: CONCLUSIONS AND RECOMMENDATIONS	153
5.1 Conclusions	153

5.1.1	Application of two-stage electrokinetic washing	153
5.1.2	Application of electrokinetic soil pretreatment process	156
5.2 Recomm	endations for future studies	158
REFEREN	CES	160
LIST OF P	UBLICATIONS AND PAPER PRESENTED	179
ADDENIDIC	CES	186
AFFENDIC	LS	180

LIST OF FIGURES

Figure 2.1	Electrokinetic process in soil treatment	17
Figure 2.2	Schematic diagram for electroosmosis process	18
Figure 2.3	Electrokinetic soil flushing equipment (Kim et al., 2008b)	33
Figure 2.4	Vertical Electrokinetic Soil Washing: (a) Downward configuration (Kim <i>et al.</i> , 2009c); (b) Upward configuration (UESR)	35
Figure 2.5	Example of electrokinetic washing equipment (Kim et al., 2012b).	37
Figure 3.1	Experimental setup for electrokinetic washing system: (i) direct current supply, (ii) wash solution reservoir, (iii) peristaltic pump, (iv) anode chamber, (v) soil column, (vi) cathode chamber, (vii) overflow tube/effluent collector	44
Figure 3.2	Connection of graphite electrode into the wash solution chamber	44
Figure 3.3	Experimental setup for soil pretreatment system using electrokinetic process	45
Figure 4.1	The outline for Chapter 4: Results and Discussion	60
Figure 4.2	FTIR spectrum for the uncontaminated soil	62
Figure 4.3	XRD spectrum for the uncontaminated soil	62
Figure 4.4	Change in Pb concentration in soil with time at solution pH 5 and the curve fitting using pseudo second order kinetic model	64
Figure 4.5	Desorption efficiency of Pb from soil using different types of wash solutions	66
Figure 4.6	Schematic diagram of two-stage electrokinetic washing process	69
Figure 4.7	Comparison on Pb removal efficiency between soil washing and two-stage electrokinetic washing	69
Figure 4.8	Properties of two-stage electrokinetic washing process using different types of wash solutions: a) soil pH; b) current density across the soil	71
Figure 4.9	Normalised Pb concentration at different soil sections after the experiments using different types of wash solutions: a) 0.01 M NaNO ₃ , b) 0.1 M HNO ₃ , c) 0.1 M citric acid, d) 0.01 M EDTA	74

Figure 4.10	Volume of effluent generated under different types of wash solutions	76
Figure 4.11	pH of wash solutions in anode chamber at different time	78
Figure 4.12	Physical appearance of graphite anode in 0.1 M HNO ₃ test: (a) before the experiment; (b) after the experiment; (c) carbon particles collected in anode chamber	79
Figure 4.13	pH of wash solutions in cathode chamber at different time	80
Figure 4.14	Physical appearance of cathode and deposits after the experiments: a) 0.01 M NaNO ₃ ; b) 0.1 M HNO ₃ ; c) 0.1 M citric acid; d) 0.01 M EDTA	82
Figure 4.15	SEM images for the soil samples after the experiments of two- stage electrokinetic washing under different types of wash solutions	84
Figure 4.16	SEM-EDX analysis for the deposits obtained after two-stage electrokinetic washing using 0.1 M citric acid	85
Figure 4.17	Normal probability plot for: a) removal efficiency; b) effluent generated; c) power consumption	92
Figure 4.18	Predicted results vs actual results plot for: a) removal efficiency; b) effluent generated; c) power consumption	93
Figure 4.19	Interaction effect of parameters on Pb removal efficiency: a) electric potential difference and wash solution concentration (initial Pb concentration: 700mg/kg); b) electric potential difference and initial Pb concentration (wash solution concentration: 0.0505M); c) wash solution concentration and initial Pb concentration (electric potential difference: 18.75V)	96
Figure 4.20	Relationship between effluent generated with electric current density	98
Figure 4.21	Interaction effect of parameters on effluent generation: a) electric potential difference and wash solution concentration (initial Pb concentration = 700 mg/kg); b) electric potential difference and initial Pb concentration (wash solution concentration = 0.0505 M).	99
Figure 4.22	Interaction effect of parameters on power consumption: a) electric potential difference and wash solution concentration (initial Pb concentration: 700mg/kg); b) electric potential difference and initial Pb concentration (wash solution concentration: 0.0505M)	100

Figure 4.23	Average current density of the system at different operating conditions: a) electric potential difference; b) wash solution concentration; c) initial Pb concentration	103
Figure 4.24	pH in anode and cathode chambers after 24 hours experiments	104
Figure 4.25	Soil pH at different soil sections after 24 hour experiments (Wash solution concentration: blue: 0.001M; black: 0.0505M; red: 0.1M)	105
Figure 4.26	Physical appearance of deposit on cathode after the experiments	106
Figure 4.27	a) Deposition efficiency for Pb on the cathode at different current density; b) Mass of Pb electrodeposited at different initial Pb concentration (Wash solution: citric acid)	107
Figure 4.28	a) Change of current density with time during the experiment; b) Change of pH in wash solution chambers with time during the experiment; c) Change in soil pH during the experiment	110
Figure 4.29	Effect of time on two-stage electrokinetic washing at 7.58V and 0.057M citric acid concentration: a) removal efficiency; b) normalised Pb concentration at different soil sections (Initial Pb concentration, 1000mg/kg)	112
Figure 4.30	Average normalised Pb concentration in the soil at different treatment time during electrokinetic process stage	114
Figure 4.31	Deposition efficiency of Pb in cathode chamber at different experimental duration at optimum conditions	115
Figure 4.32	General transport mechanisms in a two-stage electrokinetic washing process: a) initial washing; b) electrokinetic process	116
Figure 4.33	Schematic diagram for a) Pb desorption and b) Pb transport in two-stage electrokinetic washing when citric acid is applied as the wash solution	117
Figure 4.34	Schematic diagram for a) approaching anode, and b) approaching cathode (Shen <i>et al.</i> , 2007; Shen <i>et al.</i> , 2009)	123
Figure 4.35	Effect of wetting agents on: a) electric current, and b) soil pH at different soil sections.	125
Figure 4.36	Normalised Pb concentration at different soil sections after the experiments	127

Figure 4.37	Effect of approaching electrode technique on: a) electric current across the soil, and b) soil pH in electrokinetic process under different types of wetting agents (AA: approaching anode; AC: approaching cathode)	129
Figure 4.38	Normalised Pb concentration at different soil sections after the experiments for both fixed electrode tests and approaching electrode tests (AA: approaching anode; AC: approaching cathode).	132
Figure 4.39	Removal efficiency in the pretreated region and power consumption of the system under different types of wetting agents	134
Figure 4.40	a) Electric current across the soil, b) soil pH at different soil sections after the experiments	138
Figure 4.41	Normalised Pb concentration at different soil sections after the experiments.	139
Figure 4.42	Normalised Cr concentration at different soil sections after the experiments.	142
Figure 4.43	Physical observation of soil after the experiments under: a) 0.01M NaNO ₃ ; b) 0.1M citric acid; c) 0.1M EDTA (Left section: anode region (S1); Right section: cathode region (S4))	144
Figure 4.44	Electrokinetic properties for fixed electrode test and approaching cathode test using 0.1M EDTA as wetting agent: a) electric current across the soil; b) soil pH at different soil sections	146
Figure 4.45	Normalised concentration for a) Pb, and b) Cr after the experiments at different soil sections under fixed electrode and approaching electrode tests using 0.1M EDTA as the wetting agent.	149
Figure 4.46	Power consumption and removal efficiency in pretreated soil regions (S3-S4) under different types of wetting agents (AC: approaching cathode)	150

LIST OF TABLES

Table 2.1	Physical properties of Pb (Inchem, 1994; Lenntech, n.d)	7
Table 2.2	Maximum threshold value for Pb concentration in soil	12
Table 2.3	Summary of recent studies for soil washing in treating Pb contaminated soil	14
Table 2.4	Chemicals used for pH conditioning in electrokinetic soil remediation.	23
Table 2.5	Complexing agents used in electrokinetic soil remediation for heavy metals removal.	26
Table 2.6	Summary on the application of electrokinetic washing process	38
Table 3.1	Experimental parameters for two-stage electrokinetic washing	50
Table 3.2	Experimental parameters for electrokinetic soil pretreatment process.	51
Table 4.1	Characteristic of the natural soil used in this study	61
Table 4.2	Adsorption isotherms for Pb adsorption on soil under solution pH 5	65
Table 4.3	Power consumption for two-stage electrokinetic washing under different types of wash solutions	86
Table 4.4	Performance of two-stage electrokinetic washing on Pb removal using different types of wash solutions	86
Table 4.5	Parameters and levels of the experiments	88
Table 4.6	Experimental results using Response Surface Methodology based on Box-Behnken Design	89
Table 4.7	ANOVA for removal efficiency	90
Table 4.8	Predicted and experimental results for removal efficiency, effluent generated, and power consumption under the optimum conditions	109
Table 4.9	Comparison between the performance of two-stage electrokinetic washing and normal soil washing under optimum operating parameters.	119

LIST OF SYMBOLS AND ABBREVIATIONS

А	Cross sectional area of the column, m ²
AA	Approaching anode
AC	Approaching cathode
Al	Aluminium
ANOVA	Analysis of Variance
As	Arsenic
BBD	Box-Behnken Design
β	Dubinin-Radushkevich isotherm constant (mol ² kJ ⁻²)
CaCl ₂	Calcium chloride
Cd	Cadmium
CH ₃ COOH	Acetic acid
Cit	Citrate ion
Cl	Chloride ion
Cl ₂	Chlorine
Со	Cobalt
С	Metal concentration in the soil (mg/kg)
C_0	Initial metal concentration in the soil before experiment (mg/kg)
C _{ICP}	Concentration of metal in aqueous sample detected by ICP (mg/L)
C _{Pb,soil}	Normalised lead concentration in soil for modelling
C _{Pb,soil0}	Normalised lead concentration in soil after soil washing for modelling
СРС	Cetylpyridinium chloride
Cr	Chromium
Cs	Cesium
Cu	Copper
D	Dielectric constant
DC	Direct current
DF	Dilution factor
DPTA	Diethylene triamine pentaacetic acid
D _{soil}	Soil constant (porosity, tortuosity)
Ε	Electric field (V/m)
E_0	Electrode potential under standard condition
e	Electron
EDDS	Ethylenediamine-N,N'-disuccinic acid

EDTA	Ethylenediaminetetraacetic acid
EGTA	Ethylene glycol tetraacetic acid
EM	Electromigration
EOF	Electroosmotic flow
EOS	Electroosmosis
E _{PC}	Power consumed per kg Pb removed (kWh/kg)
ε ₀	permitivity of a vacuum (F/m)
EW	Electrokinetic washing
Fe	Iron
H^+	Hydogen ion
H_2O	Water
H_2SO_4	Sulfuric acid
HC1	Hydrochloric acid
Hg	Mercury
HNO ₃	Nitric acid
HPCD	Hydroxypropyl-β-cyclodextrin
Ι	Electric current, A
k	Overall kinetic constant (1/hr)
$K_2Cr_2O_7$	Potassium dichromate
K _F	Freundlich isotherm constant (mg/kg)
K_{L}	Langmuir isotherm constant (L/mg)
Mg	Magnesium
Mn	Manganese
m _{Pb}	Mass of Pb removed (kg)
ms	Mass of soil, kg
n	Adsorption intensity of the adsorbent for Freundlich isotherm
η	dynamic viscosity of solution
Na ₂ CO ₃	Sodium carbonate
Na ₂ EDTA	disodium Ethylenediaminetetraacetic acid
NaHCO ₃	sodium bicarbonate
NaNO ₃	sodium nitrate
NaOH	sodium hydroxide
Ni	Nickel
NO ₃ ⁻	Nitrate ion
NTA	Nitrilotriacetic acid

O_2	Oxygen
OH	Hydroxide ion
PAHs	Polyaromatic hydrocarbons
Pb	Lead
$Pb(NO_3)_2$	lead nitrate
Pb(OH) ₂	lead hydroxide
Pb(OH) ₃	plumbites
PCA	cementol antikorodin
PCDE	plasticizers cementol delta ekstra
PFA	Pulverised fuel ash
PPF	Polypropylene fibers
pZC	Point of zero charge
Q _{max}	Maximum adsorption capacity (mg/kg)
q _t	Adsorption of Pb per unit adsorbent at time t (mg/kg)
q _e	Adsorption capacity for Pb at equilibrium (mg/kg)
RSM	Response Surface Methodology
SDS	Sodium dodecyl sulfate
Si	Silicon
SS	Stabilisation and solidification
SW	Soil washing
t	Time, minutes (For pseudo second order kinetic model)
TCE	Trichloroethane
TCLP	Toxicity Chracteristic Leaching Procedure
Ti	Titanium
u	Ionic mobility (m^2/sV)
UESR	Upward electrokinetic soil remediation
V	Voltage or electric potential difference, V
Vhno3	Volume of HNO ₃ solution, (L)
V_0	Initial sorption rate, mg/(kg.min) (pseudo second order kinetic model)
V_{ES}	Volume of extraction solution after the extraction method (L)
\mathbf{V}_{EM}	Electromigration velocity (m/s)
\mathbf{V}_{EO}	Electroosmosis velocity (m/s)
ξ	Zeta potential
Zn	Zinc

LIST OF APPENDICES

Appendix A: Chemicals used in the analytical study	186
Appendix B: Procedures for soil characterisation study	187
Appendix B1: Procedure for cation exchange capacity (CEC)	187
determination in soil	187
Appendix B2: Procedures for specific gravity determination in soil	189
Appendix B3: Procedures for organic matter determination in soil	190
Appendix B4: Procedures for Tessier's Method	191
Appendix B5: Procedures for soil voidage	192
Appendix C: ANOVA for effluent generation and power consumption	193
Appendix C1: ANOVA for effluent generation	193
Appendix C2: ANOVA for power consumption	194

CHAPTER 1: INTRODUCTION

1.1 Background

Lead (Pb) has been identified as one of the top six toxic pollutants in the world (McCartor and Becker, 2010) as it can cause acute and chronic illnesses to human by damaging central and peripheral nervous systems, cardiovascular and reproductive systems as well as gastrointestinal and urinary tracts when it is inhaled and ingested (Duruibe *et al.*, 2007; Wuana and Okieimen, 2011). As a result of its extensive use, humans are exposed to Pb via several pathways. One of the pathways is soil contamination. Pb concentration as high as 751.98-138,000 mg/kg is reported in shooting range soil, dumpsite for lead-acid battery manufacturing and mining regions over the world (Adejumo *et al.*, 2010; Hashimoto *et al.*, 2010; Kim *et al.*, 2010; McCartor and Becker, 2010). Therefore, a proper treatment must be conducted for these soils.

Among the soil remediation methods available, soil washing is one of the *ex situ* remediation methods which is frequently used in treating Pb and heavy metals contaminated soil, especially for higher permeability soil that has lower silt and clay content (Dermont *et al.*, 2008; FRTR, n.d). There have been several studies on the treatment of Pb contaminated soil using different washing agents such as acids and chelating agents (Isoyama and Wada, 2007; Qiu *et al.*, 2010; Yang *et al.*, 2012; Voglar and Lestan, 2013) where positive results are reported. However, it is worth noting that the main disadvantages of soil washing is the wastewater generation and the presence of chemical agents in the wash fluid, which complicates the treatment process for spent solutions and is costly (Dermont *et al.*, 2008). Moreover, a high ratio for solution: soil in the system of 1.25-80 mL: 1 g (Isoyama and Wada, 2007; Qiu *et al.*, 2010; Qiu *et al.*, 2010) is another disadvantage for soil washing as the amount of spent wash solution to be treated is large.

Other than soil washing, stabilisation and solidification (SS) is another soil remediation method that is often used for treating heavy metal contaminated soils (Wuana and Okieimen, 2011). This process involves addition of cement and binding agents to the contaminated materials to reduce the mobility and migration of the contaminants (Pensaert *et al.*, 2008; Wuana and Okieimen, 2011; Falciglia *et al.*, 2014). However, it is observed that SS experiences the drawbacks of high cost cement usage and high solid waste generation (Pensaert *et al.*, 2008; Wuana and Okieimen, 2011).

In contrast, electrokinetic process is a potential soil treatment method, which uses electricity as the driving force for contaminant transport. The induction of low intensity direct current through the soil augments the transport of contaminants via two major mechanisms, namely electromigration for ion transport and electroossmosis for pore water/neutral compounds transport (Acar and Alshawabkeh, 1993; Acar *et al.*, 1995; Shenbagavalli and Mahimairaja, 2010). It is noteworthy that the transport of metal ions via electromigration is generally independent of hydraulic flow as electromigration would occur as long as an electric field is induced, regardless of the cessation of electroossmosis (Acar *et al.*, 1995) and pore flow.

In comparison to conventional soil washing which requires high amount of wash solution for high removal of heavy metals from the soil (Moutsatsou *et al.*, 2006; Zou *et al.*, 2009; Qiu *et al.*, 2010), electrokinetic process serves as an option as soil remediation method due to its ability to transport the desorbed heavy metals even in the absence of pore flow via electromigration (Acar *et al.*, 1995). Thus, the amount of wash solution needed is lower than that for normal soil washing. In addition, the ability of electrokinetic process to transport heavy metals in the absence of hydraulic and concentration gradients could also be utilised in soil pretreatment process to concentrate heavy metals into smaller soil volume before stabilisation and solidification is applied so that the solid waste generation and the amount of cement needed could be reduced.

This has created an interest to study the feasibility of electrokinetic process to be incorporated into other soil remediation methods such as soil washing and SS for maintaining/improving the treatment efficiency at low chemical solution consumption and reducing waste generation, respectively.

1.2 Research objectives

The study focuses on the incorporation of electrokinetic process into soil remediation methods. The first part of the study consists of the incorporation of electrokinetic process in soil washing as a novel two-stage electrokinetic washing in enhancing process efficiency for Pb removal at low wash solution consumption. The main objectives for this study are as follows:

- i) To determine the effectiveness of two-stage electrokinetic washing as soil remediation method and the best performing wash solution in Pb removal.
- ii) To evaluate the effects of operating parameters such as electric potential difference, wash solution concentration and initial Pb concentration on the process in terms of removal efficiency, effluent generation and power consumption.
- iii) To optimize Pb removal efficiency of two-stage electrokinetic washing at low power consumption with minimum effluent generation.
 - iv) To understand the transport mechanisms involved in two-stage electrokinetic washing process.

The second part of the study involves the technical feasibility evaluation on the application of simple electrokinetic process as soil pretreatment method to provide volume reduction on contaminated soil via concentrating Pb into smaller soil volume. The main objectives of this study are:

- To evaluate the effectiveness of electrokinetic process as a soil pretreatment method for concentrating Pb into smaller soil volume using different types of chemical solutions as wetting agents.
- ii) To investigate the effect of single Pb contaminated soil and Pb/Cr cocontaminated soil on the performance of electrokinetic process in Pb electromigration.
- iii) To investigate the effect of approaching electrode technique on the performance of electrokinetic process from the aspects of electric current, soil pH, Pb electromigration and power consumption.

1.3 Scope of study

The feasibility study on the incorporation of electrokinetic process into soil remediation methods in treating Pb contaminated soil was carried out using artificially contaminated soil in laboratory scale experiments. The first part of the study was the evaluation on the feasibility of two-stage electrokinetic washing in removing iron/mineral oxides adsorbed/bounded Pb from artificially contaminated soil using conventional chemical solutions. The process was conducted as an *ex-situ* soil remediation process in a laboratory scale column without pH control. The feasibility study focuses on the effect of chemical solutions and operating conditions on Pb removal efficiency, effluent generation, system stability and power consumption. Pb was chosen as the heavy metal contaminant in this study as it is reported to be one of the most recalcitrant divalent metal cation species that is adsorbed/bounded on the metal/mineral oxides surface in comparison to other metals (Violante *et al.*, 2007).

The second part of the study was the investigation on the ability of electrokinetic process to serve as soil pretreatment method in reducing the volume of Pb contaminated soil via electromigration. In this study, a sandbox design was used and the electrode rods were injected directly into the soil, which was further saturated with the chemical solution. The effect of Pb species and the presence of opposing charged competitive metals, Cr on Pb electromigration was investigated on a comparative basis. In addition, the effect of approaching electrode in Pb electromigration enhancement was also evaluated in this study.

1.4 Thesis outline

The thesis consists of five chapters, namely: i) introduction, ii) literature review, iii) methodology, iv) results and discussion, and v) conclusions and recommendations.

Chapter 1 provides the background of the study, research objectives, scope of study and the structure of the thesis.

Chapter 2 contains the literature review. The first section of this chapter is the introduction of Pb in terms of its properties, its production, applications and toxicity, as well as the current information on Pb contaminated soil and the environmental regulations. This chapter also explains the conventional soil remediation methods and their disadvantages. In addition, the introduction and process description of electrokinetic soil remediation method are also covered in this chapter. Lastly, recent studies on the application of conventional electrokinetic process as soil remediation methods are also included.

Chapter 3 describes the experimental set-up and procedures for desorption tests, two-stage electrokinetic washing and electrokinetic soil pretreatment process. Besides, analytical methods such as electric current measurement and determination of pH as well as Pb concentration in the soil samples are also elaborated.

Chapter 4 presents the results and discussion of the study. This chapter is categorised into three major sections. The first section provides discussion on the characterisation of the soil and maximum Pb contamination level for the selected soil. The second section deals with the performance of two-stage electrokinetic washing in removing Pb from the soil using different types of wash solutions, and the discussions are based on the aspects of removal efficiency, effluent generation, system stability and power consumption. Also, key experimental findings on the performance of two-stage electrokinetic washing under different operating conditions are discussed. Finally, the ability of electrokinetic soil pretreatment process in electromigrating and concentrating Pb into smaller soil volume under different types of wetting agent and different Pb species are discussed in the third section.

Chapter 5 covers the conclusions of this study and some recommendations for future studies.

CHAPTER 2: LITERATURE REVIEW

This chapter consists of three main sections which discuss the usage, hazards and soil contamination from lead, the types of soil remediation methods available and the application of electrokinetic process as a soil remediation method.

2.1 Lead

Lead (Pb) is a metal element which is toxic to organisms due to its ability to bioaccumulate. Pure Pb is soft, malleable and has bluish white colour while it becomes dull grayish when exposed to air (Anonymous, 2013; Lenntech, n.d.). This metal is normally found in the earth's crust in the form of minerals such as galena, anglesite, cerussite, mimetite and pyromorphite (Inchem, 1994). General physical properties of Pb are as shown in Table 2.1.

Parameters	Value
Symbol	Pb
Molecular weight	207.2
Melting point, °C	327.4
Boiling point, °C (at atmospheric pressure)	1740
Specific gravity (at 20 °C)	11.34
Vapor pressure, kPa (at 1000 °C)	0.233
Solubility in water	Highly dependent on the anion of its
	salt. Generally soluble in acids such
	as HCl and HNO _{3.}

 Table 2.1: Physical properties of Pb (Inchem, 1994; Lenntech, n.d.)

2.1.1 Production and application of lead

Production and consumption of Pb has been increasing worldwide since 1964 until present day (ILZSG, 2013; LDAI, n.d). Pb is mainly produced in China, Australia, USA, Peru, Canada, Mexico, Sweden, Morocco, South Africa and North Korea (Global Info Mine, n.d). Global Pb production as high as 10,654,000 tonnes was recorded in 2012 and 49.4% of the production came from ore mining and smelting (ILZSG, 2013) while recycle of scrap Pb products contributed as the secondary source for Pb production (LDAI, n.d).

Pb has been extensively used in battery manufacturing whereby about 75% of total Pb production was used for this industry (McCartor and Becker, 2010; LDAI, n.d). Pb has also been used as raw materials in industrial manufacturing products such as ammunition, bearings, cable covers, plumbing, lining, pipes, ceramic glazes, weights, caulks, dyes, pigments and pesticides (Inchem, 1994; McCartor and Becker, 2010; Wuana and Okieimen, 2011). Besides manufacturing products, it also serves as a main component to alloy with other metals in the manufacturing of maintenance free storage batteries, anodes in electro-winning processes, pipes and sheets in chemical installations and nuclear shielding, printing and high-detail casting (Wuana and Okieimen, 2011), type metals and bronze productions (Inchem, 1994). Since Pb has been used widely in various processes, they create significant pathways for Pb to enter a human body. It is therefore necessary to review the exposure risk to human beings.

2.1.2 Toxicological effect of lead on human

Pb is highly toxic metal, which can harm living organisms through cumulative poisoning. It has been identified as one of the top six toxic threats in the world (McCartor and Becker, 2010). It has also been positioned at the second place among top

20 hazardous substances by Agency for Toxic Substances and Disease Registry, ATSDR and U.S. Environmental Protection Agency, USEPA (Ahmedzeki, 2013). Pb accumulation occurs when >0.5 mg/day is absorbed, and fatal poisoning is reported for adults and children at a dosage of 500mg absorbed Pb and blood Pb levels of 1250 µg/L, respectively (Inchem, 1994).

Ingestion and inhalation are two major entry routes for human Pb poisoning (Inchem, 1994). Exposure of Pb by these routes is mainly from breathing air due to work environment, drinking water, eating food containing Pb and ingestion of Pb containing paint or Pb polluted soil (Inchem, 1994; Wuana and Okieimen, 2011) as a result of water and ground pollutions from ore mining, milling and smelting activities, automobile exhaust, and the discharges from manufacturing industries as mentioned in Sub-section 2.1.1 (Inchem, 1994; Duruibe *et al.*, 2007; McCartor and Becker, 2010; Wuana and Okieimen, 2011).

Ionic Pb, Pb oxides and hydroxides are reported to be the general forms of Pb that are released into the water body and soil (Wuana and Okieimen, 2011). These inorganic Pb compounds can cause chronic poisoning, acute poisoning and asymptomatic poisoning (Inchem, 1994). Other than death, Pb poisoning also causes poor intelligence quotient, shorten attention span, hyperactivity and mental deterioration for children (Duruibe *et al.*, 2007; McCartor and Becker, 2010; Wuana and Okieimen, 2011). Moreover, Pb also tends to cause damage to human body systems such as central nervous system, peripheral nervous system, cardiovascular system, reproductive system, gastrointestinal as well as urinary tracts (Inchem, 1994; Duruibe *et al.*, 2007; Wuana and Okieimen, 2011). The damage of these systems may lead to health disorders from the aspects of i) brain diseases (brain damage, neurological and nerve disorder/damage, reduce in reaction time, loss of memory, nausea, insomnia, anorexia, hypertension, chronic headaches); ii) dysfunctions for muscles, bones and blood circulation systems

(vomiting, pain/weakness of joints and muscles, cramps and paraesthesia, abdominal colic, inhibition on haemoglobin synthesis, anaemia, increased blood pressure, gout), iii) urinary system diseases (nephropathy, bloody urine, kidneys dysfunction), and iv) reproductive diseases such as infertility, damage to fetus and birth defects (Inchem, 1994; Duruibe *et al.*, 2007; McCartor and Becker, 2010; Wuana and Okieimen, 2011).

2.1.3 Soil contamination by lead

Soil contamination is one of the pathways that can lead to Pb ingestion and inhalation. Many cases for Pb contaminated soil are reported in 80-90's. High Pb concentration in soil has been reported in many countries such as United Kingdom, United States of America, Poland, New South Wales, Australia and France (USEPA, 1991; Markus and McBratney, 2001) in 80-90's with a maximum Pb concentration of 209,000 mg/kg. The report published by EPA in 1991 shows that the serious soil region are normally from Pb related industries such as battery soil, gold casing soil and scrap Pb soil (USEPA, 1991) besides normal mining and smelting areas (Duruibe *et al.*, 2007; Wuana and Okieimen, 2011). To date, Pb contaminated soil remains as an environmental issue. Soil contamination is still reported from various countries such as Australia, South Korea, Nigeria, Japan, China, Senegal, Mexico, Dominican Republic, Peru and Zambia with a Pb concentration of 751.98 – 138,000 mg/kg in shooting range soil, dumpsite for Pb-acid battery manufacturing and mining regions (Liu *et al.*, 2005; Adejumo *et al.*, 2010; Hashimoto *et al.*, 2010; Kim *et al.*, 2010; McCartor and Becker, 2010; Sanderson *et al.*, 2010).

High Pb concentration in soil causes severe effects to the residents nearby. Blacksmith Institute (USA) has reported several case studies on Pb contaminated sites. Among the most serious place is Dakar, Senegal where 18 children were reported dead from acute Pb poisoning due to Pb dust and soil exposure as a result of Pb-acid battery recycling activities (McCartor and Becker, 2010). Moreover, scrap metal smelting and mining activities also produce air borne particulates containing lead, which increase dermal exposure of the residents. Majority of the residents, especially children were reported to have high Pb level in blood in the contaminated areas in Haina, Dominican Republic; La Oroya, Peru; Kabwe, Zambia; and Zamfara, Nigeria (McCartor and Becker, 2010).

Pb contaminated soil is also reported in Malaysia. Scrap metal yard in Kuala Lumpur (Yang *et al.*, 2004) has been reported to have Pb concentration of as high as 1005 mg/kg. Besides that, places such as dumping sites, paddy fields and car wash in Kangar are also identified to have a concentration of 600-800 mg/kg Pb, mainly due to the use of Pb containing fertilizer and emission from vehicles near the busy roads (Nor Wahidatul *et al.*, 2012). Other equally hazardous activities in this country over the past century are mining and Pb-acid battery recycling. Therefore, soil remediation has become an absolute necessity in Malaysia in order to eliminate the sources for Pb contamination through these activities.

2.1.4. Environmental regulations for lead concentration in soil

In order to reduce and eliminate Pb poisoning issues, maximum threshold concentration has been regulated by the government for controlling and minimising soil contamination. Table 2.2 shows the threshold Pb concentration in soil in different countries. A high range of soil concentration of Pb from 60-1300 mg/kg is monitored. Higher threshold concentration is applied for industrial areas while a more stringent concentration is applied for residential areas. For Malaysia, the threshold concentration for Pb in the soil is 400 mg/kg (residential soil) and 800 mg/kg (industrial soil).

Country	Threshold value, mg/kg
Alberta	800 ^a
Canada	60-500 ^a
France	100 ^c
Germany	100 ^c
India	400-1300 ^d
Japan	600 ^a
Korea	200-700 ^f
Malaysia	400-800 ^b
Netherland	580 ^c
	150-600 ^a
Taiwan	100 ^c
UK	550c
USA	250-1000 ^a
	68°
	107 ^e

Table 2.2: Maximum threshold value for Pb concentration in soil

(Source: ^aBeyer, 1990; ^bDoE-Malaysia, 2009; ^cWan Zuhairi, 2011; ^dIDEM, 2012; ^cUSEPA, 2012; ^fMOE-Korea, n.d)

2.2 Popular methods for treating heavy metals contaminated soil

A number of remediation methods have been investigated by researchers for treating heavy metals contaminated soils. Among the popular methods are: i) soil washing, and ii) stabilisation and solidification (Wuana and Okieimen, 2011).

2.2.1 Soil washing

Soil washing is one of the remediation methods for heavy metals contaminated soils. It is a technology that uses liquids (water or chemical additives) and mechanical processes to scrub the soil (USEPA, 1996). In general, soil washing can be categorised into two types: namely i) physical separation, and ii) chemical extraction (Dermont *et al.*, 2008). According to Dermont *et al.* (2008), a physical separation technique that uses gravity and flotation concepts is generally more suitable to treat soil that is contaminated with particulate and discrete metals. On the other hand, chemical extraction is more suitable for the metals that are adsorbed on soil particles in their ionic form (non-detrital metals) whereby the chemical reagents remove metals from soil to aqueous phase via dissolution and complexation processes. Soil washing has been widely studied in laboratory scales, pilot scales, as well as full scale in treating heavy metals contaminated soil (USEPA, 1996; Dermont *et al.*, 2008).

Table 2.3 summarises recent studies on soil washing via chemical extraction for remediating Pb contaminated soil. A number of chemical reagents have been applied in soil washing such as acids (Cline and Reed, 1995; Tejowulan and Hendershot, 1998; Tawinteung et al., 2005; Moutsatsou et al., 2006; Isoyama and Wada, 2007; Yang et al., 2012), surfactants (Torres et al., 2012), and chelating agents (Cline and Reed, 1995; Tejowulan and Hendershot, 1998; Abumaizar and Smith, 1999; Tawinteung et al., 2005; Moutsatsou et al., 2006; Zou et al., 2009; Qiu et al., 2010; Mohanty and Mahindrakar, 2011; Yuan et al., 2012; Voglar and Lestan, 2013) and positive results on Pb removal have been reported. However, it is worth noting that chemical extraction based soil washing often involves extra cost for treatment of spent chemicals in the wash solution (Dermont et al., 2008). Moreover, the use of high amount of wash solution also creates further disadvantage as the volume of spent wash solutions/wastewater generated is large (Kim et al., 2012a). Table 2.3 shows that the solution: soil ratio used in soil washing is often in a range of 1.25-80. This causes generation of high volume of spent chemical solution even when treating small amount of soil, which in turn increases the treatment cost. Besides that, the efficiency of soil washing in removing heavy metals is also limited by soil texture such as silt/clay content (Dermont et al., 2008).

13

Heavy	Metal	Soil	Wash as lotters	Calatian.	David David David	Derestian	Defenses
metals	concentration, mg/kg	5011	Wash solution	Solution: soil ratio	Best Removal efficiency, %	Duration, h	References
Pb	100-10000	Superfund soils	HCI	10	92.5%	24	(Cline and Reed, 1995)*
			Na ₂ EDTA		93.5%		
			CH ₃ COOH		57.3%		
			CaCl ₂		47.2%		
Pb HM	800 6.3-2600	Silty sand	HCl	5	HCI: Poor removal	288 (24 per extraction)	(Tejowulan and Hendershot,
			EDTA		Pb: 49% HM: 53-84%		1998)
Pb	742	Silty sand	Na ₂ EDTA	12.5 (shake	Pb: 104.6%	48 (Shake	(Abumaizar
HM	603-1231			flask)	HM: 40.7- 112.1%	flask)	and Smith, 1999)*
				300 pore		300 pore	,
				volume	Pb: 100%	volume	
				(column	HM: 25-80%	(column	
				experiment)		experiment)	
Pb	153-1620	Surface/ subsurface	EDTA HNO3	5-10	85-95%	1	(Tawinteung et al., 2005)
		soil	Ammonium	20 pore	74-85%	20 pore	<i>ei ul.</i> , 2003)
		5011	citrate	volume	110370	volume	
Pb	64195	Top soil	HCl,	33.33	Pb: 83%	1	(Moutsatsou
HM	1030-223600				HM: 46-97%		et al., 2006)
			HNO ₃ ,		Pb: 44%		
					HM: 10-45%		
			H_2SO_4		Pb: 6% HM: 30-93%		
		•	Na ₂ EDTA		Pb: 44% HM: 13-42%		
			HCl	1.25-5	25-78%	1	(Isoyama and
Pb	1000	Surface/		1.25-5	20 1070		
Pb	1000	Surface/ subsurface soil	CaCl ₂	1.25-5	20 , 0,0		Wada,
	1000 254.4	subsurface		20	Pb: 40.91%	16 (2 per	
Pb	254.4	subsurface soil	CaCl ₂		Pb: 40.91% HM: 18.01-	16 (2 per cycle)	Wada, 2007)*
Pb HM		subsurface soil Silty sand	CaCl ₂		Pb: 40.91%	cycle)	Wada, 2007)* (Zou <i>et al.</i> , 2009)
Pb Pb HM Pb HM	254.4 1.7-325.0 254.35	subsurface soil	CaCl ₂		Pb: 40.91% HM: 18.01-		Wada, 2007)* (Zou <i>et al.</i> ,
Pb HM Pb	254.4 1.7-325.0	subsurface soil Silty sand	CaCl ₂ Na ₂ EDTA Na ₂ EDTA	20	Pb: 40.91% HM: 18.01- 72.98% Pb: 27.4% HM: 2.5-53%	cycle)	Wada, 2007)* (Zou <i>et al.</i> , 2009) (Qiu <i>et al.</i> ,
Pb HM Pb	254.4 1.7-325.0 254.35	subsurface soil Silty sand	CaCl ₂ Na ₂ EDTA	20	Pb: 40.91% HM: 18.01- 72.98% Pb: 27.4%	cycle)	Wada, 2007)* (Zou <i>et al.</i> , 2009) (Qiu <i>et al.</i> ,
Pb HM Pb	254.4 1.7-325.0 254.35	subsurface soil Silty sand	CaCl ₂ Na ₂ EDTA Na ₂ EDTA Oxalate	20	Pb: 40.91% HM: 18.01- 72.98% Pb: 27.4% HM: 2.5-53% Pb: 1.5% HM: 22-60%	cycle) 24	Wada, 2007)* (Zou <i>et al.</i> , 2009) (Qiu <i>et al.</i> ,
Pb HM Pb	254.4 1.7-325.0 254.35	subsurface soil Silty sand	CaCl ₂ Na ₂ EDTA Na ₂ EDTA	20	Pb: 40.91% HM: 18.01- 72.98% Pb: 27.4% HM: 2.5-53% Pb: 1.5%	cycle)	Wada, 2007)* (Zou <i>et al.</i> , 2009) (Qiu <i>et al.</i> ,
Pb HM Pb	254.4 1.7-325.0 254.35 1.73-325.01	subsurface soil Silty sand Silty sand	CaCl ₂ Na ₂ EDTA Na ₂ EDTA Oxalate Na ₂ EDTA/Oxalate	20	Pb: 40.91% HM: 18.01- 72.98% Pb: 27.4% HM: 2.5-53% Pb: 1.5% HM: 22-60% 99%	cycle) 24 8 (2 per	Wada, 2007)* (Zou <i>et al.</i> , 2009) (Qiu <i>et al.</i> , 2010)
Pb HM Pb HM	254.4 1.7-325.0 254.35	subsurface soil Silty sand	CaCl ₂ Na ₂ EDTA Na ₂ EDTA Oxalate	20 20 80	Pb: 40.91% HM: 18.01- 72.98% Pb: 27.4% HM: 2.5-53% Pb: 1.5% HM: 22-60%	cycle) 24 8 (2 per cycle)	Wada, 2007)* (Zou <i>et al.</i> , 2009) (Qiu <i>et al.</i> ,
Pb HM Pb HM Pb	254.4 1.7-325.0 254.35 1.73-325.01 26	subsurface soil Silty sand Silty sand	CaCl ₂ Na ₂ EDTA Na ₂ EDTA Oxalate Na ₂ EDTA/Oxalate	20 20 80	Pb: 40.91% HM: 18.01- 72.98% Pb: 27.4% HM: 2.5-53% Pb: 1.5% HM: 22-60% 99% Pb: 53.87%	cycle) 24 8 (2 per cycle)	Wada, 2007)* (Zou <i>et al.</i> , 2009) (Qiu <i>et al.</i> , 2010) (Mohanty and Mahindrakar,
Pb HM Pb HM Pb	254.4 1.7-325.0 254.35 1.73-325.01 26	subsurface soil Silty sand Silty sand	CaCl ₂ Na ₂ EDTA Na ₂ EDTA Oxalate Na ₂ EDTA/Oxalate	20 20 80	Pb: 40.91% HM: 18.01- 72.98% Pb: 27.4% HM: 2.5-53% Pb: 1.5% HM: 22-60% 99% Pb: 53.87%	cycle) 24 8 (2 per cycle)	Wada, 2007)* (Zou <i>et al.</i> , 2009) (Qiu <i>et al.</i> , 2010) (Mohanty and
Pb HM Pb HM Pb Cr Pb	254.4 1.7-325.0 254.35 1.73-325.01 26 61 70	subsurface soil Silty sand Silty sand Sand Clay-	CaCl ₂ Na ₂ EDTA Na ₂ EDTA Oxalate Na ₂ EDTA/Oxalate Na ₂ EDTA	20 20 80 5	Pb: 40.91% HM: 18.01- 72.98% Pb: 27.4% HM: 2.5-53% Pb: 1.5% HM: 22-60% 99% Pb: 53.87% Cr: 48.35% Pb: 10% HM: 10-85% Pb: 0-43%	cycle) 24 8 (2 per cycle) 6	Wada, 2007)* (Zou <i>et al.</i> , 2009) (Qiu <i>et al.</i> , 2010) (Mohanty and Mahindrakar, 2011)* (Torres <i>et al.</i> ,
Pb HM Pb HM Pb Cr Pb	254.4 1.7-325.0 254.35 1.73-325.01 26 61 70	subsurface soil Silty sand Silty sand Sand Clay-	CaCl ₂ Na ₂ EDTA Na ₂ EDTA Oxalate Na ₂ EDTA/Oxalate Na ₂ EDTA	20 20 80 5	Pb: 40.91% HM: 18.01- 72.98% Pb: 27.4% HM: 2.5-53% Pb: 1.5% HM: 22-60% 99% Pb: 53.87% Cr: 48.35% Pb: 10% HM: 10-85%	cycle) 24 8 (2 per cycle) 6	Wada, 2007)* (Zou <i>et al.</i> , 2009) (Qiu <i>et al.</i> , 2010) (Mohanty and Mahindrakar, 2011)* (Torres <i>et al.</i> ,

Table 2.3: Summary of recent studies for soil washing in treating Pb contaminated soil

* Spiked soil was used; HM: Other heavy metals
2.2.2 Stabilisation and solidification

'Stabilisation and solidification' (SS) is one of the soil remediation methods for treating heavy metal contaminated soils (Wuana and Okieimen, 2011). This process involves addition of binding agents to the contaminated materials to reduce the mobility and migration of the contaminants (Pensaert *et al.*, 2008; Wuana and Okieimen, 2011; Falciglia *et al.*, 2014). SS has also been widely used for immobilisation of hazardous wastes, landfill leachate, sludge sediments and petroleum drill cuttings (Pensaert *et al.*, 2008; Wuana and Okieimen, 2011; Falciglia *et al.*, 2014). In comparison to other treatment methods, SS is applicable to most types of soil, sediment and mineral waste (Pensaert *et al.*, 2008) and it is identified by USEPA as one of the best available technologies (Voglar and Leštan, 2011; Desogus *et al.*, 2013).

Generally, the use of chemicals such as Portland cement and Pozzolanic cement together with the additives such as barite aggregates (Falciglia *et al.*, 2014), PCDE, PCA, Tween80, PPF, Akrimal (Voglar and Leštan, 2011), potassium dihydrogen phosphate, ferric chloride hexahydrate (Desogus *et al.*, 2013), PFA (Erdem and Özverdi, 2011; Kogbara *et al.*, 2013) and lime (Erdem and Özverdi, 2011) are proven to give promising results on stabilizing and immobilizing the contaminants. However, it is observed that SS tends to generate significant amount of solid waste (Wuana and Okieimen, 2011). Besides that, Pensaert *et al.* (2008) claimed that cement based SS had been an expensive process due to the high price of cement. These disadvantages often reduce the feasibility of SS.

2.3 Electrokinetic soil remediation

Electrokinetic soil remediation is a process that operates under a low magnitude direct current in the soil, as opposed to a hydraulic pressure gradient, which promotes the migration of various contaminants under coulombic forces (Shenbagavalli and Mahimairaja, 2010). This process produces an electrical gradient that acts as a driving force for the transport of various pollutants in saturated/ unsaturated soils via electrodes. As it is free from hydraulic gradient limitation, electrokinetic process is generally suitable for treating both low and high permeability soils (Shenbagavalli and Mahimairaja, 2010; Alcántara *et al.*, 2012; Giannis *et al.*, 2012). It has been reported that electrokinetic process is capable of removing a wide range of contaminants from the soil, and these contaminants include heavy metals, phenols, petroleum oils/hydrocarbons and radioactive substances (Acar *et al.*, 1995; Doering *et al.*, 2001; Korolev, 2006; Shenbagavalli and Mahimairaja, 2010; Kim *et al.*, 2011a). However, the scale up application of this treatment process is still scarce.

2.3.1 Process description

There are two mechanisms for contaminant transport in a moist soil; the first involves the migration of dissolve ions towards their respective electrodes, whilst the second mechanism occurs as water/neutral compounds transport, whereby contaminants are flushed towards the electrode chambers from anode to cathode. These mechanisms consist of two main phenomena; namely electromigration for removing metal ions, and electroosmosis for removing organic compounds (Acar and Alshawabkeh, 1993; Acar *et al.*, 1995; Reddy and Cameselle, 2009; Ma *et al.*, 2010; Shenbagavalli and Mahimairaja, 2010; Kim *et al.*, 2011a).



Figure 2.1: Electrokinetic process in soil treatment

Figure 2.1 illustrates the mechanisms for electrokinetic process in the soil. In general, two electrodes, namely an anode and a cathode, are introduced into the soil together with two chambers with porous wall (the anode chamber and the cathode chamber). The aqueous solutions in the chambers, which act as electrolytes/washing agents, undergo electrolysis when a direct current is induced. The hydrogen ion (H⁺) is produced in the anode chamber and the hydroxide ion (OH⁻) is produced in the cathode chamber during the electrolysis process, as shown in Equations (2.1) and (2.2), respectively (Acar and Alshawabkeh, 1993; Reddy and Cameselle, 2009):

$$2H_2O \rightarrow O_2 + 4H^+ + 4e^ E_0 = -1.229V$$
 (2.1)

$$2H_2O + 2e^- \to H_2 + 2OH^ E_0 = -0.828V$$
 (2.2)

Figure 2.1 shows that the induced electric field migrates H^+ towards the cathode through the soil. As the movement of H^+ in the soil towards the cathode (acid front) is approximately 1.75 to 2 times higher than OH⁻ movement (base front) towards the anode (Acar and Alshawabkeh, 1993; Acar *et al.*, 1995; Chung and Kang, 1999; Gioannis *et al.*, 2008; Park *et al.*, 2009; Kim *et al.*, 2011b), a net acid front advancement from the anode region to the cathode region will progress along the soil normally,

consequently affecting the pH and properties of the soil from the anode to the cathode region.

As the acid and base fronts advance, the electrical field also induces cations and anions migration in the soil. This mechanism is known as electromigration and it governs the removal of desorbed metal ions from soil under electrokinetic process. Metal cations are transported towards the cathode and vice versa for anions, following a similar trend for the acid and base fronts, as shown in Figure 2.1. The velocity for electromigration v_{EM} (m/s) is proportional to ion mobility *u*, and the electric field of the system E, as shown in Equation (2.3) (Baraud *et al.*, 1997). It is important to note that, unlike electroosmosis, electromigration of the ions will still occur as long as an electric field is induced in a wet soil. This process may continue despite the cessation of electroosmosis (Acar *et al.*, 1995) and this serves as an advantage for heavy metal removal from contaminated soil.

$$v_{EM} = u.E$$

(2.3)



Figure 2.2: Schematic diagram for electroosmosis process

Electroosmosis is the second mechanism in electrokinetic process. This mechanism governs the transport of neutral organic compounds and water in the soil. Electroosmosis can be defined as the movement of pore water that transports neutral compounds in the soil due to the movement of ions, as shown in Figure 2.1. A schematic diagram illustrating this mechanism is shown in Figure 2.2. Soil normally contains a negative surface charge due to imperfection in the mineral lattice, which attracts cations in the soil close to the soil surface (Acar et al., 1995). When an electric field is induced, the excess cations which are close to the soil surface (diffuse double layer) tend to move towards the cathode. According to Acar et al. (1995), the movement of cations and their associated water molecules yield a net strain on the pore fluid and thus transform into a shear force through the viscosity of the fluid. As the surface charge of the soil is normally negative, the force for the pore fluid is likely to be cationic and the transport of the pore fluid will be directed towards the cathode region, as shown in Figure 2.2. However, it is worth noting that the direction of electroosmosis strongly follows Helmholtz-Smoluchowski theory, which is dependent on the surface charge and zeta potential of the soil, as shown in Equation (2.4) where electroosmotic flow (EOF) velocity v_{EO} (m/s) is directly proportional to permittivity of a vacuum ε_0 , electric field of the system E, dielectric constant of the solution D, zeta potential of the soil ξ , and is inversely proportional to dynamic viscosity of the solution, η (Lee *et al.*, 2009; Asadi et al., 2013). Electroosmosis may progress from cathode to anode if the soil pH is below its point of zero charge (pZC), where ξ becomes positive values (Asadi et al., 2013).

$$v_{EO} = -\frac{D\varepsilon_0 \xi}{\eta} E \tag{2.4}$$

2.3.2 Factors affecting remediation efficiency using electrokinetic process

Electrokinetic process has been applied for removal of various heavy metals and organic compounds under different types of soil, ranging from sandy soil to clayey soil (Shenbagavalli and Mahimairaja, 2010). It is worth noting that the degree of efficacy of electrokinetic soil remediation is hardly consistent among the publications as it is highly dependent on the interactions between the nature of soil, contaminants and operating conditions/configurations. Nevertheless, it is safe to conclude that the removal efficiency for heavy metals is governed by several parameters such as i) electric potential difference/current, ii) pH conditioning, iii) chemical agents, iv) soil texture, and v) contaminant species. The effect of each parameter on the performance of electrokinetic process is discussed in the following subchapters from the view of heavy metal removal and electroosmotic flow (EOF).

2.3.2.1 Electric potential difference and current

Performance of electrokinetic process is enhanced by increasing the applied electric potential difference (voltage) and current. According to Hamed and Bhadra (1997), an applied current of 10 mA produced a water flow of 750 mL in 15 days whilst the increase in electric current to 30 mA produced the same amount of water in five days. This is mainly due to higher electric field across the soil enhances the rapid movement of cations in the diffuse double layer (Figure 2.2), consequently yielding higher shear force to the pore fluid and increases the EOF. As in line with Equations (2.3) and (2.4), the increment in electricity not only increases EOF of the system but also enhances removal efficiency for heavy metals such as Pb (Chung and Kang, 1999; Murillo-Rivera *et al.*, 2010), Ni (Reddy and Karri, 2006a), and non-metal ion fluorine (Kim *et al.*, 2009a). In addition, the removal efficiency for heavy metals can be

improved under higher electricity even though EOF is not enhanced. This is observed in the works of Zhou *et al.* (2004) and Yuan and Chiang (2008) which suggested that an increase in electric potential difference could slightly enhance the removal of copper from 81% to 85% (Zhou *et al.*, 2004) and arsenic from 35.4% to 44.8% (Yuan and Chiang, 2008) via improved electromigration even though the EOF is not significantly enhanced.

In spite of the enhancements discussed above, it is important to note that an excessively high electricity application during electrokinetic process can induce air bubble generation/foaming and soil heating, and may adversely affect soil properties, such as soil buffering capacity, glucose-induced soil respiration and urease/phosphate activities (Acar *et al.*, 1995; Colacicco *et al.*, 2010; Yeung, 2011; Pazos *et al.*, 2012). Also, the use of very high voltage gradient or current densities will increase H^+ concentration to a high level that could subsequently increase the positive charge of the soil surface, and may result in the reduction or reversal of the EOF from anode to cathode (Gioannis *et al.*, 2008). This is observed in the work of Back *et al.* (2009) and Park *et al.* (2009) which showed EOF was enhanced under higher electricity, but in a reverse direction. This is found to inhibit the removal of cation metal Zn (Park *et al.*, 2009), but enhance removal of arsenic anion (Back *et al.*, 2009). In addition to H^+ concentration, excessive increase in electricity also tend to increase the electrolysis rate for OH in the cathode, which favours metal ion precipitation near the cathode region and thus decrease the removal efficiency of metal ions from soil (Zhou *et al.*, 2005).

Other than affecting electrolysis rate, excessively high electricity may also cause insufficient residence/contact time between the wash solution and the contaminants to form soluble compounds. This consequently inhibits removal efficiency, especially for recalcitrant heavy metals. This is supported by several reported works on the removal of heavy metals/organic compounds (Reddy *et al.*, 2010) and degradation of nitrate ions on

permeable reactive barrier (Suzuki *et al.*, 2012) whereby an increase in electricity is reported to give poorer removal/degradation efficiency. Considering all the criteria mentioned above, Yeung (2011) has suggested that 1V/cm is suitable for laboratory scale studies.

2.3.2.2 pH conditioning

Electrolyte/wash solution conditioning is another method to enhance the performance of electrokinetic soil remediation. In general, two types of electrolyte conditioning are available, namely i) anolyte conditioning, and ii) catholtye conditioning. Anolyte conditioning involves pH control in the anode chamber, which strongly affects the soil pH. Preservation of soil system pH via applying high pH anolyte, could prevent low soil pH especially near the anode region. In order to achieve this condition, several chemicals are often applied in the anode chamber, as shown in Table 2.4.

Higher soil pH increases the EOF from anode to cathode due to increase in negativity in the zeta potential, as shown in Equation (2.4) (Hamed and Bhadra, 1997; Yuan *et al.*, 2006; Yuan *et al.*, 2007; Baek *et al.*, 2009; Gómez *et al.*, 2009; Kim *et al.*, 2009a). The increase in EOF enhances the removal efficiency, as reported in the works for removing organic contaminant benzo[α]pyrene (Gómez *et al.*, 2009), phenolic compounds (Cong *et al.*, 2005), arsenic (Baek *et al.*, 2009), and fluorine (Kim *et al.*, 2009a). Furthermore, anolyte conditioning is also reported to contribute in lower energy consumption for removing azo dye RB5 (Pazos *et al.*, 2007) and arsenic (Baek *et al.*, 2009).

Conditioning	Chemical used for pH conditioning in electrokinetic soil remediation References	
Anolyte conditioning	NaOH	(Saichek and Reddy, 2003)
		(She <i>et al.</i> , 2003)
		(Reddy and Saichek, 2004)
		(Amrate <i>et al.</i> , 2005)
		(Yuan <i>et al.</i> , 2006)
		(Pazos <i>et al.</i> , 2007)
		(Baek <i>et al.</i> , 2009)
		(Li <i>et al.</i> , 2009)
		(Pham <i>et al.</i> , 2009)
		(Wan <i>et al.</i> , 2009)
		(Kim <i>et al.</i> , 2009a)
		(Li <i>et al.</i> , 2010)
		(Kim <i>et al.</i> , 2012c)
	Na ₂ CO ₃	(Pham <i>et al.</i> , 2009)
	1	(Li <i>et al.</i> , 2011)
	Na ₂ CO ₃ /NaHCO ₃	(Yuan <i>et al.</i> , 2006)
	1142003/11411003	(Yuan et al., 2000)
	Ammonium Acetate	(Murillo-Rivera <i>et al.</i> , 2010)
	Tris-acetate	(Alcántara <i>et al.</i> , 2012)
Catholyte conditioning	HNO ₃	(Ribeiro <i>et al.</i> , 2005)
Califoryte conditioning	IIIVO3	(Wang <i>et al.</i> , 2006, 2007)
		(Gioannis <i>et al.</i> , 2008)
	X	(Baek <i>et al.</i> , 2009)
		(Park <i>et al.</i> , 2009) (Park <i>et al.</i> , 2009)
		(Kim et al., 2009)
		(Kim et al., 2009c) (Kim et al., 2009c)
		(Murillo-Rivera <i>et al.</i> , 2010)
		(Kim <i>et al.</i> , 2011b)
		(Almeira O <i>et al.</i> , 2012)
	C	(Giannis <i>et al.</i> , 2012)
	H ₂ SO ₄	(Hansen <i>et al.</i> , 2007)
	112004	(Pazos <i>et al.</i> , 2007)
		(Li <i>et al.</i> , 2007) $(Li et al., 2009)$
· · ·		(Li <i>et al.</i> , 2010)
	HCl	(Kim <i>et al.</i> , 2008a, 2008b)
	nei	(Kim <i>et al.</i> , 2011b)
		(Suzuki <i>et al.</i> , 2012)
	Lactic acid	(Zhou <i>et al.</i> , 2004)
		(Zhou <i>et al.</i> , 2005)
		(Zhou <i>et al.</i> , 2006)
	Acetic acid	(Chung and Kang, 1999) (<i>Kim at al.</i> 2000b)
		(Kim <i>et al.</i> , 2009b) (Sumbarda-Ramos <i>et al.</i> , 2010)
	Acetate and citrate buffers	(Sumbarda-Kanos <i>et al.</i> , 2010) (Zhou <i>et al.</i> , 2004)
	i source and ontate outfors	(Kim et al., 2009)
		(Li <i>et al.</i> , 2012)

Table 2.4: Chemicals used for pH conditioning in electrokinetic soil remediation

Even though anolyte conditioning could enhance the removal efficiency of organic contaminants by inducing higher EOF, a high pH in soil may result in low metal mobility via metal hydroxide precipitation near the cathode region. Hence, catholyte conditioning using the acids as shown in Table 2.4 is introduced in order to reduce hydroxide precipitation in the cathode region. It is found that, in general, reduction in overall soil pH increases metal ion desorption and mobility, and this increases the removal efficiency (Chung and Kang, 1999; Zhou *et al.*, 2005; Wang *et al.*, 2006, 2007; Gioannis *et al.*, 2008; Baek *et al.*, 2009; Kim *et al.*, 2009b; Kim *et al.*, 2011b; Almeira O *et al.*, 2012). Other than acid buffers, soil pH could also be homogenized at low level via other operating conditions such as well-timed polarity exchange (Pazos *et al.*, 2006; Han *et al.*, 2010; Ma *et al.*, 2010), circulation of high pH catholyte into the anode chamber (Lee and Yang, 2000), and using longer cathode chamber to prevent OH-migration into the soil (Rajić *et al.*, 2012).

However, it is worth noting that low pH does not enhance organic contaminant removal. Wang *et al.* (2007) found that the use of HNO₃ to obtain a lower soil pH impeded p-xylene and phenanthrene transport. Furthermore, it may be noted that the decrease in soil pH causes EOF to reduce from anode to cathode, and even results in EOF reversal, as reported by Zhou *et al.* (2004), Zhou *et al.* (2005) and Baek *et al.* (2009). The reversal of EOF can further risk cation removal due to a counter direction of cation transport between electromigration and electroosmosis (Park *et al.*, 2009; Kim *et al.*, 2009b). Thus, a suitable range of pH must be used based on soil properties and the type of contaminant that needs to be removed.

Moreover, it is to be noted that low soil pH does not always increase metal desorption as the affinity of metal ions on the soil surface also governs metal desorption and removal efficiency in the electrokinetic process, as shown in the works of Kim *et al.* (2005b), Turer and Genc (2005), Zhou *et al.* (2005), Li *et al.* (2010), Giannis *et al.*

(2009) and Giannis *et al.* (2012) for treating soil contaminated with multiple heavy metals. For example, Pb was reported to have a much higher affinity to the soil surface even under low pH conditions, and thus a low removal efficiency was observed compared to other types of metal ions (Turer and Genc, 2005; Kim *et al.*, 2005b; Giannis *et al.*, 2009).

2.3.2.3 Complexing agents

Sub-section 2.3.2.2 presents evidence that pH adjustment using acids, bases and buffer solutions affect the behaviour of the electrokinetic process whereby lower pH enhances metal cation desorption whilst a higher pH enhances the EOF and organic compounds removal. However, the use of lower pH to enhance metal desorption risks reversal of EOF at the same time. This not only retards the removal efficiency of metal cations but also causes problems when dealing with simultaneous treatments of organic compounds and metal cations. Based on this concern, studies on the utilisation of complexing agents to enhance metal desorption process without affecting the pH significantly have been conducted whereby surfactants and aminopolycarboxylates based chelating agents are often used, as summarized in Table 2.5.

Table 2.5 shows that ionic and non-ionic surfactants have been applied as the wash solutions in electrokinetic process for heavy metal removal. However, surfactants often show lower effectiveness in enhancing metal desorption than organic contaminants (Maturi and Reddy, 2006; Reddy *et al.*, 2006b; Li *et al.*, 2010; Reddy *et al.*, 2010; Alcántara *et al.*, 2012). As shown in Table 2.5, heavy metal removal efficiency demonstrated by surfactants is low, in general, at <35%. High Pb removal by Tween80 in the work of Alcantara *et al.* (2012) is mainly due to the inclusion of ethylenediaminetetraacetic acid (EDTA) in Tween80 solution as chelants. A simple

batch extraction test also shows that low Pb extraction at 40% is reported when only Tween80 is used as the extracting agent (Alcántara *et al.*, 2012), which indicates that surfactants are weak desorption agents for heavy metals.

Group	removal Complexing Contaminants Contaminant Removal References						
Group	agents	Containnants	concentration,	Efficiency, %	Kelefences		
	agents			Efficiency, 70			
C	CDC	A	mg/kg	21.700/			
Surfactants	CPC	Arsenic	966	31.70%	(Yuan and Chiang, 2008)		
	SDS	Copper	180 mg	Not mentioned	(Han et al., 2010)		
	Igepal CA- 720	Heavy metals	1.8-38000 3.53-47000	0-30.00%	(Reddy <i>et al.</i> , 2006b) (Reddy <i>et al.</i> , 2010)		
	Tergitol 15-S-	Zinc	910	24.30%	(Park <i>et al.</i> , 2009)		
	Tween 80	Lead	5000	82.40%	(Alcántara et al., 2012)		
		Heavy metals	1.8-38000 17.4-714.1	0-15.00%	(Reddy <i>et al.</i> , 2006b) (Colacicco <i>et al.</i> , 2010)		
	HPCD	Nickel	500	0-0.93%	(Maturi and Reddy, 2006)		
			42.1		(Li et al., 2010)		
		Zinc	2313-3123 2415	0-14.00%	(Li <i>et al.</i> , 2009) (Li <i>et al.</i> , 2010)		
		Heavy metals	<0.5-21300 1.8-38000	Negligible	(Reddy and Ala, 2005) (Reddy <i>et al.</i> , 2006b)		
Chelating agent	EDTA	Arsenic	966	44.80%	(Yuan and Chiang, 2008)		
		$\cdot \times \mathbb{N}$	953-972 0.54-183.90	77.00% 39.80%	(Yuan <i>et al.</i> , 2009) (Kim <i>et al.</i> , 2012c)		
		Cesium	514 Not mentioned	59.30% 57.90%	(Kim <i>et al.</i> , 2009c) (Kim <i>et al.</i> , 2008a)		
			Not mentioned	57.90%	(Kim <i>et al.</i> , 2008b)		
		Cobalt	238	84.60%	(Kim <i>et al.</i> , 2009c)		
			Not mentioned Not mentioned	83.50% 83.50%	(Kim <i>et al.</i> , 2008a) (Kim <i>et al.</i> , 2008b)		
	5	Copper	438 13.25-220.25 180 mg	73.00% 17.40% Not mentioned	(Zhou <i>et al.</i> , 2004) (Kim <i>et al.</i> , 2012c) (Han <i>et al.</i> , 2010)		
		Lead	5000	82.40%	(Alcántara <i>et al.</i> , 2010)		
		Loud	1000 4432	72.30-83.80% 36.28%	(Zhang <i>et al.</i> , 2014) (Amrate <i>et al.</i> , 2005)		
			11.49-504.61	23.90%	(Kim <i>et al.</i> , 2012c) $(Kim et al., 2012c)$		
		Heavy metals	<0.5-21300 18-270	Low removal 1.60-13.20%	(Reddy and Ala, 2005) (Kim <i>et al.</i> , 2011b)		
			7.7-94080 3.53-47000 17.4-714.1	2.00-80.00% 0-80.00% 28.00-84.00%	(Gioannis <i>et al.</i> , 2008) (Reddy <i>et al.</i> , 2010) (Colorized <i>et al.</i> , 2010)		
	NTA	Cadmium	1/.4-/14.1	<u>28.00-84.00%</u> 95.00%	(Colacicco <i>et al.</i> , 2010) (Giannis <i>et al.</i> , 2009)		
	11111	Copper	1005	60.00%	(Giannis <i>et al.</i> , 2009)		
		Lead	455	25.00%	(Giannis <i>et al.</i> , 2009)		
	DPTA	Cadmium	145	70.00%	(Giannis <i>et al.</i> , 2009)		
		Copper	1005	20.00%	(Giannis <i>et al.</i> , 2009)		
		Lead	455	20.00%	(Giannis <i>et al.</i> , 2009)		
		Heavy metals	<0.5-21300	Low removal	(Reddy and Ala, 2005)		
	EGTA	Cadmium	145	65.00%	(Giannis <i>et al.</i> , 2009)		
		Copper	1005	10.00%	(Giannis <i>et al.</i> , 2009)		
		Lead	455	15.00%	(Giannis <i>et al.</i> , 2009)		
	EDDS	Cadmium	4000	≈50.00%	(Suzuki <i>et al.</i> , 2014)		
		Lead	7366.67	≈30.00% ≈45.00%	(Suzuki <i>et al.</i> , 2014)		

 Table 2.5: Complexing agents used in electrokinetic soil remediation for heavy metals removal

In contrast, aminopolycarboxylates based chelating agents emerge as superior chemicals in enhancing heavy metals desorption from soil through the formation of water soluble complexes (Yeung and Gu, 2011). Among the agents, EDTA is widely applied in electrokinetic soil remediation, as shown in Table 2.5. EDTA is reported to provide higher metal removal efficiency than unenhanced systems (Amrate *et al.*, 2005; Reddy and Ala, 2005; Gioannis *et al.*, 2008; Yuan and Chiang, 2008; Han *et al.*, 2009; Yuan *et al.*, 2009; Colacicco *et al.*, 2010). As a chelant for anion complex formation, EDTA is observed to form complexes more effectively with metal ions than organic contaminant PAHs, especially when treating real contaminated soil (Colacicco *et al.*, 2010; Reddy *et al.*, 2010). Nevertheless, EDTA is still reported to be capable of removing organic contaminants, such as diesel (Han *et al.*, 2009) and PAHs (Reddy *et al.*, 2010) via soil mineral dissolution and change in charge of solid particle which releases PAHs (Reddy *et al.*, 2010).

In spite of outperforming an unenhanced electrokinetic system, EDTA generally has a lower metal removal rate than acidic agent such as citric acid, acetic acid, oxalic acid, HNO₃ and HCl due to its weaker acidity for metal desorption process (Kim *et al.*, 2008a, 2008b; Kim *et al.*, 2009c; Kim *et al.*, 2011b). Moreover, as a result of complex formation, EDTA causes an accumulation of metal ions in the soil matrix at an early stage of the treatment due to the counter migration of positive metal ions from the anode region and negatively charged metal complexes from the cathode region (Giannis *et al.*, 2009). This phenomenon results in lower metal removal efficiency.

The performances of chelants are also dependent on the applied electricity, pH condition and soil properties (Gioannis *et al.*, 2008; Yuan and Chiang, 2008; Kim *et al.*, 2009c; Kim *et al.*, 2011b; Alcántara *et al.*, 2012). A higher voltage may result in improved EDTA performance in comparison to citric acid (Yuan and Chiang, 2008). Besides that, the effectiveness of EDTA is also strongly dependent on the pH of the

system as EDTA has poor solubility under acidic condition, which consequently affects the complexation process in the soil (Alcántara *et al.*, 2012; Harris, 2013). Moreover, the difference in soil properties also causes the variation in the performance of EDTA. For example, Gioannis *et al.* (2008) claimed that EDTA displays superior performance in relation to citric acid as citric acid preferentially forms H-citrate complex which reduces the availability of citrate for metal complexes. On the other hand, poorer EDTA performance when compared to citric acid was observed in another study when Fe(III) was present in the soil. EDTA forms highly stable complex with Fe(III), which reduces EDTA availability for contaminants (Kim *et al.*, 2011b). It may also be suggested that the nature of target metal for remediation also governed the effectiveness of EDTA (Reddy and Ala, 2005; Kim *et al.*, 2011b), NTA, DTPA and EGTA as chelants (Giannis *et al.*, 2009).

2.3.2.4 Acidic based agents

Acidic based agents have been utilised as catholyte for pH control, as discussed in Sub-section 2.3.2.2. Besides that, due to the acidic characteristic of these agents, they are often used in processing fluids (wash solutions) for desorbing heavy metals. Strong acid HNO₃ is frequently utilized by researchers to reduce pH in cathode region and cathode chamber. When HNO₃ is applied in electrokinetic process, it depolarizes the cathode and enhances metal ion desorption/deprecipitation, especially near the cathode region (Chung and Kang, 1999; Wang *et al.*, 2006, 2007; Gioannis *et al.*, 2008; Park *et al.*, 2009; Kim *et al.*, 2009b; Giannis *et al.*, 2012). However, an excessive low soil pH may cause reversal of EOF towards anode chamber, which is in opposite direction for the electromigration for metal cations (Ribeiro *et al.*, 2005; Gioannis *et al.*, 2008; Baek *et al.*, 2009; Park *et al.*, 2009; Kim *et al.*, 2009b). This may reduce the removal efficiency, as reported by Park *et al.* (2009), where Zn is found to accumulate in the middle region of the soil, as a result of reverse EOF. Furthermore, excessive use of HNO₃ in catholyte beyond the optimum concentration of >0.12 M is found to give undesirable effect in cathode chamber's pH, as a result of NO₃⁻ reduction (Almeira O *et al.*, 2012).

Besides providing better desorption condition, other acids such as HCl, acetic acid and citric acid also contribute to desorption via complexation with heavy metals. HCl provides Cl⁻ ions for formation of water soluble metal complexes (Kim *et al.*, 2011b) besides enhancing soil acidity. This is reported to give higher removal efficiency for Ni, Cu, Zn and Pb at 67-69.5% than HNO₃ at 7.9-42%, as shown in the work of Kim *et al.* (2011b). Besides strong acid, weak acids such as acetic acid and citric acid are also often used as buffer solutions for controlling pH (Park *et al.*, 2007; Shen *et al.*, 2007; Gioannis *et al.*, 2008; Kim *et al.*, 2009b; Sumbarda-Ramos *et al.*, 2010; Kim *et al.*, 2011b; Li *et al.*, 2012) and solubilizing agent/processing fluids (Chung and Kang, 1999; Yuan and Chiang, 2008; Kim *et al.*, 2008a, 2008b; Kim *et al.*, 2009c; Kim *et al.*, 2009d; Han *et al.*, 2010; Murillo-Rivera *et al.*, 2010; Pazos *et al.*, 2012).

It is noted that the performance of acid based agents is strongly case dependent. For example, acetic acid is reported to show higher removal efficiency than citric acid for removing Co, Cs (Kim *et al.*, 2008a, 2008b; Kim *et al.*, 2009c) and Pb (Murillo-Rivera *et al.*, 2010). It is claimed that acetic acid attacks and dissolves hydrous oxides in the soil and creates lower pH condition than citric acid (Kim *et al.*, 2008a, 2008b; Kim *et al.*, 2008a, 2008b; Kim *et al.*, 2009c). Besides that, Murillo-Rivera *et al.* (2010) also suggested that acetic acid, which results in higher negativity of zeta potential of soil could increase the permeability of the soil and thus improve Pb removal efficiency. When citric acid was applied, Pb-citrate complex of low mobility was formed that was subsequently adsorbed on soil carbonate content, hindering Pb removal (Murillo-Rivera *et al.*, 2010).

Furthermore, citric acid is also reported to have lower Cu removal efficiency than H_2SO_4 , as a result of possible reactions in the oxidation-reduction process when bipolar electrode plates were used (Hansen *et al.*, 2007).

However, contradictory observation is reported by Li *et al.* (2012) whereby citric acid is claimed to be more effective when treating Cr contaminated soil, as it enhanced desorption and formed more stable complexes than acetic acid. In addition, citric acid also performs better than EDTA and HNO₃ when treating sandy soil (Kim *et al.*, 2008a, 2008b; Kim *et al.*, 2009c; Murillo-Rivera *et al.*, 2010; Kim *et al.*, 2011b), even though metal-EDTA complexes have higher stability than citrate complexes (Kim *et al.*, 2011b). The reduction of available EDTA in the system, as a result of complexation with other metals, such as Fe(III) in the soil and inability of NO₃⁻ on metal complexation, are suggested as the main reasons for citric acid to outperform EDTA and HNO₃, respectively (Kim *et al.*, 2011b). However, the characteristics of contaminants influence the effectiveness of citric acid as well. This is reported by Yuan and Chiang (2008) that EDTA performed better than citric acid in mobilising As from As-Fe soil, especially under higher applied voltage.

2.3.2.5 Soil texture

Soil texture is another important factor that causes changes in electrokinetic phenomena. It influences the performance of wash solutions in electrokinetic process in terms of electroosmosis and contaminant removal efficiency (Reddy and Saichek, 2003). Several soil properties are reported to affect the performance of electrokinetic process such as particle size, buffering capacity and organic matter/carbon content (Yang and Liu, 2001; Reddy and Saichek, 2003; Kim *et al.*, 2006; Gioannis *et al.*, 2008; Sumbarda-Ramos *et al.*, 2010; Alcántara *et al.*, 2012).

The size of soil particles affects electroosmosis. Electroosmosis was found to be effective when treating sandy soil by electric field applied under zero hydraulic gradient whilst no EOF was recorded by clayey soil, due to high permeability of sandy soil (Sumbarda-Ramos *et al.*, 2010). Other than particle size, soil buffering capacity is also responsible for governing the effectiveness of electroosmosis in soil especially for low permeability soil (Saichek and Reddy, 2003b, as cited in Alcántara *et al.*, 2012). This observation is reported by Kim *et al.* (2006) based on Hadong clay, which has higher buffering capacity, yielded higher EOF than low buffering capacity kaolin. Reddy and Saichek (2003) also suggest that glacial till soil with high carbonate content and buffering capacity provides higher electroosmosis than kaolin when 3% Tween80 is applied as wash solution due to soil's ability to maintain negative zeta potential. This is reported to enhance the removal efficiency for phenanthrene in glacial soil. On the other hand, soil with high buffering capacity soil acidification (Reddy *et al.*, 2006b; Gioannis *et al.*, 2008) for metal desorption.

In addition, soil with higher organic matter content is also found to give lower removal efficiency, especially for organic contaminants such as phenenthrene, as a result of its stronger binding on the organic matter (Reddy and Saichek, 2003). This conforms to the work of Alcantara *et al.* (2012) who reported that the removal efficiency for Pb and phenanthrene for sandy soil was lower than that for kaolin clay (73.5-82.4% vs 93.3-95.1%) under similar operating conditions, as the organic matter content for sandy soil was higher than that for kaolin clay. Furthermore, Yang and Liu (2001) who worked on remediation of trichloroethane, TCE contaminated soil using electrokinetic-fenton process further claimed that higher organic matter, but also

reduces TCE destruction efficiency via extra consumption of hydroxyl radicals on the organic matters.

2.3.2.6 Contaminant species

The effectiveness of electrokinetic soil remediation on heavy metal removal is also dependent on the types of metals. For example, Pb is reported as one of the most recalcitrant metal species to adsorb/bind on soil mineral surface (Violante et al., 2007). Its strong affinity for soil often causes relatively low removal efficiency in comparison to other metals such as Cd and Zn when a co-contaminated soil is treated (Turer and Genc, 2005; Kim et al., 2005b; Giannis et al., 2009; Giannis et al., 2012). This is also well demonstrated in the works of Zhou et al. (2005) and Li et al. (2010) who claimed that low removal was achieved for Ni and Cu in comparison to Zn, as a result of their higher affinity to bind on high clay (soil particle size $< 2\mu m$) containing soil as well as high iron containing red soil (Violante et al., 2007). In addition, Turer and Genc (2005) also suggested that competitive migration of metal species in a co-contaminated soil affected the removal efficiency in comparison to soil containing single contaminant. The presence of significant amount of Zn and Cu in the soil reduced Pb removal efficiency from 48% to 32% (Turer and Genc, 2005). However, it is worth noting that metal mobility could be enhanced by using proper chemical solutions/complexing agents in electrokinetic process. For example, Cu has lower mobility than Ni (Violante et al., 2007) and it often showed lower removal efficiency than Ni when HNO₃ and EDTA were used as the wash solutions (Kim et al., 2011b). However, Cu removal efficiency (69.8%) was reported to increase compared to Ni (49.4%) when citric acid was used (Kim et al., 2011b).

2.3.3 Incorporation of electrokinetic process with soil washing/ flushing

Electrokinetic process has also been incorporated into soil washing/flushing for improving heavy metal removal efficiency. The different types of electrokinetic washing/flushing are discussed in the sub-sections below.

2.3.3.1 Electrokinetic flushing

Electrokinetic flushing is an electrokinetic process which is aided by a hydraulic pump for transporting the wash solution through the soil, as shown in Figure 2.3. This configuration is reported to provide significant enhancement in removal efficiency than normal electrokinetic process, especially when treating higher permeability sandy soil at lower remediation time (Kim *et al.*, 2008a, 2008b; Reddy *et al.*, 2010). Kim *et al.* (2008a) reported that electrokinetic flushing enhanced the removal of Co and Cs by about 25% and 35%, respectively, in comparison to normal electrokinetic process in a five day treatment period. This is also in line with the work of Kim *et al.* (2008b) which claimed that an enhancement of 30% was achieved in four days as flushing process detached the adhered Co and Cs from soil surface by pressure flushing. The removal enhancement by flushing is also highly dependent on soil permeability and treatment duration whilst it is less dependent on wash solution injection rate (Kim *et al.*, 2008a).



Figure 2.3: Electrokinetic soil flushing equipment (Kim et al., 2008b)

In terms of effluent generation, electrokinetic flushing produced more effluent than that for a normal electrokinetic process, where a solution: soil ratio of 2.4 mL/g was recorded for the former whilst 1.6 mL/g for the later, as reported by Kim *et al.* (2008a). In another study, Kim *et al.* (2008b) observed effluent generation to the extent of 96.5 mL/day in electrokinetic flushing in comparison to normal electrokinetic process, which showed 53.4 mL/day. Nevertheless, the authors also claimed that electrokinetic flushing only produced an effluent volume of less than 10% of that for soil washing. This observation was also reported by Kim *et al.* (2008a) that a much higher solution: soil ratio of 47.1 mL/g was needed for soil washing in order to achieve high removal efficiency in comparison to electrokinetic flushing at 2.4 mL/g.

2.3.3.2 Vertical electrokinetic flushing

Electrokinetic flushing has also been conducted in vertical configuration. Figure 2.4 illustrates two types of vertical configurations, namely i) downward configuration, and ii) upward configuration. Unlike normal electrokinetic configuration, a downward configuration consists of an anode which is installed on the top of the soil tank and a cathode at the bottom. Downward configuration is claimed to minimize accumulation of contaminants in the middle of soil while preventing generation of waste solution via anode chamber contamination, as a result of possible reverse EOF at low soil pH condition (Kim *et al.*, 2009c). In terms of removal efficiency, Kim *et al.* (2009c) reported that the use of vertical electrokinetic flushing achieved high removal efficiency of 98.30% and 88.80% for Co and Cs, respectively. Moreover, the effluent generated by this configuration was also reported to be much lower, at 2.4 mL/g soil in comparison to normal soil washing at 30 mL/g.



Figure 2.4: Vertical Electrokinetic Soil Washing: (a) Downward configuration (Kim *et al.*, 2009c); (b) Upward configuration (UESR)

Other than downward module, the feasibility of upward electrokinetic soil remediation (UESR) has also been investigated using a cathode at the top and anode at the bottom, as shown in Figure 2.4b. Upward contaminant transport is claimed to produce minimal site disturbance and reduces overall treatment cost as contaminants tend to accumulate on the surface of the cathode, which can be further treated easily (Wang *et al.*, 2006, 2007). UESR has been applied in electrokinetic process on the removal of heavy metals such as As, Cd, Cr, Cu, Pb, Ni, Zn (Wang *et al.*, 2006, 2007; Giannis *et al.*, 2012) and organic contaminant such as p-xylene, pyrene, phenanthrene and other PAHs (Wang *et al.*, 2007; Lima *et al.*, 2011; Giannis *et al.*, 2012). In general, heavy metals electromigrate upwards and are concentrated in the cathode region or precipitated on the cathode (Wang *et al.*, 2006; Giannis *et al.*, 2012). The removal of heavy metals via effluent is also noticed, especially when HNO₃ is used as the wash solution instead of non-acidic distilled water (Wang *et al.*, 2006, 2007; Giannis *et al.*, 2007; Giannis *et al.*, 2007; Jiannis *et al.*, 2006; Jiannis *et al.*, 2007; Jiannis *et al.*, 2012).

In contrast, the migration of organic compounds is dominated by pore water movement. This was well documented in the works of Wang *et al.* (2007) and Giannis *et al.* (2012) which reported that p-xylene and pyrene were transported by electroosmosis and leachate flow, respectively. The mobility for organic compounds was higher when distilled water was used as the wash solution due to favourable soil zeta potential for electroosmosis (Wang *et al.*, 2007). In addition, light non-aqueous phase liquid phenanthrene could be removed in UESR by the upward movement of pore solution flow in the system that provided additional upward force towards the cathode at the upper surface (Wang *et al.*, 2007; Giannis *et al.*, 2012). Nevertheless, the removal of heavy metals and PAHs by UESR is also dependent on the diameter and height of the cell (Wang *et al.*, 2006, 2007), which could be correlated with current density and contaminant solubilities in water (Wang *et al.*, 2006, 2007; Giannis *et al.*, 2007; Giannis *et al.*, 2012).

The concept of UESR has also been applied together with conventional horizontal electrode configuration, known as 2D crossed electric field, and has been observed as having a positive influence on chromium removal (Zhang *et al.*, 2010). With regard to electromigration enhancement by horizontal electrodes, the risk of chromium diffusion into deep soil layers and groundwater is found to be minimized by the application of a UESR. This is mainly due to the fact that UESR transports Cr(VI) anions upward, counteracting the downward gravitational and concentration gradient migration (Zhang *et al.*, 2010). Therefore, improved remediation efficiency is obtained in comparison to any individual application of horizontal and vertical configurations.

2.3.3.3 Electrokinetic washing

Other than the use of flushing in electrokinetic process, soil washing has also been incorporated with electrokinetic process for the removal of heavy metals and radionuclides from soil (Kim *et al.*, 2012b), concrete (Kim *et al.*, 2009d) and ashes (Kim *et al.*, 2013; Liao *et al.*, 2014). This process is normally applied in two different stages involving initial washing and the electrokinetic process. The washing is applied to provide desorption condition and initial contaminant removal. Then, the soil/concrete/ash is further transferred into an electrokinetic equipment for further removal process. An example of electrokinetic washing equipment and the summary of these studies are as shown in Figure 2.5 and Table 2.6, respectively.



Figure 2.5: Example of electrokinetic washing equipment (Kim et al., 2012b)

The use of electrokinetic washing is reported to provide treatment for ash to non-hazardous waste via reduction in the reactive fraction of heavy metals. This is well reported in the work of Liao *et al.* (2014) which applied initial Pb contaminated ash washing using deionized water before the application of electrokinetic process. The initial washing provided significant impact especially on Pb mobility, whereby it removed chloride ions in the ash and this prevented inhibition of Pb fraction changes into less reactive forms during electrokinetic process. A comparison on Toxicity Characteristic Leaching Procedure (TCLP) further proved that electrokinetic washing process recorded better performance than normal electrokinetic process as TCLP detection for Pb was found to decrease from 16.5mg/L to 0.746 mg/L and 15.2 mg/L, respectively (Liao *et al.*, 2014). A normal electrokinetic process failed to remove Pb, which was strongly bound to the carbonate content.

References	Medium	Contaminant	Process condition	Removal	Remarks
				efficiency	
(Kim et al.,	Concrete	Cs	<u>Washing</u>	<u>Washing</u>	Acid washing reduces
2009d)		Co	Wash solution: 3 M	Cs: 76.83%	pH and CaCO ₃ content
			HCl	Co: 85.0%	on the concrete. This
			Solution: concrete		enhances Co and Cs
			ratio: 1.5	Electrokinetic:	desorption.
			Duration: 4 h	Cs: 99.6%	
				Co: 99.7%	Solution: soil ratio
			Electrokinetic process		<u>(mL:g)</u>
			Wash solution: 0.01		Washing: 1.5
			M Acetic acid		Electrokinetic: 2.7-2.8
			Duration: 14.83 days		Electrokinetic washing:
					7.2
(Kim et al.,	Soil	Cs	Washing	Washing:	Longer duration needed
2012b)			Wash solution: HNO ₃	75-85%	when treating higher Cs
			2 times scrubbing		concentration ash
				Electrokinetic:	
			Electrokinetic process	>95.0%	
			Wash solution: 0.1 M		
			HNO ₃		
			Duration: 60 days		
(Kim et al.,	Radioactive	¹³⁷ Cs	<u>Washing</u>	Washing:	Electrokinetic process
2013)	ash	¹³⁴ Cs	Wash solution: HNO ₃	87-89%	provides further
			Solution: ash ratio: 1		removal of Cs to
			Duration: 2 h	Electrokinetic:	achieve the clearance
•			3 times washing	93.9-94.0%	concentration level
			Electrokinetic process		Longer duration needed
			Wash solution:		when treating higher Cs
			water/HNO ₃		concentration ash
			Duration: 10 days		
(Liao et al.,	Municipal	Pb	Washing	Not stated	Enhancement given by
2014)	solid waste	Cd	Wash solution:		washing is dependent
	incineration	Cr	deionized water	Metals	on the fraction of
	ash	Cu	Solution: ash ratio: 20	transform to	contaminants. E.g.
			Duration: 0.5 h	strongly binded	removal of metals
				species	bounded with carbonate
			Electrokinetic process		content could be
			Wash solution:		enhanced by washing
			deionized water		process.
			Duration: 168 h		

Table 2.6: Summary on the application of electrokinetic washing process

Furthermore, electrokinetic washing is also found to have favourable impact in remediating radionuclides polluted concrete and soil. Kim *et al.* (2009d) claimed that the use of electrokinetic process after concrete washing could further enhance the removal efficiency by 14.7-23.3% in comparison to normal washing. Application of initial washing by 3 M HCl was reported to break down the carbonate content in concrete and provided lower pH condition for desorption. The use of subsequent electrokinetic process could boost the overall removal of Co and Cs to >99% in comparison to single electrokinetic process, which only gave <1% Co and <65% Cs removal. This was also agreed by Kim *et al.* (2012b) and Kim *et al.* (2013) who showed that initial soil washing in multiple cycles could provide early conditioning and removal of the radioactive waste, ranging from 75% to 89% while subsequent application of electrokinetic process could further enhance the removal to below guideline concentration level.

However, electrokinetic washing is often applied in two different stages and processes. This causes a relatively high chemical consumption and effluent generation than normal soil washing and electrokinetic process. An example is given in the work of Kim *et al.* (2009d) where electrokinetic washing showed a high solution: concrete ratio of 7.2 mL/g, as a result of spent solution that is generated from soil washing and electrokinetic process only needed a ratio of 1.5 mL/g and 2.7-2.8 mL/g, respectively. Thus, a study on the other types of equipment process/configuration is required in order to enhance the feasibility of electrokinetic washing by maintaining high removal efficiency at lower wash solution consumption.

2.4 Summary of literature review

Soil washing and SS have been proven as effective methods for treating heavy metal contaminated soils. However, such processes have major shortcomings such as high chemicals consumption and high solid waste generation, which in turn increase the treatment cost. As discussed in Sub-section 2.2.1, high solution: soil ratio used in soil washing is responsible in increasing the cost for spent wash solution treatment. For SS, a high amount of waste solid generated increases the cost for landfill and cement as well.

In contrast, electrokinetic process is an effective method in remediating heavy metals contaminated soil and it is applicable for both high and low permeability soils. Its ability to promote migration of heavy metals even in the absence of pore flow (Acar *et al.*, 1995) makes it an attractive option along with other remediation technologies to improve the removal efficiency of heavy metals at lower chemical solution consumption.

Electrokinetic process has been incorporated into soil washing to enhance the removal process for heavy metals from soil, concrete and ash, as discussed in Subsection 2.3.3.3. However, it is worth noting that the wash solution consumption, which is the main disadvantage of soil washing, is not reduced by the introduction of electrokinetic process. In order to reduce spent wash solution generation while maintaining the removal efficiency, a different configuration for electrokinetic washing process, namely two-stage electrokinetic washing was introduced in this study for remediating Pb contaminated soil. This configuration involves the incorporation of soil washing and electrokinetic process in a single equipment, but operating at two different stages. The use of spent wash solution as catholyte after initial soil washing can reduce wash solution consumption while maintaining the removal efficiency for electrokinetic washing. The details of this process are elaborated in Section 4.4.

Other than incorporating in soil washing, electrokinetic process can also be applied as a soil pretreatment method to reduce contaminated soil volume before applying other treatment process such as SS due to its ability to migrate heavy metals even in the absence of pore flow. Thus, high chemical solution consumption can be prevented at the latter stage. This is investigated in the present study by using minimum chemical solution as wetting agent to saturate the soil, desorb the heavy metals and concentrate them into smaller soil volume under the influence of electric field. The details of the study are further discussed in Section 4.5.

CHAPTER 3: METHODOLOGY

This chapter discusses the methodologies used in the study. The chapter consists of five main sections, which are: i) materials, ii) equipment and apparatus, iii) experimental procedures, iv) analytical methods, and v) precautionary steps and minimisation of error.

3.1 Materials

3.1.1 Chemicals

The main chemicals used could be categorized into three major groups, namely i) heavy metals, ii) wash solutions/wetting agents, and iii) analytical chemicals. The heavy metals used in this study were lead nitrate, $Pb(NO_3)_2$ and potassium dichromate $K_2Cr_2O_7$, both of which were supplied in solid powder. The wash solutions investigated in this study were supplied in the forms of: i) HCl 37 wt% solution, ii) H₂SO₄ 98 wt% solution, iii) HNO₃ 67 wt% solution, iv) acetic acid 99% solution, v) EDTA 0.5 M solution at pH 8-9, vi) citric acid monohydrate crystal powder, and vii) NaNO₃ powder. The wash solutions of desired concentrations were prepared by diluting with/dissolving in distilled water prior to use. Other chemicals, which were used for analytical purpose are as summarized in Appendix A. The chemicals used were of analytical grade and were supplied by R&M Chemicals, Malaysia. These chemicals were used without any further treatment.

3.1.2 Soil

The soil was collected from first aquifer layer of the soil in Hulu Langat area, Selangor, Malaysia. The soil was pretreated by removing rocks, dried branches and grass and was dried in open air for one week. This soil was classified as sandy soil according to USDA Soil Classification as it has 92% sand content, with 8% silt and clay. The properties of uncontaminated soil such as cation exchange capacity (CEC), specific gravity, and organic matter content, were analysed based on the procedures as shown in Appendices B1-B3. The metal content in the soil was analyzed using USEPA 3050b, as discussed in Section 3.4.2.2. General properties of the uncontaminated soil used in this study are discussed in Section 4.2.

3.2 Equipment and apparatus

3.2.1 Electrokinetic washing system

Figure 3.1 illustrates the set-up for electrokinetic washing. The set-up fabricated was at a laboratory scale. The soil column was made of PTFE and had an inner diameter of 4 cm and 15 cm length. It was installed with four passive electrodes so that the voltage drop across the soil could be monitored when necessary. Two ends of the column were connected to two chambers, namely anode chamber and cathode chamber, each having a size of 4 cm diameter and 7.4 cm length. The column and chambers were separated by filter papers which functioned as selective barriers to prevent soil movement into the chamber. The other ends of the chambers were connected with the graphite electrode discs (diameter 4.9 cm; thickness 1 cm), as shown in Figure 3.2. These electrodes were connected to a direct current supply (Atten APS3003S-3D). The chambers were further connected with tubing and the overflow levels for both chambers were held at the same height in order to prevent extra effluent generation via hydraulic

head difference. The configuration of the setup for electrokinetic washing is as shown in Figure 3.1.



Figure 3.1: Experimental setup for electrokinetic washing system: (i) direct current supply, (ii) wash solution reservoir, (iii) peristaltic pump, (iv) anode chamber, (v) soil column, (vi) cathode chamber, (vii) overflow tube/effluent collector



Figure 3.2: Connection of graphite electrode into the wash solution chamber

3.2.2 Electrokinetic soil pretreatment system

The ability of the electrokinetic process to serve as a soil pretreatment method for concentrating heavy metals into smaller soil volume was investigated in a simple system. This system consisted of a polypropylene sand box with a dimension of 8.5 cm x 6.2 cm x 5 cm. In this study, wash solution chamber was absent. Instead, the electrodes were in direct contact with the soil so that a consistent electrode environment/condition could be established for both fixed electrode tests and approaching electrode tests. Six graphite rods with dimension of 0.7 cm diameter and 4 cm length were used as the electrodes. They are positioned at left, middle and right of the sandbox, as shown in Figure 3.3, which indicates anodes, approaching electrodes and cathodes, respectively. A pair of anodes (positioned at S1) and cathodes (positioned at S4) were used in this study to enhance the effective electric field in the soil. These electrodes are connected to the direct current supply, as shown in Figure 3.3.



Figure 3.3: Experimental setup for soil pretreatment system using electrokinetic process

3.2.3 Analytical instruments

CRISON Multimeter MM26+ was used for measuring pH. The electrical properties such as current and electric potential difference were measured by Multimeter Sunwa TE-832B. For metal ion concentration, Inductive Couple Plasma – Optical Emission System (ICP-OES) Perkin-Elmer Optima 7000DV was used for analysis. The other analytical equipment used for soil characterisation were as follows: Scanning Electron Microscopy with Energy Dispersive X-Ray Analysis (SEM-EDX), Fourier Transform Infrared Spectroscopy (FTIR), and X-Ray Diffraction (XRD).

3.3 Experimental procedures

3.3.1 Degree of soil contamination

The ability of the soil studied to be contaminated by Pb was justified. The study was conducted using shake flask adsorption test at a soil particle size of <0.85 mm. Based on some preliminary experiments (Ng *et al.*, 2016), Pb was favoured to adsorb on the soil at solution pH 5. Thus, the maximum Pb contamination level was investigated at solution pH 5. The tests were carried out by contacting 10 g of soil with 200 mL of Pb solution with a concentration of 150 ppm in a conical flask. The mixture was homogenized using an orbital shaker operating at a speed of 175 rpm for 6 h under room temperature. Aqueous sample was taken at different time interval and was filtered through 0.45 μ m syringe filter. These samples were then acidified with 0.1 M HNO₃ and Pb concentration in the aqueous samples was analysed by ICP-OES. The adsorption of Pb by the soil was determined using Equation (3.1). The adsorption capacity and kinetics were further evaluated by fitting the results for Pb adsorption into different kinetic model, Elovich model, second order kinetic model, film diffusion mass transfer kinetic

model and double exponential model. Pb speciation in the soil was determined using Tessier's Method (Tessier *et al.*, 1979; Gworek and Mocek, 2003; Venkateswaran *et al.*, 2007). The detailed procedure for this method is discussed in Appendix B4.

Adsorption isotherm study was carried out by contacting 4 g of soil with 100 mL of solution, with Pb concentration of 20 mg/L, 40 mg/L, 60 mg/L, 80 mg/L and 100 mg/L. The tests were conducted for 5 h in an orbital shaker at a rotational speed of 175 rpm under room temperature. Aqueous sample was taken after the experiment and was filtered through 0.45 μ m syringe filter. Then, the sample was acidified using 0.1 M HNO₃ and Pb concentration in the aqueous sample was analysed using ICP-OES. Pb adsorption by the soil was calculated using Equation (3.1). The results obtained were fitted into different isotherm models to determine the adsorption mechanism.

Adsorption by soil,
$$mg/kg = \frac{\text{Initial Pb in the solution - final Pb in the solution (mg)}}{\text{mass of soil used (kg)}}$$
 (3.1)

3.3.2 Soil spiking

Artificially contaminated soil was used in this study instead of real contaminated soil. The former was more suitable for use in laboratory experiments as the results obtained had less effect on the variation in contamination levels. Artificially Pb contaminated soil at desired contamination level was prepared by spiking the pretreated soil at a particle size of <0.85 mm with Pb(NO₃)₂ solution at pre-calculated concentration. The contaminated soil was then homogenized using a trowel/spatula and was air-dried for one week with occasional mixing. The contaminated soil was determined. The soil was then stored in a dark and closed space prior to use.

For Pb and Cr co-contaminated soil, the contamination was carried out by spiking the soil with solution containing $Pb(NO_3)_2$ and $K_2Cr_2O_7$. The mixture was spiked according to the procedure mentioned above. The concentration for Pb and Cr was determined and the soil was stored in a dark and closed space prior to use.

3.3.3 Screening for wash solutions

Chemical agents such as acids (Tawinteung *et al.*, 2005; Isoyama and Wada, 2007; Yang *et al.*, 2012) and chelating agent EDTA (Tawinteung *et al.*, 2005; Qiu *et al.*, 2010; Yang *et al.*, 2012; Voglar and Lestan, 2013) are widely used in soil washing to remove Pb. These groups of agents have also been applied to an electrokinetic process to remove Pb, as mentioned in Chapter 2. It is to be noted that the effectiveness of a wash solution is soil dependent, as discussed in Sub-sections 2.3.2.3 and 2.3.2.4. For example, EDTA showed high Pb removal in the works of Reddy and Ala (2005) and Gioannis *et al.* (2008), but low removal in the work of Kim *et al.* (2011b). Another example was citric acid, whereby it was reported to give high Pb removal from dredged marine sediment (Kim *et al.*, 2011b) but low removal for treating industrial harbor soil (Gioannis *et al.*, 2008). As the performance of wash solution is strongly soil dependent, a preliminary screening of wash solution was necessary before they were further applied in electrokinetic washing.

The screening process was conducted using shake flask desorption experiments. The tests were initiated by contacting 10 g of dry 1000 mg/kg Pb contaminated soil with 100 mL of wash solution in a conical flask. Three groups of wash solutions were investigated, namely i) strong acids (HCl, H₂SO₄, HNO₃), ii) weak acids (acetic acid, citric acid), and iii) chelating agent EDTA at a concentration of 0.001 M, 0.01 M and 0.1 M. The mixtures were homogenised using a mechanical shaker at a rotational speed of 150 rpm for 24 h under room temperature. Following that, aqueous sample was taken from each test and was filtered through 0.45 µm syringe filter to remove the suspended particles in the solution. Pb concentration in the aqueous sample was analyzed using ICP-OES after acidifying with 0.1 M HNO₃. The desorption efficiency achieved by each wash solution was determined using Equation (3.2). The best wash solution from each group that gave the highest desorption efficiency with minimum physical nuisance was selected for further application in electrokinetic washing.

Desorption Efficiency,
$$\% = \frac{\text{Mass of Pb detected in aqueous phase, mg}}{\text{Initial mass of Pb in soil, mg}} \times 100$$
 (3.2)

3.3.4 Two-stage electrokinetic washing

A mass of 250 g Pb contaminated soil was compacted into the soil column. The soil was packed slowly and an average soil voidage of 0.47 was obtained. The procedure in determining soil voidage is discussed in Appendix B5. The two ends of the column were then connected to the anode chambers and cathode chamber, with filter papers in between them to prevent soil movement into the chamber. The graphite electrode discs were further installed to the other end of the chambers, as shown in Figure 3.1.

The experiment was conducted in two stages, namely i) initial washing, and ii) electrokinetic process. Initial washing was carried out by pumping the wash solution from a reservoir into the anode chamber from the bottom of the chamber at a speed of 15mL/min using a peristaltic pump while the cathode chamber was left empty. Due to the hydraulic head gradient between the chambers, the wash solution was transported through the soil column and filled up the cathode chamber. This process not only provided soil wetting and desorption condition but also early flushing of desorbed Pb into the cathode chamber. As the overflow levels for both chambers were the same, soil

washing process attenuated with time as a result of the reduction in hydraulic head gradient. It eventually stopped when the cathode chamber was completely filled up. At this point, a constant electric potential difference (voltage) was applied across the soil for 24 h via the graphite electrodes in the chambers as a second driving force to further transport the desorbed Pb from the soil into the cathode chamber under room temperature. The electric current across the soil and effluent generation/electroosmotic flow (EOF) were observed throughout the experiment. The study for two-stage electrokinetic washing was conducted in two main sections, namely i) effect of wash solution, and ii) effect of operating parameters. The details of the experimental parameters in both sections are as shown in Table 3.1.

Effect of wash solution					
Variable		Parameter	Value		
Independent variables	i) Types of wash solutions		0.01 M NaNO ₃ *		
			0.1 M HNO ₃		
			0.1 M citric acid		
			0.01 M EDTA		
Constant variables	i)	Initial Pb concentration, mg/kg	1000		
	ii)	Experimental duration, h	24		
	iii)	Electric potential difference, V	15		
Main dependent variables	i)	Removal efficiency, %			
	ii)	Effluent generation, mL			
	iii)	System stability			
	iv)	Power consumption, kWh/kg Pb removed			
Effect of operating parameters**					
Independent variables	i)	Electric potential difference, V	7.5-30		
	ii)	Wash solution concentration, M	0.001-0.1		
	iii)	Initial Pb concentration, mg/kg	400-1000		
Constant variables	i)	Types of wash solution	Citric acid		
	ii)	Experimental duration, h	24		
Main dependent variables	i)	Removal efficiency, %			
	ii)	Effluent generation, mL			
*D.C	iii)	Power consumption, kWh/kg Pb removed			

Table 3.1: Experimental parameters for two-stage electrokinetic washing

* Reference test: Soil washing was initiated using distilled water as the wash solution. When electricity was applied, NaNO₃ was added into the anode reservoir/chamber as background electrolyte to increase the conductivity in the anode chamber for completing the electric circuit in the soil column.

** Conducted using Response Surface Methodology based on Box-Behnken Design
Experiments for normal soil washing in treating 250 g contaminated soil were conducted for removal efficiency comparisons only. The experiment for normal soil washing was conducted by repeating the procedures as discussed above (Sub-section 3.3.4), but without the application of electricity.

3.3.5 Electrokinetic soil pretreatment process

Experiments for single contaminated soil					
Variable	Parameter Value				
Independent variables	i) Types of wetting agents	0.01 M NaNO ₃			
		0.1 M citric acid			
		0.1 M EDTA			
	ii) Electrode switching time	Nil or 12 h			
Constant variables	i) Voltage gradient, V/cm	1			
	ii) Experimental duration, h	24			
	iii) Volume of wetting agent, mL	35 (Saturated with soil) *			
	iv) Initial Pb concentration, mg/kg	g 750			
Main dependent variables	i) Normalised Pb concentration a	t different soil sections, C/C_0			
	ii) Electric current generated, mA				
	iii) Soil pH				
	iv) Power consumption, kWh	Power consumption, kWh			
	v) Removal efficiency at pretreate	ed region, %			
	Experiments for co-contaminated	d soil			
Independent variables	i) Types of wetting agents	0.01 M NaNO ₃			
		0.1 M citric acid			
		0.1 M EDTA			
Constant variables	i) Voltage gradient, V/cm	1			
	ii) Experimental duration, h	24			
	iii) Electrode switching time, h	12 (Best wash solution only)			
	iv) Volume of wetting agent, mL	35 (Saturated with soil) *			
	v) Initial Pb concentration, mg/kg	$400 \text{ (mainly in PbCrO}_4 \text{ form)}$			
	vi) Initial Cr concentration, mg/kg	g 800 (Cr(VI))			
Main dependent variables	i) Normalised concentrations of I	Pb/Cr at different soil sections, C/C_0			
	ii) Electric current generated, mA	Electric current generated, mA			
	iii) Soil pH				
	iv) Power consumption, kWh				
	v) Removal efficiency at pretreated region, %				

 Table 3.2: Experimental parameters for electrokinetic soil pretreatment process

* Preliminary test was conducted to determine the maximum amount of wetting agent that can be uphold by the soil a.k.a soil saturation. Wetting agent was slowly added into a specific amount of soil until the formation of liquid layer on the soil surface. In the soil studied, 35mL of wetting agent is needed to saturate the soil. Further addition of wetting agent will form a liquid layer on the top of the soil.

The equipment as discussed in Sub-section 3.2.2 was used in this study. A mass of 100 g contaminated soil was compacted into the sand box. Three pairs of electrodes (diameter: 0.7 cm; length: 4 cm) were directly injected into the soil and were connected to a direct current supply. The system configuration is as shown in Figure 3.3. Then, the wash solution, a.k.a wetting agent in this study, was sprinkled from the top of the soil. When the soil was saturated with the wetting agent, constant voltage gradient of 1 V/cm (7.5 V) was applied for 24 h under room temperature. During the experiment, 1-2 mL of distilled water was added from time to time to maintain the soil saturation condition and to compensate water losses via evaporation and electrolysis. The electric current across the soil was closely monitored throughout the experiment. For approaching electrode tests, the anode (approaching anode) or cathode (approaching cathode), depending on the direction of heavy metal migration, was switched to the middle electrode (Figure 3.3) at the 12th hour of the experiment and the electric potential difference was adjusted to maintain a voltage gradient of 1 V/cm (3.75 V). The study was conducted for two types of contaminated soils, namely i) single Pb contaminated soil, and ii) Pb and Cr cocontaminated soil. The details of the experimental conditions for each study are as shown in Table 3.2.

3.3.6 Response surface methodology

Response Surface Methodology (RSM) was used to evaluate the effects of operating parameters in the study. RSM is a collection of mathematical and statistical techniques that are useful for the modelling and analysis of problems for which a response on outcome is influenced by several variables (Montgomery, 2001). The use of RSM and Analysis of Variance (ANOVA) could provide an approximation for the true functional relationship between the set of independent variables and the responses in the form of a polynomial equation as shown in Equation (3.3), where y is the response, β is the regression coefficient, x is the independent parameters (Montgomery, 2001). The equation is useful to navigate through the effects of the independent variables on the responses not only from the aspect of individual effect but also the interaction effects between these parameters.

$$y = \beta_0 + \sum_{i=1}^3 \beta_i x_i + \sum_{i=1}^3 \beta_{ii} x_i^2 + \sum_{i=1}^2 \sum_{j=i+1}^3 \beta_{ij} x_i x_j$$
(3.3)

In this study, RSM based on Box-Behnken Design (BBD) was conducted by the aid of Design Expert software (Design Expert 6.0.1). BBD is a spherical three level design by combining 2^k factorial design with incomplete block design. It is most efficient in terms of the number of required runs in comparison to Central Composite Design as this design requires fewer numbers of experiments (Montgomery, 2001). The details in number of experiments required and the combination level on the parameters are discussed in Sub-section 4.4.4. Based on ANOVA analysis, the numerical models generated from the analysis were used to analyse the importance of the parameters and their interaction effects.

3.4 Analytical methods

In general, the analysis could be categorised into three main sections, which are: i) electrical properties, ii) properties in soil phase and iii) supporting analysis. The details for the analytical procedures involved in each section were discussed in the following sub-sections.

3.4.1 Electric current

The electric current that passed through the soil was measured by connecting the ammeter (Multimeter Sunwa TE-832B) in series between the direct current supply and the anode/cathode. Three reading were taken and the average value for the electric current was used in results analysis. The current density was calculated using Equation (3.4).

Current density =
$$\frac{\text{Electric current detected, mA}}{\text{Cross sectional area for the soil column, cm}^2}$$
 (3.4)

3.4.2 Soil properties

After the experiment, the soil was divided into several sections, i.e. five sections for two-stage electrokinetic washing and four sections for electrokinetic soil pretreatment study and they were dried before the analysis. The pH and metal concentration (Pb and Cr) in the soil at each soil section were analysed using different methods.

3.4.2.1 Soil pH

Soil pH was determined using USEPA SW-846 Method 9045D at room temperature. The soil sample was mixed with distilled water at a soil: water ratio of 1 g : 2.5 mL. The suspension was homogenised for five minutes. Then, the mixture was left undisturbed for 1 h to settle the suspended particles. The pH of the supernatant was then determined by CRISON Multimeter MM26+ at room temperature.

3.4.2.2 Pb concentration in soil

Two types of Pb species were investigated in this study, namely: i) metal oxides adsorbed/bounded Pb (Pb(NO₃)₂) and ii) water insoluble Pb (PbCrO₄). For the former, KSTM method (Kim *et al.*, 2005b) was used as it gives similar high recovery as USEPA 3050b method at faster extraction time with simpler procedure. This method was discussed in detail by Kim *et al.* (2005b) whereby the soil was dried at 105 °C. Following this, a mass of 5 g dry soil was contacted with 50 mL 0.1 M HCl in a conical flask and the mixture was agitated at a rotational speed of 100 rpm under 30 °C for 1 h. The supernatant was then filtered and was analyzed using ICP-OES.

For water insoluble Pb species, USEPA 3050b method was used. This method involved acid digestion using HNO₃, H₂O₂ and HCl. The digestion was conducted in a fume hood. 1 g of soil sample was placed in a digestion vessel. Then, 10 mL of HNO₃ solution at a HNO₃: distilled water ratio of 1:1 was added and the vessel was covered with a watch glass. The mixture was then heated to 95 ± 5 °C on a hot plate and was refluxed for 10-15 minutes without boiling. Following this, the sample was allowed to cool down and 5 mL of concentrated HNO₃ was added into the vessel. The mixture was again refluxed for 30 minutes. This step was repeated until no brown fume was given off by the sample. The solution was then evaporated to approximately 5 mL and was cooled down.

Digestion was continued by adding 2 mL of water and 3 mL of 30% H₂O₂ into the sample. The vessel was then covered with a watch glass and the sample was warmed for peroxide reaction. Warming was continued until effervescence subsided and the vessel was cooled down. These steps were repeated by adding 1 mL of 30% H₂O₂ into the sample until the general appearance of the sample was unchanged. Following this, the solution was again being reduced to approximately 5 mL by evaporation. Finally, 10 mL of concentrated HCl was added to the sample and the mixture was heated at 95 ± 5 °C at reflux condition for 15 minutes with a watch glass cover. Then, the digestate was cooled down and was filtered through a filter paper and the supernatant was collected in a 100 mL volumetric flask. The volume of the supernatant was adjusted to 100 mL and the Pb concentration in the supernatant sample was analysed by ICP-OES. Pb concentration in the soil can be calculated based on Equation (3.5), where C_{ICP} is the Pb concentration in the supernatant sample detected by ICP (mg/L), *DF* is the dilution factor, V_{ES} is the volume of extraction solution/supernatant after the digestion (L), and m_s is the mass of soil sample used (kg). The Pb removal efficiency achieved by the system and normalised concentration for Pb were further determined using Equations (3.6) and (3.7), respectively, where C is the Pb concentration left in the soil after the experiment (mg/kg) and C₀ is the initial Pb concentration in the soil (mg/kg).

Pb concentration in soil, C, mg/kg =
$$\frac{C_{ICP} (DF) (V_{ES})}{m_s}$$
 (3.5)

Removal efficiency,
$$\% = \frac{C_0 - C}{C_0} \times 100$$
 (3.6)

Normalised concentration =
$$\frac{C}{C_0}$$
 (3.7)

3.4.2.3 Cr concentration in soil

Cr concentration in the soil sample was determined from two aspects: i) Cr(total), and ii) Cr(VI). Cr(total) in the soil, which consisted of Cr(III) and Cr(VI) was determined from the aqueous sample obtained from acid digestion USEPA 3050b via ICP analysis, as discussed in Sub-section 3.4.2.2. The removal efficiency and normalised concentration of Cr at different soil sections was determined using Equations (3.6) and (3.7), respectively. Cr(VI) concentration in the soil after the experiment was determined using alkaline digestion as specified in USEPA 3060A. A mass of 2.5 g soil was mixed with 50mL of digestion solution containing 20 g/L NaOH and 30 g/L Na₂CO₃. The mixture was further mixed with approximately 400 mg of MgCl₂ and 0.5 mL of 1.0 M KH₂PO₄/K₂HPO₄ buffer. Following this, the mixture was stirred for five minutes before it was heated to 90-95 °C for 60 minutes with continuous stirring. After the heating process, the mixture was gradually cooled down to room temperature under continuous stirring. The aliquot was collected by filtering the mixture through a filter paper. The pH of the aqueous sample was further adjusted to 2-3 using HNO₃ before it was diluted to 100 mL in a volumetric flask. The Cr(VI) concentration in the aqueous sample was then determined by ICP-OES.

3.4.3 Pb deposition efficiency

Deposition efficiency can be defined as the percentage of Pb that was recovered as solid deposit on cathode from the contaminated soil. This was analysed for two-stage electrokinetic washing process whereby the deposit on the cathode surface was taken and dissolved in 150 mL of 0.1 M HNO₃ in a conical flask for 24 h under vigorous stirring. The cathode was also immersed in another beaker with 150 mL of 0.1 M HNO₃ for 24 h to dissolve the deposit that was left over on the cathode surface. Following this, both of the aqueous samples were diluted and analysed for Pb concentration using ICP-OES to identify the amount of Pb that was deposited on the cathode. Deposition efficiency was calculated using Equation (3.8), where V_{HNO3} is the volume of HNO₃ solution used (L) and m_s is the mass of soil used in the experiments (kg).

Deposition Efficiency, % =
$$\frac{\sum C_{ICP} (DF) V_{HNO_3}}{m_s C_0} \times 100$$
 (3.8)

3.4.4 Aqueous properties

The properties in aqueous phase were evaluated only for two-stage electrokinetic washing as electrokinetic soil pretreatment study did not involve wash solution chamber. The wash solution was removed from the chambers after the experiments. Aqueous samples from effluent, wash solution reservoir, anode chamber, and cathode chamber were collected for further analysis for the purpose of supporting the results only. The main properties monitored were pH and Pb concentration. The pH was determined by contacting a calibrated CRISON Multimeter MM26+ directly with the aqueous samples collected. Pb concentrations in the samples mentioned above were also determined via ICP-OES after diluting with 0.1 M HNO₃ solution.

3.5 Precautious steps for chemicals handling

The precautionary steps taken in the handling of chemicals and solutions are as follow:

- i. Mask and gloves were worn during the preparation of acids and other chemical solutions.
- ii. The acid solutions were prepared in fume cupboard.
- The used glassware and experimental apparatus were cleaned by soaking with detergent and rinsing with water. They were then soaked in 2 M HNO₃ for 1 h. Then, another rinsing was carried out using distilled water. Following this, they were dried before used.
- iv. All the chemicals were stored in a dark place to prevent any possible degradation caused by light.

3.6 Minimisation of error

The analytical instrument such as pH meter and ICP-OES were calibrated before each experiment to minimise systematic error. The experiments and soil analysis were at least duplicated to validate the results and the average values with standard errors were reported in this thesis. In addition, readings for pH and metal concentrations were triplicated during the analysis and average readings were taken with an error of within $\pm 4\%$.

university

CHAPTER 4: RESULTS AND DISCUSSION

4.1 Study outlines

The study consists of two major sections, namely, i) characterisation study for the soil, and ii) incorporation of electrokinetic process in soil remediation methods, as shown in Figure 4.1. The first major section (Section 4.2) is devoted mainly to the properties of the soil, the contamination mechanism as well as the maximum Pb contamination level for the soil. Then, a study on screening of the wash solutions for desorbing Pb from the soil was carried out in order to determine the most efficient wash solution from different groups of chemical agents (Section 4.3). These solutions were further utilised as chemical agents/wash solutions in electrokinetic soil remediation process, which is the second major section of this study (Sections 4.4 and 4.5).



Figure 4.1: The outline for Chapter 4: Results and Discussion

The study on the incorporation of electrokinetic process in soil remediation techniques consists of two parts: i) incorporation of electrokinetic process in soil washing as a two-stage electrokinetic washing process, and ii) incorporation of electrokinetic process as a soil pretreatment method to concentrate heavy metals and reduce contaminated soil volume. The details of these studies are further elaborated in their respective sections (Sections 4.4 and 4.5).

4.2 Soil characterisation

4.2.1 Soil properties

Soil properties	Value	Methods
pH	3.97	USEPA SW-846 Method 9045D
Specific gravity	2.5	ASTM D 854 - Water pycnometer method
Cation exchange capacity (meq/100g)	5.1	Ammonium acetate method
Organic matter content, % LOI	1.4%	Loss of weight on ignition
Soil metal content, (mg/kg)		Methods
Iron, Fe (mg/kg)	3719	
Aluminium, Al (mg/kg)	2400	
Magnesium, Mg (mg/kg)	635	USEPA 3050b (Acid digestion). Aqueous
Manganese, Mn (mg/kg)	185	samples obtained from the acid digestion were analyzed using ICP-OES
Zinc, Zn (mg/kg)	18	
Lead, Pb (mg/kg)	11	

Table 4.1: Characteristic of the natural soil used in this study

The soil used in this work was taken from Hulu Langat, Malaysia. The properties of the natural soil such as soil pH, specific gravity, cation exchange capacity, organic matter content and metal content were determined using standard methods after passing through a sieve with mesh number of 20. Table 4.1 shows that the sandy soil had a pH of 3.97 with cation exchange capacity and organic matter content of 5.1 meq/100g and 1.4%, respectively. The soil had high iron and aluminium content, which served as good adsorbent to entrap heavy metals (Violante *et al.*, 2007).



Figure 4.2: FTIR spectrum for the uncontaminated soil



Figure 4.3: XRD spectrum for the uncontaminated soil

Studies of Fourier Transform Infrared Spectroscopy (FTIR) and X-Ray Diffraction (XRD) on the soil further revealed the metal species available on the soil. The FTIR spectrum for the soil as shown in Figure 4.2 indicates that iron oxides in the soil are found at 464.64 cm⁻¹, 529.23 cm⁻¹, 690.32 cm⁻¹ and 781.95 cm⁻¹, which correspond to Fe-O stretch peaks (Rubim *et al.*, 2001; Sahoo *et al.*, 2010; Avram *et al.*,

2011). The XRD spectrum as shown in Figure 4.3 further suggests that the iron oxide species is maghemite (Isometric $Fe_{21.333}O_{32}$, XRD displacement 0.001), with the presence of other metal species such as silicon dioxide (SiO₂, XRD displacement 0.158), aluminosilicate such as feldspar albite (AlNaO₈Si₃, XRD displacement -0.053), mica ((Mg,Al)₆(Si,Al)₄O₁₀(OH)₈, XRD displacement 0.119), and sylvine, sodian (Cl₁K_{0.9}Na_{0.1}, XRD displacement -0.171).

4.2.2 Degree of contamination and the mechanisms

The degree of contamination for Pb in the soil studied was determined. Preliminary study shows that Pb was favoured as an adsorbate on soil surface at a solution pH of 5 in comparison to lower pH values (Ng et al., 2016). On the other hand, precipitation of Pb(OH)₂ occurred at pH 6 and above, and this was also reported in the work of Niu et al. (2013). Thus, maximum Pb contamination level for this soil was investigated at solution pH 5 via adsorption capacity test. The change in Pb concentration in soil with time at pH 5 is as shown in Figure 4.4. The figure reveals that rapid adsorption occurs at initial stage once the solution is in contact with the soil and equilibrium is achieved in less than 1 h. Similar equilibrium time was also observed when iron oxides and maghemite nanotubes were applied as adsorbents for removing Pb (Lai and Chen, 2001; Boujelben et al., 2009; Mohapatra et al., 2009; Cheng et al., 2012; Ahmedzeki, 2013), Cd (Mohapatra et al., 2009; Ahmedzeki, 2013), Zn (Mohapatra et al., 2009), Cu, Ni and Cr (Hu et al., 2006). Fast adsorption rate could be mainly due to external surface adsorption mechanism as the adsorption sites were available on the soil surface (Hu et al., 2006). The adsorption data could be fitted into pseudo second order rate equation (Equation 4.1) with an R^2 value of 0.9969 whilst it failed to fit into other kinetic models. This kinetic model was also in line with other research publications on iron coated sand (Mohapatra et al., 2009; Ahmedzeki, 2013) and maghemite nanotubes

(Roy and Bhattacharya, 2012) for Pb adsorption. From this model, the adsorption capacity/maximum Pb contamination level for the soil was determined as 1000mg/kg, as shown in Figure 4.4.



$$\frac{t}{q_t} = \frac{1}{V_0} + \frac{1}{q_e}t$$
(4.1)

Figure 4.4: Change in Pb concentration in soil with time at solution pH 5 and the curve fitting using pseudo second order kinetic model

The contamination mechanism of Pb was investigated by fitting the experimental data to different isotherm models. Table 4.2 shows that Langmuir isotherm has the best fit with R^2 value of 0.9954 while the maximum adsorption capacity, Q_{max} was determined as 909.091mg/kg. Several studies also reported that the adsorption of Pb on iron oxides coated sand and maghemite nanotubes followed Langmuir adsorption isotherm (Lai and Chen, 2001; Boujelben *et al.*, 2009; Mohapatra *et al.*, 2009; Roy and Bhattacharya, 2012). Langmuir isotherm is valid only for monolayer adsorption on a surface containing finite number of identical sites without transmigration of adsorbate in the plane of the surface (Dada *et al.*, 2012). Thus, it is suggested that Pb adsorption on this soil conforms to another published work that proposed monolayer adsorption of Pb on the iron coated sand (Mohapatra *et al.*, 2009)

and maghemite nanotubes (Roy and Bhattacharya, 2012). In this study, significant presence of iron and other minerals such as aluminosilicate in the soil served as binding sites for Pb adsorption on the soil particles and no further binding was detected after saturation, indicating the occurrence of monolayer adsorption process.

	-			-			-	
Isotherm	Slope	Intercept	\mathbf{R}^2	Q _{max}	K _L	n	K _F	β
Langmuir	0.0011	0.0047	0.9954	909.091	0.234	-	-	-
Freundlich	0.2058	2.5791	0.9876	-	-	4.859	13.185	-
Dubinin- Radushkevich	7.00E-07	6.6726	0.8887	790.4481	-	-		7.00E-07

Table 4.2: Adsorption isotherms for Pb adsorption on soil under solution pH 5

It is believed that the type of soil used in this work poses the risk for Pb contamination, depending on the surrounding activities as Pb could be adsorbed on the soil surface easily when Pb solution is in contact with the soil. The adsorption process takes place through monolayer adsorption on the iron (maghemite) and other minerals on the soil surface such as aluminosilicates during the contact. This was further justified through the study of Pb speciation in the soil using Tessier's Method (Tessier *et al.*, 1979; Gworek and Mocek, 2003; Venkateswaran *et al.*, 2007) whereby the results showed that >96% of the Pb in the soil was in Fe-Mn oxide bounded, carbonate bounded and exchangeable forms whilst the Pb species in organic matter bounded and residual form was low and negligible. Maximum Pb contamination level for this soil was determined as 900-1000 mg/kg, which is much higher than the maximum threshold concentration of 400mg/kg in Malaysia (DoE-Malaysia, 2009). Thus, it is taken as soil for investigation.

4.3 Screening of wash solutions



Figure 4.5: Desorption efficiency of Pb from soil using different types of wash solutions

Performance of wash solution was evaluated from Pb desorption efficiency from the soil by contacting excess volume of wash solution (100 mL) with artificial Pb contaminated soil (10 g). The wash solutions used in this study were HCl, HNO₃, H₂SO₄, citric acid, acetic acid and EDTA. Figure 4.5 shows that EDTA could desorb nearly all Pb from the soil effectively regardless of EDTA concentration from 0.001-0.1 M. The use of high EDTA concentration was not necessary as an EDTA: Pb mole ratio of ≈ 2 (at 0.001 M EDTA concentration) was adequate to completely removed Pb from the soil. On the other hand, the acids also showed positive results. However, the desorption efficiency was highly dependent on the acid concentration. Generally, higher acid concentration gave higher desorption efficiency as it provided lower pH condition, which favoured Pb desorption/dissolution (Carlon *et al.*, 2004). In terms of the nature of acids, strong acids such as HCl and HNO₃ showed better performance than weaker citric acid and acetic acid. However, as the acid concentration increased to 0.1 M, complete desorption was achieved for all types of acids. The acids not only enhanced Pb desorption but also served as complexation agents (citrate, acetate, chloride) for enhancing Pb solubility in aqueous phase (Chung and Kang, 1999; Kim *et al.*, 2011b). Nevertheless, an exception is observed for H_2SO_4 whereby the increase in acid concentration from 0.001 M to 0.01 M and 0.1 M reduces the desorption efficiency from 63.87% to 14.39% and 11.94%, respectively, as shown in Figure 4.5. This was mainly due to the formation of water insoluble PbSO₄ at high H_2SO_4 concentration (Reger *et al.*, 2010).

Wash solutions were screened for further application in electrokinetic washing based on the highest desorption efficiencies achieved. Generally, 0.1 M HNO₃, 0.1 M citric acid, and 0.01 M EDTA showed highest desorption efficiencies among the chemical agents. In contrast, other chemical agents showed disadvantages not only in terms of desorption efficiency but also in other operational aspects. Strong acids such as HCl and H₂SO₄ were not selected as HCl could produce harmful chlorine gas (Cl₂) in the electrokinetic process (Giannis and Gidarakos, 2005) while H₂SO₄ was poor in Pb desorption. On the other hand, among the organic acids, acetic acid was not considered as it not only did not show superior desorption efficiency than citric acid but also caused strong odour to the soil during the experiments and this is not desirable for large scale treatment. Hence, 0.1 M HNO₃, 0.1 M citric acid and 0.01 M EDTA were selected for further study in electrokinetic washing process.

4.4 Two-stage electrokinetic washing

4.4.1 Introduction

Two-stage electrokinetic washing is a configuration where soil washing and electrokinetic process are operated at two different stages in a single equipment. Sub-section 2.3.3.3 suggests that the use of electrokinetic process after soil washing could further enhance the removal efficiency of heavy metals to below regulatory concentration. However, this generates a huge amount of spent wash solution in comparison to both normal soil washing as well as normal electrokinetic process (Kim *et al.*, 2009d). In contrast, two-stage electrokinetic washing that utilises spent wash solution as the catholyte in cathode chamber could reduce wash solution consumption while maintaining the removal efficiency for electrokinetic washing.

Figure 4.6 illustrates the process flow diagram of a two-stage electrokinetic washing for Pb removal. This process is carried out in two stages: i) initial soil washing, followed by ii) electrokinetic process. Soil washing is initiated by filling up the anode chamber with the wash solution while the cathode chamber is left empty. Due to the hydraulic gradient between the two chambers, the solution diffuses into the soil, wets the soil, desorbs and transports Pb into the empty cathode chamber. Soil washing stops when the cathode chamber is fully filled up. At this stage, electrokinetic process is introduced as the second stage of the treatment, which serves as another driving force to further transport the remaining desorbed Pb towards the cathode chamber, as shown in Figure 4.6. The desorption process can also be enhanced as electromigration generally reduces Pb concentration polarization at soil: solution interface in the absence of pore flow. Furthermore, as the spent wash solution is utilised as catholyte, the amount of wash solution used in the process is reduced in comparison to the electrokinetic washing as discussed in Section 2.3.3.



Figure 4.6: Schematic diagram of two-stage electrokinetic washing process

4.4.2 Efficiency comparison



Figure 4.7: Comparison on Pb removal efficiency between soil washing and two-stage electrokinetic washing

Figure 4.7 illustrates that two-stage electrokinetic washing provided higher Pb removal efficiency than normal soil washing whereby an enhancement of 4.98-20.45% was obtained. The use of electricity across the soil provided electromigration and electroosmosis as the driving forces to remove Pb from the soil to the cathode chamber. In addition, the removal of desorbed Pb from the soil by electromigration also prevented local concentration polarisation and maintained Pb desorption from soil: solution interface, which further aided the removal process. However, the enhancement presented by the two-stage electrokinetic washing is wash solution dependent, as shown in Figure 4.7. Electrokinetic washing generally provided higher enhancement when NaNO₃ and citric acid were used as the wash solution. In contrast, the enhancements were relatively less significant in the tests that utilized HNO₃ and EDTA.

4.4.3 Effect of wash solution on two-stage electrokinetic washing

The type of wash solutions used was found to significantly affect the performance of two-stage electrokinetic washing, as discussed in Sub-section 4.4.2. Thus, the effect of wash solutions on the process was further evaluated not only for removal efficiency but also for other aspects such as effluent generation, stability of the system and power consumption.

4.4.3.1 Removal efficiency

Figure 4.7 shows that the removal efficiencies for two-stage electrokinetic washing follows a trend of $HNO_3 > citric acid > NaNO_3 > EDTA$. This trend is in line with the order of soil pH, as shown in Figure 4.8a whereby lower soil pH provides higher removal efficiency. Compared to non-acidic NaNO₃, low soil pH of \approx 2.5 created

by HNO₃ and citric acid was found to enhance Pb desorption from the soil (Yang *et al.*, 2006). This not only increased Pb removal via soil washing but also provided better Pb mobility for electromigration in soil column, resulting in higher removal efficiency. Among the acids, HNO₃ which is a strong acid showed higher removal efficiency than weaker citric acids in soil washing.



Figure 4.8: Properties of two-stage electrokinetic washing process using different types of wash solutions: a) soil pH; b) current density across the soil

Other than soil pH, the current density across the soil from anode to cathode is another factor that affected Pb transport in electrokinetic process, especially for acid based agents. Figure 4.8b shows that the current density across the soil is generally stable with no significant drop during the experiments. This is perhaps due to the relatively short experimental time in comparison to other studies (Amrate *et al.*, 2005; Kim et al., 2011b). The magnitude of current density in the process was governed by the ionic strength and desorption efficiency provided by the wash solutions, in the order of HNO₃> citric acid > EDTA > NaNO₃. Figure 4.8b shows that the acids have higher current density than EDTA and NaNO₃ as they are in higher concentration and they can enhance metal desorption/dissolution by providing low pH condition (Mellis et al., 2012). This in turn increased dissolved ions concentration and reduced electrical resistance in the overall system. In terms of removal efficiency, Figure 4.7 shows that HNO₃ is found to provide nearly complete Pb removal, as a result of its high desorption capacity for metals and low electrical resistance for electromigration. On the other hand, a complete removal was not achieved by weaker citric acid. Instead, it enhances the removal efficiency from 70.39% to 90.84%, as shown in Figure 4.7. However, it is worth noting that similar efficiency enhancement of 20.4% is also observed when NaNO₃ is used as a wash solution even though it has lower current density, as shown in Figures 4.7 and 4.8b. Nevertheless, a relatively higher amount of effluent at 492.9mL was generated by NaNO₃ via electroosmosis than that for citric acid (65.9mL) and this could be one of the reasons for high removal efficiency enhancement in NaNO₃ test. The effect of wash solution on effluent generation is further elaborated in Sub-section 4.4.3.2.

On the other hand, a different trend was observed when 0.01 M EDTA was applied. Figure 4.7 shows that EDTA has the lowest removal efficiency in a two-stage electrokinetic washing as a result of lower enhancement by electrokinetic process. Unlike NaNO₃, HNO₃ and citric acid, the main mechanism for Pb removal in EDTA test was water soluble complex formation as Pb desorption and mobility was generally poor at a high soil pH of 7-8 (Figure 4.8a) due to possible Pb(OH)₂ precipitation (Niu *et al.*, 2013). The presence of iron and other mineral content in the soil may compete with Pb for EDTA complex formation, which may reduce the availability of EDTA for Pb to form complex (Kim *et al.*, 2011b). This phenomenon may cause lower Pb removal in comparison to acidic based solutions. In addition, anionic Pb-EDTA complexes were formed at this pH (Niinae *et al.*, 2008; Zhang *et al.*, 2014) and their migration towards the anode in electrokinetic process may also cause low enhancement in removal efficiency. This is observed from the analysis of Pb residual in the soil sections, as shown in Figure 4.9d.

Figure 4.9 shows that, in general, lower Pb residual is observed in any soil section in a two-stage electrokinetic washing in comparison to the tests that utilised soil washing only due to the removal enhancement by electromigration and electroosmosis. This was true for all wash solutions in this study except for EDTA test where more Pb residual was detected near the anode region (normalised distance from anode chamber <0.5) in the two-stage electrokinetic washing than in soil washing. Unlike soil washing, which showed one way Pb transport, a higher Pb concentration near anode region in two-stage electrokinetic washing indicated the transport of Pb-EDTA complexes towards anode. This was further justified where Pb concentration of 3.476 mg/L was detected in anode chamber whilst this was not detected in NaNO₃, HNO₃, and citric acid tests. Counter migration of anionic Pb-EDTA towards the anode chamber could be the main reason for the delay in Pb removal from the soil and this eventually showed low enhancement in removal efficiency. This trend was also reported by Yang and Lin (1998). Nevertheless, it is worth noting that majority of Pb removed was still accumulated in the cathode chamber in comparison to anode chamber at the end of the

experiment. The electromigration of anionic Pb-EDTA complexes towards the anode may be slowed down by the electroosmosis, which was in opposite direction, as reported in the work of Villen-Guzman *et al.* (2015).



Figure 4.9: Normalised Pb concentration at different soil sections after the experiments using different types of wash solutions: a) 0.01 M NaNO₃, b) 0.1 M HNO₃, c) 0.1 M citric acid, d) 0.01 M EDTA.



Figure 4.9, continued'

4.4.3.2 Effluent generation

Application of electrokinetic process was found to cause extra effluent generation due to the presence of electroosmosis. The extra effluent generated is collected in an effluent collecter, as shown in Figure 3.1. Figure 4.10 illustrates that the amount of effluent generated under a constant electric potential difference of 15V follows a trend of 0.01 M NaNO₃ > 0.1 M HNO₃ > 0.1 M citric acid > 0.01 M EDTA.



Figure 4.10: Volume of effluent generated under different types of wash solutions

Figure 4.10 illustrates that the volume of effluent generated via electrosmosis is dependent on the types of wash solutions used. It was found that NaNO₃ generated the highest amount of effluent at 492.9 mL. This could be due to this test which utilised 0.01 M NaNO₃ as wash solution showed a higher average soil pH at 4.2 (Figure 4.8a) compared to HNO₃ and citric acid. Higher soil pH condition resulted in higher negativity of soil zeta potential (Acar and Alshawabkeh, 1993; Reddy and Saichek, 2003; Baek *et al.*, 2009; Lee *et al.*, 2009). According to Helmholtz-Smoluchowski theory, as shown in Equation (2.4) (Lee *et al.*, 2009; Asadi *et al.*, 2013), higher effluent generation was expected as electroosmosis was proportional to the negativity of soil zeta potential. Besides that, the use of NaNO₃ as electrolyte was also reported to favour electroosmotic flow (Alcántara *et al.*, 2008b; as cited in Alcántara *et al.*, 2012).

On the other hand, 0.1 M HNO₃ also generated high effluent volume via electroosmosis at 288.5 mL during the experiment. However, it was worth noting that HNO₃ test showed lower effluent generation than NaNO₃ test even though it recorded higher electric current. This was probably caused by relatively low soil pH condition at about \approx 2.5 in HNO₃ test, as low pH was reported to reduce and even reverse the

direction of electroosmosis in electrokinetic soil treatment (Zhou *et al.*, 2004; Zhou *et al.*, 2005; Baek *et al.*, 2009) as a result of reduction in negativity of soil zeta potential. Thus, according to Helmholtz-Smoluchowski theory (Lee *et al.*, 2009; Asadi *et al.*, 2013), electroosmotic force was expected to be smaller than that in NaNO₃.

Similar reason could be applied for citric acid test. In comparison to NaNO₃ and HNO₃, 0.1 M citric acid shows relatively low effluent generation at 65.9 mL, as shown in Figure 4.10. This was attributed to the low electric current and low pH condition. As citric acid provided similar low soil pH as in HNO₃ test, it showed weaker electroosmosis than NaNO₃. Under similar soil pH condition, citric acid only provided about 10% of current density as in HNO₃. As electroosmosis was found to be higher at higher current density (Yang and Long, 1999; Kim et al., 2005a), lower amount of effluent was collected in citric acid test. In contrast, lowest volume of effluent is collected at 39.8 mL when 0.01 M EDTA is employed as the wash solution even though it has high soil pH, as shown in Figure 4.10. This trend was in agreement with other sandy soil treatments using electrokinetic process where EDTA showed lower electroosmosis rate than citric acid (Kim et al., 2008a, 2008b). This could be due to electromigration of anionic EDTA complexes in opposite direction to electroosmosis/cation transport, which may compress electrical double layer, as suggested by Colaccico et al. (2010). Thus, lower electroosmosis rate was observed.

The presence of electroosmosis may affect the removal enhancement given by electrokinetic process. An example is given for NaNO₃ and citric acid. Figure 4.7 shows that two-stage electrokinetic washing provides similar Pb removal enhancement of \approx 20% when both NaNO₃ and citric acid are used as the wash solutions, even though citric acid has higher Pb desorption efficiency from soil washing and lower resistance for electromigration. In this case, electroosmosis flow of 492.9 mL (\approx 5.6 pore volume) in NaNO₃ test may provide additional flushing effect in the system in which 43% of Pb

was removed out of the cathode chamber/system via effluent in comparison to citric acid test, which only showed 2% removal via 65.9mL of effluent. From this observation, it can be suggested that both electromigration and electroosmosis were responsible in enhancing Pb removal efficiency in two-stage electrokinetic washing, especially when 0.01 M NaNO₃ was applied. In contrast, for citric acid, similar enhancement in removal efficiency could be achieved at lower effluent generation, as discussed in Sub-section 4.4.3.1. This mechanism will be considered as one of the criteria for selection of the best wash solution for the two-stage electrokinetic washing as the effect of electroosmosis could be controlled by the current intensity and the nature of wash solution (Murillo-Rivera *et al.*, 2010).

4.4.3.3 System stability – anode chamber



System stability in anode chamber during two-stage electrokinetic washing was analysed from two aspects, namely anode chamber pH and physical appearance of anode. Figure 4.11 reveals that the application of electricity caused a decrease in anode chamber pH from 6.85 to 2.95 and 8.85 to 7.05 when NaNO₃ and EDTA were used, respectively. The pH drop in anode chamber was mainly due to electrolysis of water at anode which generated H^+ ions, as explained by Equation (2.1) (Acar and Alshawabkeh, 1993). It was also found that the pH drop was slower in EDTA test as EDTA solution which had an initial pH of \approx 9 acted as a buffer solution. Among them, the electrolysis and acid front in NaNO₃ test provided slight acidification for soil in the anode region (normalised distance from anode: 0.1-0.5), reducing the pH from an average of 4.71 to 4.09. In contrast, stable low pH values are observed for acidic wash solutions, as shown in Figure 4.11. The increase in H⁺ concentration in anode chamber via electrolysis showed less variation in pH as the H⁺ content in the chamber was already at a high concentration.



Figure 4.12: Physical appearance of graphite anode in 0.1 M HNO₃ test: (a) before the experiment; (b) after the experiment; (c) carbon particles collected in anode chamber

As for electrode stability, it was found that graphite anode was stable throughout the experiments when NaNO₃, citric acid and EDTA were used as the wash solution. However, when 0.1 M HNO₃ was used, corrosion of graphite anode was observed. Figure 4.12b shows that in contrast to the original anode (Figure 4.12a), the used anode had corroded. Carbon particles were also collected in the wash solution in the anode chamber after the experiment, as shown in Figure 4.12c. The carbonaceous particles could be graphite oxides as they could be produced in the presence of HNO₃ under strong oxidizing condition (Brodie, 1859, as cited in Hummers and Offeman, 1958), which was achieved in the anode chamber. Thus, it was suggested that HNO₃ was unsuitable as the wash solution in electrokinetic washing when graphite was used as an electrode.

4.4.3.4 System stability – cathode chamber

Similar evaluation for system stability was conducted for cathode chamber. Figure 4.13 shows that the pH in the cathode chamber increases from 4.18 to 9.85 and 4.87 to 11.93 when NaNO₃ and EDTA are used, respectively. Wash solution pH in the cathode chamber was reduced to a range of 4-5 after soil washing process as the soil was acidic. When electric current was induced, water was electrolysed in the cathode chamber and OH⁻ was produced, as shown in Equation (2.2) (Acar and Alshawabkeh, 1993). Thus, the pH in cathode chamber increased. It was observed that pH rise in EDTA test was faster as it had relatively higher current and lower EOF in comparison to NaNO₃. The high cathode chamber's pH in this test eventually increased the soil pH in the cathode region (normalised distance from anode: 0.5-0.9) from 5.34 to 7.07.



Figure 4.13: pH of wash solutions in cathode chamber at different time

In contrast, the use of citric acid maintains low pH in cathode chamber close to its initial pH at 2, as shown in Figure 4.13. Due to its acidic characteristic, citric acid served as a buffer solution to prevent pH rise in the cathode chamber throughout the experiment. However, different behaviour was noticed for HNO₃. Figure 4.13 reveals that HNO₃ provides a pH of < 2 in cathode chamber until the 6th hour. Then, a rapid increase in pH occurred and a maximum pH of 8.28 was achieved at the 10th hour. This unusual trend was also noticed by Almeira O *et al.* (2012) when HNO₃ concentration was higher than 0.12M. It was suggested that under high HNO₃ concentration, NO₃⁻ in cathode chamber underwent reduction and more OH⁻ ions were produced, as shown in Equations (4.2-4.5) (Almeira O *et al.*, 2012). Excessive OH⁻ prevailed over the H⁺ ions in the cathode chamber, which increased the chamber's pH. However, it was worth noting that the pH was again decreased after 10th hour and eventually reduced to 1.41, as a result of electroosmosis and acid front after the consumption of NO₃⁻ in the cathode chamber.

$$NO_{3}^{-} + 2e^{-} + H_{2}O \rightarrow NO_{2}^{-} + 2OH^{-}$$
(4.2)

$$2NO_{3}^{-} + 16e^{-} + 12H_{2}O \rightarrow 2NH_{3} + 18OH^{-}$$
(4.3)

$$2NO_{3}^{-} + 10e^{-} + 6H_{2}O \rightarrow N_{2} + 2OH^{-}$$
(4.4)

$$NO_3^- + 7H^+ \rightarrow NH_4 + 3OH^- \tag{4.5}$$

With regard to physical appearance of the cathode, no corrosion on graphite cathode was observed. Instead, Pb and other metal ions were found to deposit on cathode surface during electrokinetic washing process. This phenomenon, known as electrodeposition, provided Pb recovery as solid from the spent wash solution in the cathode chamber. The electrodeposition was strongly dependent on the types of wash solution, whereby acid-based agents yielded Pb deposition efficiency as high as 60%-70% whilst EDTA showed lower deposition efficiency at 46.41%. For NaNO₃, the deposition

efficiency only accounted for the precipitates that deposited on the cathode and the observed Pb deposition efficiency was 18.56%.



Figure 4.14: Physical appearance of cathode and deposits after the experiments: a) 0.01 M NaNO₃; b) 0.1 M HNO₃; c) 0.1 M citric acid; d) 0.01 M EDTA

Figure 4.14 shows that the physical appearance of deposits is dependent on the types of wash solution used. The figure reveals that when NaNO₃ is used, white/brownish precipitates are deposited on the cathode surface (Figure 4.14a-i) and some of the precipitates are suspended in the effluent (Figure 4.14a-ii). The white precipitate was identified as $Pb(OH)_2$ as a result of high pH in the cathode chamber. This was supported by a test which showed that Pb was detected in the sample of 0.1 M

HNO₃ solution after the white precipitate was dissolved in the solution. In contrast, soft and malleable grey solids were deposited on cathode surface when HNO₃, citric acid and EDTA were used. The solids in HNO₃ and EDTA tests were identified as mainly Pb as the colour of the deposit changed from grey to whitish yellow when they were dried in open air for one week, indicating Pb oxidation in the presence of moisture and oxygen. The illustrations are as in Figures 4.14b and 4.14d. On the other hand, Figure 4.14c shows a mixture of black and silver deposits is obtained on the cathode surface when citric acid is used. The colour of the deposits remained the same after drying at room temperature, suggesting a mixture of metals was deposited. This is verified through SEM analysis for soil surface, as shown in Figure 4.15. Unlike other tests which showed similar soil texture to virgin soil after the experiment, citric acid provided smoother soil surface, supporting the extraction of Pb and other metals from the soil. This is further justified through SEM-EDX analysis which shows that the deposits are primarily Pb, with the presence of other elements such as aluminium (Al) and titanium (Ti), as shown in Figure 4.16.



Figure 4.15: SEM images for the soil samples after the experiments of two-stage electrokinetic washing under different types of wash solutions



Figure 4.16: SEM-EDX analysis for the deposits obtained after two stage electrokinetic washing using 0.1 M citric acid

4.4.3.5 Power consumption

Application of two-stage electrokinetic washing process results in additional electric power consumption in the electrokinetic process (Stage 2) and this contributes to additional operating cost in comparison to normal soil washing. Power consumption can be expressed in terms of average power consumed by electrokinetic process per unit mass of Pb removed in electrokinetic washing system. This was calculated using Equation (4.6), where E_{PC} is the power consumed per kg Pb removed (kWh/kg), m_{Pb} is the mass of Pb removed (kg), V is the electric potential difference used (V), I is the current across the soil (A), and t is the time (h).

$$E_{PC} = \frac{1}{m_{Pb}} \int_{0}^{t} VI.dt$$
 (4.6)

Table 4.3 shows that acidic wash solutions consume more electric power in electrokinetic washing process but yield higher removal efficiency than the other two. Among the acidic wash solutions, citric acid only consumed 10% of the power than that of HNO₃ test for achieving high removal efficiency at 90.84%. In contrast, even though HNO₃ provides 99.99% Pb removal efficiency in two-stage electrokinetic washing, its relatively small removal enhancement of 4.98% compared to soil washing at high power consumption makes it less attractive as a wash solution in two-stage electrokinetic washing.

 Table 4.3: Power consumption for two-stage electrokinetic washing under different types of wash solutions

Wash solution	0.01 M	0.1 M	0.1 M	0.01 M
	NaNO ₃	HNO ₃	citric acid	EDTA
Removal efficiency by two-stage electrokinetic process, %	78.67	99.99	90.84	70.97
Power consumption, kWh/kg Pb removed	3.09	114.47	12.93	10.43

4.4.3.6 Selection of best wash solutions

 Table 4.4: Performance of two-stage electrokinetic washing on Pb removal using different types of wash solutions

Wash solution	0.01 M	0.1 M	0.1 M	0.01 M	
	NaNO ₃	HNO ₃	citric acid	EDTA	
Removal Efficiency, %	78.67	99.99	90.84	70.97	
Removal enhancement by electrokinetic process, %	20.40	4.98	20.45	7.84	
Effluent generation, mL	492.9	288.5	65.9	39.8	
Final current, mA	3.90	85.30	8.85	5.76	
Average soil pH	4.18	2.63	2.45	7.41	
Final pH in anode chamber	2.95	1.05	1.91	7.05	
Final pH in cathode chamber	9.85	1.41	1.99	11.93	
Electrode condition	Stable	Corrosion	Stable	Stable	
Power consumption, kWh/kg Pb removed	3.09	114.47	12.93	10.43	
Transport way	One way	One way	One way	Counter migration	
Table 4.4 summarizes the performance of each wash solution studied in twostage electrokinetic washing process based on the experimental results. Among the four wash solutions used in this work, citric acid was found to be the best wash solution in removing Pb from the soil in a two-stage electrokinetic washing. It could enhance the removal efficiency of Pb than normal soil washing by 20.45% at lower electroosmosis/effluent generation in comparison to NaNO₃. HNO₃ is not suitable as wash solution in two-stage electrokinetic washing as it showed insignificant removal enhancement than soil washing. Moreover, it also caused variation in cathode chamber pH, corroded graphite anode and increased power consumption. In contrast, citric acid did not exhibit such problems. Furthermore, citric acid also provides stable one way Pb migration into the cathode chamber in comparison to EDTA, which complicates the transport mechanisms via Pb counter-migration.

Two-stage electrokinetic washing using 0.1 M citric acid has been proven to successfully maintain high removal efficiency at 90.84% with low wash solution consumption at a solution: soil ratio of ≤ 1 mL: 1 g (≈ 247 mL: 250 g), where the wash solution used consisted of the solution in soil column, cathode chamber and the effluent generated via electroosmosis (wash solution in anode chamber is not considered as it has negligible Pb concentration). However, the effluent generated, at 65.9 mL, can be considered to be significantly high, and this was not desirable for the objective of this study. Hence, the effects of operating parameters such as electric potential difference and wash solution concentration on the system performance were further investigated using citric acid as the wash solution in order to suppress electroosmosis while maintaining the removal efficiency as these parameters had significant influence on electroosmosis and electromigration, according to Equations (2.3) and (2.4) (Baraud *et al.*, 1997; Lee *et al.*, 2009; Asadi *et al.*, 2013).

4.4.4 Effect of operating parameters

Equations (2.3) and (2.4) show that the velocities for electromigration and electroosmosis are highly dependent on the electricity that passes through the soil. Besides that, the concentration of wash solution in the system, which is citric acid in this study, also governs the effectiveness of both soil washing and electrokinetic processes in terms of Pb desorption. Thus, it is necessary to study the effect of these parameters on the performance of two-stage electrokinetic washing.

The effects of electric potential difference (A, range from 7.5-30 V), wash solution concentration (B, range from 0.001-0.1 M) and initial Pb concentration (C, range from 400-1000 mg/kg) on the system performance was investigated in order to minimize electroosmosis (effluent generation) while maintaining high removal efficiency at low power consumption. The study was investigated using Response Surface Methodology based on Box-Behnken Design. A total of 17 experiments were carried out with five replicates for the centre point experiment to analyze the error. The experimental design and results for i) removal efficiency, ii) effluent generated, and iii) power consumption are as shown in Tables 4.5 and 4.6, respectively.

Parameters	Code	Level		
		-1	0	1
Electric potential difference, V	А	7.5	18.75	30
Wash solution concentration, M	В	0.001	0.0505	0.1
Initial Pb concentration, mg/kg	С	400	700	1000

 Table 4.5: Parameters and levels of the experiments

Run No.	A, Electric potential difference,	B, Wash solution concentration,	C. Initial Pb concentration, mg/kg	Removal efficiency, %	Effluent generation, mL	Power consumption, kWh/kg Pb
	V	Μ				removed
1	18.75	0.0505	700	86.29	2.25	25.45
2	7.50	0.001	700	53.99	0	1.06
3	18.75	0.1	400	78.60	27.7	51.17
4	30.00	0.0505	1000	96.98	115.3	45.04
5	7.50	0.1	700	77.38	0	4.22
6	7.50	0.0505	1000	82.93	0	1.69
7	18.75	0.001	1000	84.52	0	6.23
8	18.75	0.0505	700	82.78	4.35	25.55
9	18.75	0.1	1000	90.79	27	21.49
10	18.75	0.0505	700	84.85	1.5	24.82
11	18.75	0.001	400	58.24	0	14.11
12	30.00	0.001	700	83.72	0	21.98
13	18.75	0.0505	700	85.19	3	25.37
14	30.00	0.1	700	94.62	103.35	79.53
15	18.75	0.0505	700	84.07	6.25	26.68
16	7.50	0.0505	400	61.12	0	5.82
17	30.00	0.0505	400	88.34	43.5	92.48

 Table 4.6: Experimental results using Response Surface Methodology based on Box-Behnken Design

4.4.4.1 ANOVA analysis

Analysis of Variance (ANOVA) was carried out to evaluate the significance and fitness of the operating parameters to the numerical model generated based on the experimental results obtained (Nosrati *et al.*, 2011). In general, the models were evaluated based on the criteria as shown below:

- (i) The model must be significant (Prob >F of less than 0.0001),
 - The terms that have Prob >F of less than 0.05 are significant terms for the model (95% confidence level),
 - iii) Insignificant lack of fit (Prob >F of more than 0.05),
 - iv) Difference between Predicted R^2 and Adjusted R^2 of less than 0.2 (Nosrati *et al.*, 2011),
 - v) Signal to noise ratio (Adeq Precision) of more than 4. (Long *et al.*, 2013)

	Sum of					
Source	Square	DF	Mean Square	F Value	Prob > F	
Model	2319.02	8	289.88	79.22	< 0.0001	significant
А	973.29	1	973.29	265.99	< 0.0001	
В	463.91	1	463.91	126.78	< 0.0001	
С	593.75	1	593.75	162.27	< 0.0001	
A^2	9.45	1	9.45	2.58	0.1467	
B^2	142.09	1	142.09	38.83	0.0003	
AB	39.00	1	39.00	10.66	0.0114	
AC	43.36	1	43.36	11.85	0.0088	
BC	49.63	1	49.63	13.56	0.0062	
Residual	29.27	8	3.66			
Lack of Fit	22.42	4	5.60	3.27	0.1388	not significant
Pure Error	6.85	4	1.71			
Cor Total	2348.30	16				
Std. Dev.	1.91		R-Squared	0.9875		
Mean	80.85		Adj R-Squared	0.9751		
C.V.	2.37		Pred R-Squared	0.9025		
PRESS	229.07		Adeq Precision	31.56		

Table 4.7: ANOVA for removal efficiency

Table 4.7 shows an example of ANOVA for removal efficiency. The results show that the model is significant as there is only less than 0.01% chance that the "Model F-value" this large could occur due to noise. The significant terms for the models are of A, B, C, B², AB, AC and BC. The lack of fit analysis indicates the model has insignificant lack of fit. The "Lack of fit F-value" of 3.27 implies that the lack of fit is not significant relative to the pure error as there is 13.88% lack of fit "F-value" in the model this large could occur due to noise; hence this model is acceptable for this study. The term "Adeq. Precision" of 31.56 and the difference between Predicted R² and Adjusted R² of 0.0726 further justify the suitability of the model for space navigation. ANOVAs for the effluent generated and power consumption were also analysed based on these criteria and the ANOVA tables were summarized in Appendix C. Based on the analysis as discussed above, the coded numerical models obtained are as shown in Equations (4.7) to (4.9). The values for A, B and C are in the range of -1 to 1, which is the level of the parameters studied, as shown in Table 4.5.

Removal Efficiency,
$$\% = 84.28 + 11.03A + 7.62B + 8.62C - 1.5A^2 - 5.8B^2$$

-3.12AB-3.29AC-3.52BC (4.7)

Effluent generated, mL =
$$3.47 + 39.70A + 13.68B - 0.18C + 24.2A^2 - 1.83B^2$$

+ $12.03C^2 + 25.84AB + 17.95AC + 12.16A^2B + 18.13A^2C$ (4.8)
- $13.86AB^2$

ln(power consumption/kg Pb removed) = $3.24 + 1.5A + 0.65B - 0.42C - 0.61A^{2}$ - $0.39B^{2} + 0.021C^{2} - 0.024AB + 0.13AC - 0.068A^{2}C$ (4.9)

The numerical models generated are further analysed via the diagnostic plots. Linear plots for normal probability and predicted versus actual results as shown in Figures 4.17 and 4.18, respectively, indicate that the models generated by ANOVA are suitable for results analysis, as the predicted results by the models are close with the experimental results.



Figure 4.17: Normal probability plot for: a) removal efficiency; b) effluent generated; c) power consumption



Figure 4.18: Predicted results vs actual results plot for: a) removal efficiency; b) effluent generated; c) power consumption

4.4.4.2 Removal efficiency

Figure 4.19 illustrates the response surface plots for removal efficiency under different parameters based on the mathematical model as shown in Equation (4.7). It was found that the increases in electric potential difference, wash solution concentration and initial Pb concentration created positive impact on the removal efficiency. Figures 4.19a and 4.19b show that the removal efficiency is proportional to electric potential difference, regardless of wash solution concentration and initial Pb concentration. Figure 4.19a reveals that an increase in electric potential difference from 7.5 V to 30 V enhances the removal efficiency steadily from 53.99% to 83.72% and 77.38% to 94.62% when the wash solution concentration used are 0.001 M and 0.1 M, respectively. According to Equations (2.3) and (2.4), the use of high electric potential difference could promote high removal efficiency, as a result of higher electric field strength and electric current through the soil for better electromigration as well as electroosmosis. This trend is in agreement with other works on the removal of heavy metals such as Cu (Zhou et al., 2004), Ni (Reddy and Karri, 2006a) and As (Yuan and Chiang, 2008) whereby higher removal efficiency and electromigration were reported when higher electric potential difference was used.

In addition, wash solution concentration also showed positive impact on the removal efficiency. As shown in Figures 4.19a, when the electric potential difference is constant, the increase in wash solution concentration from 0.001 M to 0.1 M steadily enhances the removal efficiency. When other parameters are held constant, the presence of higher citric acid concentration provided higher amount of H^+ ions for enhancing Pb desorption via low pH condition (Chaari *et al.*, 2008; Mohapatra *et al.*, 2009; Kim *et al.*, 2011b). Moreover, it also prevented the increase of pH in cathode chamber, which might reduce Pb electromigration via Pb hydroxide precipitation, especially near the cathode region (Kim *et al.*, 2011b). In addition, citric acid also can increase Pb

94

solubility via water soluble complex formation (Niinae *et al.*, 2008). Citric acid could promote dissolution of Fe and Al, which were the main binding sites for Pb on soil surface, as shown in Figure 4.16. All of the mechanisms mentioned above could contribute to better Pb desorption and removal efficiency. However, it is noted that the effect of wash solution concentration is more significant at low electric potential difference, as shown in Figure 4.19a. As the electric potential difference was increased from 7.5 V to 30 V, the enhancement caused by the increment in wash solution concentration from 0.001 M to 0.1 M was decreased from 23.39% to 10.9%. Moreover, the effect of wash solution is optimum at a concentration of 0.075 M for any electric potential difference used, as shown in Figure 4.19a. Further improvement on the removal efficiency can only be achieved by increasing the electric potential difference for better electromigration/electroosmosis rate while improving H⁺ generation via electrolysis to enhance Pb desorption. Thus, it is suggested that high removal efficiency could be attained by fine tuning the electric potential difference and wash solution concentration. This is important when both efficiency and cost are to be considered.

The study also suggested that two-stage electrokinetic washing is suitable for the soil with high contamination level as higher removal efficiency is achieved at higher initial Pb concentration when other parameters are held constant, as illustrated in Figures 4.19b and 4.19c. This was mainly due to the fact that higher amount of Pb was removed under high initial Pb concentration. In terms of Pb residual in the soil, initial Pb concentration did not show significant effect. This was supported by the residual Pb concentration in the soil after 24 h experiment. The residual Pb concentration in the soil after 24 h experiment. The residual Pb concentration in the soil after soil pb concentrations (400 mg/kg vs 1000 mg/kg) were used but other parameters were held constant. Examples are given in four cases, namely, i) Tests 11 vs 7 (171 mg/kg vs 163 mg/kg), ii) Tests 3 vs 9 (88 mg/kg vs 97 mg/kg), iii) Tests 17 vs 4 (48 mg/kg vs 32 mg/kg), and and iv) Test 16 vs 6 (160 mg/kg

vs 180 mg/kg). This proved that even though higher removal efficiency could be achieved, the amount of residual Pb was less dependent of initial Pb concentration.



Figure 4.19: Interaction effect of parameters on Pb removal efficiency: a) electric potential difference and wash solution concentration (initial Pb concentration: 700 mg/kg); b) electric potential difference and initial Pb concentration (wash solution concentration: 0.0505 M); c) wash solution concentration and initial Pb concentration (electric potential difference: 18.75 V)

4.4.4.3 Effluent generation

Sub-section 4.4.4.2 concluded that high Pb removal efficiency could be achieved under high electric potential difference with high wash solution concentration. However, it should be noted that the generation of high electric field/current under this condition can induce higher electroosmosis, resulting in higher effluent generation. This is not favoured as it requires higher cost for spent wash solution treatment. Thus, a study on the effect of the parameters mentioned in Table 4.5 on effluent generated was conducted.

The experimental setup had been configured by adjusting the overflow level in the cathode chamber to ≈ 0.6 cm higher than that of anode chamber's such that a consistent volume of effluent that was mainly attributed to strong electroosmosis only can be observed. Even though the adjustment was expected to reduce effluent generation, Table 4.6 shows that significant high amount of effluent is still generated in Tests 3, 4, 9, 14 and 17. These tests shared similar conditions, which were: i) high electric potential difference of >18.75 V, and ii) high wash solution concentration of >0.0505 M. According to Helmholtz-Smoluchowski theory as shown in Equation (2.4), electroosmosis is directly proportional to electric field strength across the soil and the negativity of the zeta potential of the soil (Lee et al., 2009; Asadi et al., 2013). In the present study, electric field strength had more impact than soil zeta potential, considering that the range for soil pH in this study was less than one (2.66-3.35). Thus, the effect of soil zeta potential difference was relatively small. This is supported by the experimental results illustrated in Figure 4.20, showing effluent generation via electroosmosis is strongly proportional to the electric current density in the system whereby a current density of ≥ 0.7 mA/cm² generates significant amount of effluent.



Figure 4.20: Relationship between effluent generated with electric current density

The response surface curves as shown in Figure 4.21 suggest that the increase in electric potential difference causes higher effluent generation. This is in line with the work of Zhou et al. (2004) which claimed that high electroosmosis was observed at high electric potential difference. However, the effect given by electric potential difference was also dependent on wash solution concentration and initial Pb concentration, as they governed the electric current in the system. Figure 4.21a shows that under low wash solution concentration of 0.001 M, insignificant amount of effluent was observed when electric potential difference was increased from 7.5 V to 30 V. This may be due to the relatively low electric current density (maximum at 0.34 mA/cm²), which indicated high electrical resistance in the system. In contrast, higher wash solution concentration and initial Pb concentration were responsible in reducing soil electrical resistance, which enhanced the electric current. Thus, higher electroosmosis was obtained, especially at an electric potential difference of >18.75 V. This was also in line with the works of Yang and Long (1999) and Kim et al. (2005a) which reported that higher electroosmosis was achieved when the current density across the system was enhanced. Nevertheless, it is worth noting that the effects of wash solution and initial Pb concentrations are low

unless high electrical driving force is applied, i.e. electric potential difference of \geq 18.75 V, as shown in Figures 4.21a and 4.21b.



Figure 4.21: Interaction effect of parameters on effluent generation: a) electric potential difference and wash solution concentration (initial Pb concentration = 700 mg/kg); b) electric potential difference and initial Pb concentration (wash solution concentration = 0.0505 M)



Figure 4.22: Interaction effect of parameters on power consumption: a) electric potential difference and wash solution concentration (initial Pb concentration: 700 mg/kg); b) electric potential difference and initial Pb concentration (wash solution concentration: 0.0505 M)

Power consumption was evaluated by the power consumed per kg Pb removed in a two-stage electrokinetic washing process, based on Equation (4.6). The ANOVA analysis suggests that the power consumption is an exponential function of the parameters studied, as shown in Equation (4.9). The response surface curves of ln (power consumption/kg Pb removed) versus the above-mentioned parameters are as illustrated in Figure 4.22. The figure reveals that the use of higher electric potential difference and wash solution concentration increased the power consumption due to the presence of excessive current, as a result of higher electrical driving force and amount of free ions available via electrolysis and wash solution concentration. Excessive current not only enhances Pb transport towards the cathode chamber but also transports non-targeted ions such as H^+ , Fe^{2+} , Fe^{3+} , Al^{3+} and other ions that are dissolved in the wash solution, which in turn consumes unnecessary electric power. Thus, the use of excessive electric potential difference and wash solution concentration to enhance removal efficiency is not an attractive option from the economic point of view. In contrast, the power consumption per kg Pb removed is found to be reduced when treating the soil that has higher initial Pb concentration, as shown in Figures 4.22b. A more effective Pb transport was achieved under this condition as the availability of Pb in the system was higher. Thus, it is suggested that two-stage electrokinetic washing is favoured for the soil that is highly contaminated.

4.4.4.5 Electric current density and pH in wash solution chambers and soil

Besides the properties mentioned above, other electrokinetic properties such as electric current density across the soil, the pH in wash solution chambers and soil were also investigated. The experiments showed that the current density in the system was stable during 24 h experiment. Therefore, the effect of operating parameters on the current density can be evaluated using "average current density". Figure 4.23a and 4.23b illustrate that the current density was proportional to the electric potential difference and wash solution concentration. The increase in electric potential difference from 7.5 V to 30 V and wash solution concentration from 0.001 M to 0.1 M increase the current density, when other parameters are held constant, as shown in Figure 4.23a and 4.23b. This followed normal electrokinetic process trend; according to Ohm's law,

electric potential difference is directly proportional to the electric current (Zhou *et al.*, 2004; Park *et al.*, 2009), which in turn increases ions' mobilities in the system (Reddy and Karri, 2006a). On the other hand, the use of higher wash solution concentration provided high amount of mobile ions in the system, which reduced the overall electrical resistance of the system. Thus, the current density was increased. In contrast, Figure 4.23c reveals that the effect of initial Pb concentration on current density is not as high as the other two. The effect of initial Pb concentration on the current density of the system was appreciable only when high electric potential difference and wash solution concentration were used.



Figure 4.23: Average current density of the system at different operating conditions: a) electric potential difference; b) wash solution concentration; c) initial Pb concentration



Figure 4.24: pH in anode and cathode chambers after 24 h experiments

Figure 4.24 shows that, in general, the pH in both anode and cathode chambers are low and stable throughout the experiments. As discussed in Sub-section 4.4.3.4, citric acid serves as buffer solution to prevent pH change in cathode chamber and this consequently eliminates base front from the cathode. However, it should be mentioned that high cathode chamber pH is observed in three cases, which are Tests 7, 11 and 12, as shown in Figure 4.24. These tests shared a similarity of low citric acid concentration (0.001 M) and high electric potential difference (>18.75 V). In these tests, the electrolysis, which generated high amount of OH⁻ ion outperformed H⁺ available in the cathode chamber. Thus, the pH increased. For other cases, which utilised citric acid concentration of >0.0505 M, the pH in the cathode chamber only increased slightly to \leq 3, indicating the ability of citric acid to maintain cathode chamber's pH.



Figure 4.25: Soil pH at different soil sections after 24 h experiments (Wash solution concentration: blue: 0.001 M; black: 0.0505 M; red: 0.1 M)

Other than affecting the pH of wash solution chambers, the use of higher citric acid concentration from 0.001 M to 0.1 M also results in lower soil pH, as shown in Figure 4.25. The soil pH was generally stable throughout the soil sections for all tests without drastic pH jump. This was mainly due to the increase in soil acidity when citric acid was flushed through the soil, considering the soil had low acid buffering capacity. In contrast, electric potential difference and initial Pb concentration showed relatively less impact on the soil pH. The acid front from electrolysis and citric acid in the soil column served as buffer for preventing pH change. Considering H⁺ electromigrated at a speed of \approx 1.8 times than OH⁻ in the soil (Acar and Alshawabkeh, 1993; Acar *et al.*, 1995; Chung and Kang, 1999; Gioannis *et al.*, 2008; Park *et al.*, 2009; Kim *et al.*, 2011b) and H⁺ buffer in the soil, the high pH generated in the cathode chamber in Tests 7, 11 and 12 was insufficient to create a strong base front to induce pH jump in the soil within the experimental duration.

4.4.4.6 Pb deposition efficiency



Figure 4.26: Physical appearance of deposit on cathode after the experiments



Figure 4.27: a) Deposition efficiency for Pb on the cathode at different current density;b) Mass of Pb electrodeposited at different initial Pb concentration (Wash solution: citric acid)

Figure 4.26 shows that Pb (grey solid) is recovered on the cathode during the experiments via electrodeposition. In general, it was found that the amount of Pb deposited was strongly dependent on the operating parameters studied. Figure 4.27a shows that the deposition efficiency is highly dependent on the current density across the system when citric acid is applied as the wash solution. This was in line with the Faraday 1st Law of Electrolysis that the mass of electrodeposited Pb during electrolysis is proportional to the quantity of electricity transferred at the electrode (Lou and Huang,

2007). Based on the discussion in Sub-section 4.4.4.5, the increase in electric potential difference and wash solution concentration which increased the current density of the system provided higher electrodeposition. Other than providing high electric current, the use of higher electric potential difference and wash solution concentration also improved Pb removal from soil and this could increase the availability of Pb in the cathode chamber for electrodeposition and thus, deposition efficiency. On the other hand, initial Pb concentration did not show a consistent trend in deposition efficiency, as the deposit on the cathode may affect the surface chemistry of the electrode and thus, the Pb electrodeposition rate. Nevertheless, the mass of electrodeposited Pb is found higher at higher initial Pb concentration, as shown in Figure 4.27b. This is mainly due to electrodeposition rate for a metal is higher when it has higher availability in the cathode chamber's solution (Senese, 2010).

4.4.4.7 Optimisation study

Sub-sections 4.4.4.2-4.4.4 highlight the fact that the increase in electric potential difference and wash solution concentration increase the removal efficiency. However, undesirable effluent generation and power consumption were also increased. Thus, an optimisation study was conducted on two-stage electrokinetic washing in order to determine the conditions to achieve optimum removal efficiency with negligible effluent generation at low power consumption. The optimisation study was conducted using Design Expert software 6.0.1 for treating 1000 mg/kg Pb contaminated soil. The criteria were set for: i) maximum removal efficiency; ii) minimum effluent generation, and iii) minimum power consumption.

The software suggests that a combination of 7.58 V and 0.057 M wash solution concentration can give an optimum removal efficiency of 84.58% at low power consumption (1.89 kWh/kg Pb removed) and \approx 0 mL of effluent generation, with a criteria desirability of 0.887. An experimental work was conducted based on these conditions for validation purpose. As shown in Table 4.8, the experimental results are in close agreement with the predicted results whereby the removal efficiency of 84.14% is obtained, which is within 1% error. The power consumption under these conditions was 2.27 kWh/kg Pb removed and no effluent was detected during the experiment.

Table 4.8: Predicted and experimental results for removal efficiency, effluent generated, and power consumption under the optimum conditions

Set Parameters	Predicted Value	Experimental
Electric potential difference, V	7.58	7.58
Wash solution concentration, M	0.057	0.057
Responses	Predicted Value	Experimental
Effluent generated, mL	0.017	0
Removal efficiency, %	84.58	84.14
Power consumption, kWh/kg Pb removed	1.89	2.27

4.4.5 Performance of two-stage electrokinetic washing under optimum conditions

The performance of two-stage electrokinetic washing under optimum conditions as suggested in Table 4.8 was further evaluated for its electrokinetic properties, such as current density, pH and electrodeposition. In addition, a comparison study on the performance of two-stage electrokinetic washing and normal soil washing on Pb removal efficiency was also carried out. Based on the results obtained, a possible transport mechanism has been proposed.



Figure 4.28: a) Change of current density with time during the experiment; b) Change of pH in wash solution chambers with time during the experiment; c) Change in soil pH during the experiment

The electrokinetic properties were evaluated from the aspect of variation in current density and pH in both wash solution chambers and soil sections. Figure 4.28a shows that a stable current density on average of 0.22 mA/cm^2 is achieved under the optimum condition during the experimental duration of 24 h. The low electric current provided minimum changes in the pH in wash solution chambers as a result of low electrolysis rate. This is observed throughout the experiment, whereby the pH values in both anode and cathode chambers remain in the range of 2.1-2.2, regardless of experimental duration, as shown Figure 4.28b. A stable pH in the cathode chamber was mainly attributed to the use of 0.057 M citric acid, which served as buffer solution in the cathode chamber to depolarize cathode reaction and maintain the pH (Gioannis et al., 2008; Murillo-Rivera et al., 2010; Li et al., 2012). Under low electrolysis rate, the amount of OH⁻ generated was relatively lower than citric acid buffer in the cathode chamber in the 24 h experiment. Thus, base front was eliminated in this operating condition. This eventually caused a stable low soil pH of 2.8 on average without any significant pH gradient, regardless of experimental duration and soil location, as illustrated in Figure 4.28c.

4.4.5.2 Removal efficiency

Figure 4.29a illustrates the effect of time on the removal efficiency for two-stage electrokinetic washing under optimum conditions. The figure shows that, in general, the application of electrokinetic process after soil washing increases the removal efficiency up to \approx 84% in 24 h in comparison to normal soil washing (0 h) at \approx 68% without extra wash solution consumption. The enhancement achieved by the electrokinetic process is mainly attributed to the electromigration of Pb towards the cathode chamber, whereby the increase in the experimental duration resultant improved removal efficiency, as

shown in Figure 4.29a. On the other hand, the effect of electroosmosis was negligible as no effluent was generated at these operating conditions.



Figure 4.29: Effect of time on two-stage electrokinetic washing at 7.58 V and 0.057 M citric acid concentration: a) removal efficiency; b) normalised Pb concentration at different soil sections (Initial Pb concentration, 1000 mg/kg)

In relation to Pb distribution in the soil, Figure 4.29b shows that low normalised Pb concentration is detected at anode region whilst higher normalised Pb concentration is found at cathode region after initial soil washing (0 h) as the washing process flushed the desorbed Pb towards the cathode chamber. When electrokinetic process is further applied, the normalised concentration decreases with time, following a similar Pb distribution trend as in soil washing, as shown in Figure 4.29b. Thus, it can be suggested that electrokinetic process provided one way Pb electromigration towards the cathode chamber and the effect of counter migration as observed in EDTA case (Subsection 4.4.3.1) was not found.

The above argument was strengthened via a general mathematical model. Based on the experimental results, three main assumptions were made:

- i) the removal efficiency enhancement was mainly due to net one way electromigration towards the cathode chamber,
- the effect of diffusion was neglected, as ionic mobility of a charged species is normally much higher than the diffusion coefficient under electric field (Acar and Alshawabkeh, 1993; Reddy and Cameselle, 2009), and
- iii) the change of electric potential difference across the soil with time is minimum at the experimental duration of 24 h.

Based on these assumptions, a general kinetic model for Pb electromigration in this study can be suggested as in Equation (4.10) (Acar and Alshawabkeh, 1993; Baraud *et al.*, 1997), where $C_{Pb,soil}$ is the normalised Pb concentration in soil, v_{EM} is electromigration velocity, A is cross sectional area of the column, $D_{soil/sys}$ is other soil and system constants such as porosity, tortuosity, and distribution coefficient. By grouping the constants A, v_{EM} , and D_{soil} as a general kinetic constant k, Equation (4.10) can be simplified as a first order kinetic model, as shown in Equation (4.11). A kinetic model that explains Pb concentration in the soil is obtained as Equation (4.12) after integrating the differential equation (4.11) from t=0 to t=t.

$$\frac{dC_{Pb,soil}}{dt} = -v_{EM} A D_{soil/sys} C_{Pb,soil}$$
(4.10)

$$\frac{dC_{Pb,soil}}{dt} = -kC_{Pb,soil} \tag{4.11}$$

$$C_{Pb\ soil} = \mathrm{e}^{\mathrm{ln}(C_{Pb\ soil}) - kt} \tag{4.12}$$



Figure 4.30: Average normalised Pb concentration in the soil at different treatment time during electrokinetic process stage

Figure 4.30 shows that the kinetic model expressed in Equation (4.12) explains the experimental results well, with an overall kinetic constant, k of 0.0281 h⁻¹ and normalised Pb concentration after soil washing, $C_{Pb,soil0}$ of 0.34 when the process is conducted for 24 h. Hence, it is suggested that a net transport for Pb towards the cathode chamber is achieved in electrokinetic process via first order electromigration. In addition, it is also suggested that further increase in treatment time to 70 h may achieve an equilibrium removal efficiency of 95%, if other system conditions such as cathode chamber pH and current are maintained throughout the experiments.

4.4.5.3 Deposition efficiency



Figure 4.31: Deposition efficiency of Pb in cathode chamber at different experimental duration at optimum conditions.

Figure 4.31 shows that Pb electrodeposition efficiency on cathode increases with time. However, unlike the trend for removal efficiency, the deposition rate slowed down as the experiment progressed and eventually achieved 49% deposition efficiency in 24 h test. This could be due to the fact that electrodeposition reduced Pb concentration in the wash solution in the cathode chamber with respect to time. This in turn reduced Pb availability for maintaining the electrodeposition rate as it was one of the main factors that affect electrodeposition rate (Senese, 2010). Moreover, the thickening of Pb deposits on the cathode during the experiment may also influence the electrodeposition process by affecting uniformity of current distribution on the cathode surface (Lou and Huang, 2007).

4.4.5.4 Possible transport mechanisms



Figure 4.32: General transport mechanisms in a two-stage electrokinetic washing process: a) initial washing; b) electrokinetic process

Based on the experimental results, transport mechanisms for Pb in a two-stage electrokinetic washing process using citric acid can be suggested. Figure 4.32 illustrates a general mechanism involved in two-stage electrokinetic washing. In general, the transport mechanisms can be divided into two sections: i) Pb desorption via initial washing process, and ii) Pb transport via electrokinetic process. Figure 4.32a illustrates Pb desorption when citric acid is in contact with the soil surface. Some of the desorbed Pb ions are transported to the cathode chamber via wash solution during the fill up process. Then, application of electricity further transports the suspended ions in the soil column towards their respective chamber, as shown in Figure 4.32b.



Figure 4.33: Schematic diagram for a) Pb desorption and b) Pb transport in two-stage electrokinetic washing when citric acid is applied as the wash solution

Figure 4.33 illustrates a) Pb desorption from a soil particle via soil washing, and b) Pb transport in electrokinetic process when citric acid is applied as the wash solution at the optimum condition in Sub-section 4.4.4.7. In general, Pb desorption process is pH dependent, in which low pH condition could give higher Pb desorption (Carlon *et al.*, 2004). Citric acid not only provides low pH condition for Pb desorption (Kim *et al.*, 2011b) but also acts as buffer solution for maintaining cathode chamber pH by nullifying OH⁻ via dissociated H⁺. Besides that, citric acid also causes detachment of surface ions via its viscosity and possible Pb complexation with citrate anions (Li *et al.*, 2012). As shown in Figure 4.33a, Pb is desorbed from soil surface due to low pH condition. Then, the desorbed Pb could undergo two mechanisms: i) remain as Pb²⁺ ions, and ii) react with citrate ions to form water soluble complexes. The possible complexes formed are suggested to be mainly PbH₂Citrate⁺, as the pH of the soil system is \approx 2.8 for the case of optimum condition, while PbH₂Citrate⁺ has higher stability constant (Niinae *et al.*, 2008). This was supported by Gu and Yeung (2011) in their work on Cd removal, whereby the major species in acidic condition (pH<3) were reportedly to be Cd²⁺ and CdH₂Citrate⁺. Under this pH, the citrate ions in the system will be mostly in undissociated and H₂Citrate⁻ forms, as the acid dissociation constant, pKa for citric acids are 3.13, 4.76, and 6.40 (Silva *et al.*, 2009), respectively, and the dissociations of citric acid at different pKa are as shown in Equations (4.13)-(4.15).

$$H_3Cit \Leftrightarrow H_2Cit^- + H^+ \qquad \text{pKa}_1: 3.13 \tag{4.13}$$

$$H_2Cit^- \Leftrightarrow HCit^{2-} + H^+ \qquad \text{pKa}_2: 4.76 \tag{4.14}$$

$$HCit^{2-} \Leftrightarrow Cit^{3-} + H^{+} \qquad \text{pKa}_{3}: 6.40 \qquad (4.15)$$

After the initial soil washing is completed, the use of electricity could provide electromigration as additional driving force to transport the desorbed Pb towards the cathode chamber, as illustrated in Figures 4.33b. One way Pb transport is achieved when citric acid is applied at low pH as Pb ions/complexes are in cationic forms and they are mainly electromigrated towards the cathode chamber. The electroosmosis is weak at the optimum parameters. However, it is worth noting that electroosmosis may also involve in Pb transport if it is significant, as discussed in Sub-section 4.4.3.2.

4.4.6 Summary and limitations

Table 4.9 shows that two-stage electrokinetic washing can enhance Pb removal efficiency from 68% to 84% without extra effluent generation at 0.057 M citric acid concentration and low magnitude electricity of 7.58 V (0.22 mA/cm^2). Enhanced removal of 16% could be achieved by this process in comparison to normal soil washing at similar low solution: soil ratio of <0.8 mL/g soil and operating condition. This proves that two-stage electrokinetic washing is effective in enhancing Pb removal efficiency at low amount of wash solution consumption in comparison to soil washing which often requires high solution: soil ratio for high removal efficiency, as discussed in Sub-section 2.2.1. It is worth noting that Pb has more affinity towards the soil in comparison to other divalent heavy metals (Turer and Genc, 2005; Kim *et al.*, 2005b; Violante *et al.*, 2007; Giannis *et al.*, 2009; Giannis *et al.*, 2012). Hence, based on the present study, it is suggested that two-stage electrokinetic washing is generally suitable for removal of divalent heavy metal cations that are mainly bound with the soil in the exchangeable, carbonate bounded and Fe-Mn oxides bounded forms.

and normal soil washing under optimum operating parameters								
and normal soil washing under optimum operating parameters Method Electric Citric acid Removal Solution: soil								
Table 4.9: Comparison between the performance of two-stage electrokinetic washing								

Method	Electric	Citric acid	Removal	Solution: soil
	potential	concentration, M	Efficiency, %	concentration,
	difference, V			mL/g
Soil washing	-	0.057	68	<0.8:1
Two-stage electrokinetic	7.58	0.057	84	<0.8:1
washing				

In spite of enhanced Pb removal efficiency, it is to be noted that there are several limitations for the present setup. As it involves initial soil washing, this configuration is only applicable to the soil that has higher hydraulic permeability, considering soil washing is normally poor for clayey soil (Kim *et al.*, 2012c). Besides that, electrokinetic process is also limited by the degree of solubilisation of the contaminants, as reported

by Acar *et al.* (1995) and Shenbagavali and Mahimairaja (2010). Two-stage electrokinetic washing is applicable if and only if the wash solution used can dissolve the contaminants and form uniformly charged (either in cationic only or anionic only) compounds for major contaminants for electromigration. It is worth noting that the performance of wash solution strongly affects the feasibility of two-stage electrokinetic washing from the aspects of removal efficiency, treatment time and electromigration direction. Thus, an evaluation of wash solution on the contaminant removal must be conducted before the application of this process.

4.5 Application of electrokinetic process as soil pretreatment method

4.5.1 Introduction

Electrokinetic process can transport dissolved heavy metals in saturated soil via electromigration even in the absence of pore flow (Acar *et al.*, 1995). This suggests that electrokinetic process can be utilised as soil pretreatment method to concentrate the heavy metals into smaller soil volume. The volume reduction of contaminated soil can in turn reduce the chemical costs when further remediation methods such as stabilisation and solidification (SS) is applied as the amount of chemical solution needed and waste produced are lower.

In this study, application of electrokinetic process as soil pretreatment method in migrating and concentrating heavy metals was investigated using a simple design. Unlike two-stage electrokinetic washing, there was no wash solution chamber for flushing in this setup. Instead, the electrode was in direct contact with the soil, as shown in Figure 3.3. The soil was saturated with the wash solution via sprinkling from the top of the soil, as proposed by Finzgar *et al.* (2006). Then, electrokinetic process was initiated by the application of low magnitude direct current through the electrodes for heavy metals migration. The details for the experimental setup and procedure were discussed in Sub-sections 3.2.2 & 3.3.5, respectively

The remediation study was investigated for two types of Pb contaminated soil, which were: i) single contaminated soil, and ii) co-contaminated soil. The effects of different types of wash solutions on electrokientic properties and Pb migration were evaluated. In addition, the feasibility of approaching electrode techniques in enhancing Pb migration using reduced power consumption was also investigated.

4.5.2 Approaching electrode assisted electrokinetic process

Approaching electrode is a technique in electrokinetic process. The term approaching electrode can be defined as the switching of either anode or cathode towards the other electrode from time to time with the purpose of reducing the distance between the electrodes and the migration path. In general, two types of approaching electrodes have been studied, namely: i) approaching anode (Shen *et al.*, 2007; Li *et al.*, 2012; Zhang *et al.*, 2014), and ii) approaching cathode (Shen *et al.*, 2009; Zhou *et al.*, 2014). The schematic diagrams for these configurations are as shown in Figures 4.34a and 4.34b.

The use of approaching electrode has been investigated to remove both cationic and anionic contaminants such as Cd (Shen et al., 2007), Hg (Shen et al., 2009), Cr (Li et al., 2012), Pb (Zhang et al., 2014) and fluorine (Zhou et al., 2014). It is noteworthy that the selection of approaching electrode is dependent on the types of contaminants to be removed, whereby approaching anode is mainly applied for cation contaminants whilst approaching cathode for anions. In general, the removal enhancement caused by approaching electrodes is mainly due to progressive soil conditioning, as approaching anode and cathode are reported to provide progressive soil acidification (Shen et al., 2007; Li et al., 2012; Zhang et al., 2014) and alkalisation (Shen et al., 2009; Zhou et al., 2014), respectively, for enhancing the desorption/migration conditions and preventing focusing effect of the contaminants in the soil. Besides that, electric current can also be enhanced by approaching electrode via reduction in migration distance (Shen et al., 2007; Shen et al., 2009; Li et al., 2012; Zhou et al., 2014). This could provide more intense electric field for electromigration in the soil and thus increase the removal efficiency. In addition, approaching electrode is also claimed to reduce the power consumption of the system by 16-44% (Shen et al., 2007; Shen et al., 2009; Zhou et al., 2014), as far less electrical energy is wasted in the cleaned soil (Shen et al., 2009). In
this study, the feasibility of approaching electrode in improving Pb electromigration in single and co-contaminated soils as well as the power consumption was investigated.



Figure 4.34: Schematic diagram for a) approaching anode, and b) approaching cathode (Shen *et al.*, 2007; Shen *et al.*, 2009)

4.5.3 Application of electrokinetic process in treating single contaminated soil

Technical feasibility of an electrokinetic process in treating Pb contaminated soil was investigated. The ability of the electrokinetic process to concentrate Pb into smaller soil volume via electromigration was evaluated using three wetting agents, namely i) 0.01 M NaNO₃, ii) 0.1 M citric acid, and iii) 0.1 M EDTA solution as they showed promising Pb desorption efficiency, as reported in Section 4.3. In contrast, HNO₃ was not utilised as the wetting agent in this study as it could corrode the graphite electrodes, as discussed in Sub-section 4.4.3.3.

4.5.3.1 Effect of wetting agents on electric current and soil pH

The variation in electric current and soil pH under different wetting agents are as illustrated in Figures 4.35a and 4.35b, respectively. Figure 4.35a shows that 0.1 M EDTA yields highest electric current, followed by 0.1 M citric acid and 0.01 M NaNO₃. This was perhaps due to the fact that EDTA produced more mobile ions in the system as it underwent multiple dissociations at solution pH of \approx 8-9 in comparison to citric acid and NaNO₃, which only dissociated partially at low pH and was in low concentration, respectively. Moreover, considering the relatively high availability of other metal ions such as Fe and Al in the soil, EDTA, as a non-selective metal chelating agent could also provide dissolution of metal-oxides and subsequently the formation of water soluble metal-EDTA complexes via ligand exchange reaction (Komárek *et al.*, 2007) besides the targeted contaminants, as observed in the work of Kim *et al.* (2011b). These increased the number of mobile ions in the soil system and thus, highest electric current was observed in 0.1 M EDTA test. The electric current was found to reduce steadily throughout the experiments for all types of wetting agents studied. This was mainly due to the loss of free ions and ionic strength in the system during the experiments and the

increase in resistance polarisation in the soil as they accumulated at their respective electrodes due to electromigration (Saichek and Reddy, 2003; Shen *et al.*, 2007; Shen *et al.*, 2009; Shrestha *et al.*, 2009; Giannis *et al.*, 2012; Li *et al.*, 2012; Zhang *et al.*, 2014). As wash solution/electrolyte chambers were absent for continuous supply of wetting agent, the amount of mobile ions decreased with time due to electromigration. Thus, the electric current across the soil decreased.



Figure 4.35: Effect of wetting agents on: a) electric current, and b) soil pH at different soil sections

The use of different types of wetting agents is found to show different soil pH development, as shown in Figure 4.35b. The soil is divided into four equivalent sections after the experiments, which are Soil Section 1 (S1), Soil Section 2 (S2), Soil Section 3 (S3) and Soil Section 4 (S4), as labelled in Figure 3.3. Among the soil sections, S1 and S2 are denoted as the anode region while S3 and S4 were denoted as the cathode region. In general, low pH was found in anode region S1 and high pH was observed in cathode region S4 as a result of electrolysis on the electrode, as shown in Equations (2.1) and (2.2). However, the pH gradient between both regions was dependent on the types of wetting agents. Figure 4.35b reveals that the use of 0.1 M EDTA shows a high pH gradient of 7.89. This could be attributed to high initial solution pH of 8-9, which increased the soil pH, especially at the cathode region. In addition, high pH gradient was also caused by high electric current generated in this test. This phenomenon enhanced the electrolysis at both anode and cathode (Tsai et al., 2010) for H⁺ and OH⁻ generation, respectively. Thus, a high pH gradient was observed. This argument was also applicable for 0.01 M NaNO₃ test, which had lower electric current in comparison to 0.1 M EDTA test. In this test, a lower pH gradient of 3.31 is observed, as shown in Figure 4.35b, as a result of lower electrolysis rate. However, in addition to the electric current, the nature of wetting agent may also have significant impact on the soil pH. As shown in Figure 4.35b, 0.1 M citric acid creates an acidic soil environment with a pH gradient of less than one even though it records higher electric current than 0.01 M NaNO₃ test. This was mainly attributed to the characteristic of citric acid which not only dissociated and provided additional H⁺ for soil acidification but also served as a buffer solution to prevent pH change in the cathode region (Kim *et al.*, 2011b). Thus, a drastic pH change, especially in S3-S4 region was not observed in the present study which utilised an experimental duration of 24 h.

4.5.3.2 Effect of wetting agents on Pb migration



Figure 4.36: Normalised Pb concentration at different soil sections after the experiments

The electromigration trend of Pb in the soil under different types of wetting agents is as shown in Figure 4.36. In general, it was found that the electrokinetic process had the potential to provide volume reduction in contaminated soil as the Pb concentration in \approx 50% of the soil sections was reduced to below the threshold concentration of 400 mg/kg (normalised concentration = 0.54) set up by Department of Environment Malaysia (DoE-Malaysia, 2009) via electromigration. In term of Pb migration, it was found that the use of 0.01 M NaNO₃ and 0.1 M citric acid provided Pb electromigration from anode region towards cathode region, and the latter generally provided better Pb migration than 0.01 M NaNO₃, as the normalised Pb concentrations in S1 and S2 were lower. The experimental evidence supported the fact that citric acid dissociated to provide additional H⁺ for soil acidification in all soil sections. Low pH environment provided favourable condition for Pb desorption (Yang *et al.*, 2006). Thus, better Pb mobility and electromigration was achieved. In contrast, NaNO₃ only served as an electrolyte for conducting electricity and initiating electrolysis in the system. Its inability to buffer OH⁻ generated during electrolysis caused relatively high pH at S3-S4

region than citric acid test, as shown in Figure 4.35b. High pH region may be the main cause for cessation of Pb electromigration and accumulation in S3 region. In this test, the pH at S3 is 4.61, which was reported to favour Pb adsorption (Chaari *et al.*, 2008; Mohapatra *et al.*, 2009). Moreover, a pH of higher than 6 at S4, which can cause Pb hydrolysis into Pb(OH)₂ precipitate or even Pb(OH)₃⁻ (Niu *et al.*, 2013) also served as another reason for Pb accumulation in S3 for 0.01 M NaNO₃ test.

On the other hand, 0.1 M EDTA showed a different migration trend. Figure 4.36 depicts that lower Pb concentration is recorded at the cathode region whilst higher Pb concentration is observed in anode region, with an accumulation of Pb at normalised concentration of 1.83 in S2 region. The different trend could be mainly due to the nature of the wetting agent used. In this test, the desorption process was mainly attributed to complex formation between Pb and EDTA, considering Pb mobility was poor at high pH condition (Niu et al., 2013). EDTA could be attached to Pb up to six sites, which helped in desorbing Pb effectively from the soil surface (Zhang et al., 2014). As the complexes formed were in negatively charged, the complexes electromigrated towards anode when electricity was applied (Amrate et al., 2005; Niinae et al., 2008; Alcántara et al., 2012; Zhang et al., 2014). However, Figure 4.36 also shows that Pb significantly accumulated in the S2 region. This trend was also observed in the work of Alcantara et al. (2012) whereby Pb accumulated in the middle soil section when EDTA was applied as the wash solution in electrokinetic process. This could be attributed to low soil pH of 2.5-3.6 in S1-S2 region, as shown in Figure 4.35b. Low pH condition was not conducive for EDTA as it was generally less soluble at low pH condition for complex formation (Allen and Hsien, 1993, as cited in Alcántara et al., 2012). In addition, the negatively charged Pb-EDTA complex may also be adsorbed on the soil surface (Reddy et al., 2010), especially at low pH condition. These mechanisms may contribute to the lower Pb electromigration, causing Pb accumulation at low pH S2 region.

4.5.3.3 Effect of approaching electrode on electric current and soil pH



Figure 4.37: Effect of approaching electrode technique on: a) electric current across the soil, and b) soil pH in electrokinetic process under different types of wetting agents (AA: approaching anode; AC: approaching cathode)

The ability of approaching electrode in enhancing system performance was investigated by switching one of the electrode pairs (anode or cathode) to the middle electrodes (Figure 3.3) at 12th hour of the experiments while maintaining a voltage gradient of 1 V/cm by adjusting the electric potential difference from 7.5 V to 3.75 V. Approaching anode was used for 0.01 M NaNO₃ and 0.1 M citric acid tests as Pb migration headed towards cathode. On the other hand, approaching cathode was applied for 0.1 M EDTA test, considering Pb electromigration towards anode was dominant.

Figure 4.37a illustrates that the use of approaching anode does not affect the electric current across the soil significantly when 0.01 M NaNO₃ and 0.1 M citric acid are applied as the wetting agents. This observation was different from other approaching anode studies which reported that the electric current was enhanced (Shen et al., 2007; Li et al., 2012). Unlike other studies, this could be due to the fact that the present system was operated in a close system without continuous supply of wetting agent/electrolyte from wash solution chambers. As majority of mobile ions were accumulated in S3-S4 region, the reduction in migration distance did not enhance ions mobility significantly and thus the current enhancement was less significant. The reduction in migration distance also did not enhance electric current for EDTA test. Instead, the electric current is reduced from 14.2 mA to 7.5 mA when the cathode is switched to the middle electrodes, as shown in Figure 4.37a. This trend was also observed in the work of Zhang et al. (2014) in which the electric current obtained by approaching anode technique was lower than fixed electrode tests. This could be attributed to several reasons. Firstly, the present study was conducted in a closed system which did not have any wash solution chamber for continuous supply of wetting agent. Considering there was significant amount of ions present in S3-S4 regions, such as OH⁻ and EDTA complexes, the switching of cathode into the middle soil section (S2) after 12th hour experiment may reduce the amount of mobile ions available under the electric

field, as the ions in S3-S4 were no longer involved in electrokinetic process. In addition, the reduction in electric current may also be attributed to the nature of EDTA and Pb-EDTA complexes, which had lower mobility at low pH S1-S2 region (Amrate *et al.*, 2005; Alcántara *et al.*, 2012), as discussed in Sub-section 4.5.3.2. This suggested that the amount of free mobile ions in this region was low. Thus, the electric current was reduced.

The use of approaching electrode technique in electrokinetic process had a strong bearing on the soil pH. Figure 4.37b shows that approaching anode provides progressive acidification for S3-S4 region in 0.01 M NaNO₃ test. As the anode was switched to the middle electrodes, H^+ was produced in the S2-S3 region. This shortened the migration distance for H^+ transport towards S4 region and improved the acid front, considering the electromigration speed of H^+ is about 1.8 times faster than OH⁻ (Acar and Alshawabkeh, 1993; Acar *et al.*, 1995; Chung and Kang, 1999; Gioannis *et al.*, 2008; Park *et al.*, 2009; Kim *et al.*, 2011b). Thus, high pH region was compressed. This observation was also in line with the study carried out by Li *et al.* (2012). On the other hand, Figure 4.37b reveals that progressive soil acidification is not significant in 0.1 M citric acid test as low soil pH was already achieved by the amount of H⁺ supplied by citric acid.

In contrast, the application of approaching cathode was found to provide progressive soil alkalisation. As shown in Figure 4.37b, soil pH in S2 and S3 are slightly higher when approaching cathode is adopted in 0.1 M EDTA test even though the electric current was reduced. This was also in line with the work of Shen *et al.* (2009) and Zhou *et al.* (2014) which reported that the use of approaching cathode increased soil pH via progressive soil alkalisation. The electrode switching into the middle electrode not only reduced the acid front towards S2-S3 but also induced electrolysis in S2 for

providing more OH⁻ to nullify the H⁺ available in S2 region and thus the soil pH was increased.



4.5.3.4 Effect of approaching electrode on Pb migration



Figure 4.38 illustrates the distribution of Pb at different soil sections between fixed electrode tests and approaching electrode tests. For approaching anode tests, the results showed that 0.01 M NaNO₃ provided a steady transport of Pb from anode to cathode, without accumulation of Pb in S3 region. The reduction in soil pH in cathode region provided weaker Pb adsorption on soil surface (Yang *et al.*, 2006), which improved Pb mobility for electromigration. Thus, higher normalised Pb concentration at 1.41 was observed in S4 in comparison to fixed anode test of 1.24. This was also in line with the work of Li *et al.* (2012) who claimed that Cr focusing effect was reduced via better soil acidification when approaching anode technique was applied. However, the shift of anode at the 12th hour during the experiment causes lower Pb migration in S1-S2 region as higher normalised Pb concentration is detected, as illustrated in Figure 4.38. This could be due to incomplete Pb electromigration during electrode switching when 0.01 M NaNO₃ was used as the wetting agent. The lack of electrical force to maintain Pb electromigration in S1-S2 regions after anode switching was suggested to be the main cause for higher Pb concentration in S1-S2 region.

In contrast, the use of approaching anode does not affect Pb electromigration trend significantly under citric acid test, as shown in Figure 4.38. A highly acidic condition established by citric acid provided high Pb desorption and mobility in comparison to NaNO₃. Thus, Pb electromigration rate was higher and majority of Pb migrated to S3-S4 region before anode switching. Nevertheless, as discussed above, normalised Pb concentration in S1-S2 is slightly increased by 0.06-0.09 (47-65 mg/kg), as shown in Figure 4.38, resultant from anode switching.

On the other hand, approaching cathode showed different Pb migration trend in 0.1 M EDTA test. Unlike the work of Shen *et al.* (2009) and Zhou *et al.* (2014), Figure 4.38 reveals that approaching cathode does not enhance Pb electromigration as Pb concentration remained high in S3-S4 region. This was mainly due to incomplete Pb electromigration in this region, as a result of the absence of electric field after cathode switching. Besides that, Figure 4.38 also shows that there is insignificant enhancement in Pb electromigration in S1-S2 region when approaching cathode is applied in 0.1 M EDTA test. Pb accumulation was still observed in S2 region. This could be attributed to the soil pH in this region favoured for Pb-EDTA adsorption as the pH was still low at 4.4 even though alkalisation occurred. In addition, a lowering in electric current after the switching (Figure 4.37a) also suggested that low amount of mobile ions were available in S1-S2 regions, as discussed in Sub-section 4.5.3.3. The low Pb-EDTA mobility at low pH condition was perhaps the main reason for insignificant enhancement in Pb migration in S1-S2 region during approaching cathode tests.

4.5.3.5 Removal efficiency and power consumption



Figure 4.39: Removal efficiency in the pretreated region and power consumption of the system under different types of wetting agents

Sub-sections 4.5.3.2 set forth electrokinetic process as an effective method in reducing contaminated soil volume via concentrating Pb into smaller soil volume. Figure 4.36 shows that, in general, a contaminated soil volume reduction of 50% could be achieved in this study for all types of wetting agents, as Pb concentration of less than the threshold concentration of 400 mg/kg (normalised concentration: 0.54) (DoE-Malaysia, 2009) is detected in either S1-S2 (NaNO₃ and citric acid) or S3-S4 region (EDTA). Pb removal efficiency in these regions is illustrated in Figure 4.39. The results showed that high removal efficiency in the range of 60% to 71% was achieved, in a sequence of 0.1 M citric acid \approx 0.1 M EDTA > 0.01 M NaNO₃. On the other hand, the power consumption, which was calculated based on Equation (4.16), was in the sequence of 0.1 M EDTA > 0.1 M citric acid > 0.01 M NaNO₃. This suggested that high power consumption did not always yield high Pb electromigration. Instead, a proper wetting agent which could desorb Pb effectively was more important in governing the removal efficiency.

Power Consumption, kWh =
$$\int_{0}^{t} VI \cdot dt$$
 (4.16)

On the other hand, Figure 4.39 also reveals that the use of approaching electrode did not enhance the removal efficiency in this study. As shown in Figure 4.39, the application of approaching electrode reduces the power consumption by 18% to 42%. This was attributed to the reduction in the applied voltage from 7.5 V to 3.75 V for maintaining the voltage gradient at 1V/cm, as the migration distance is reduced after electrode switching. However, this also reduces the removal efficiency by 7% to 30%. Approaching electrodes did not enhance overall Pb migration and contaminated soil volume reduction in 0.01 M NaNO₃ and 0.1 M EDTA tests. Instead, soil volume reduction decreased from 50% to 25%, as Pb concentration of >400 mg/kg (normalised concentration >0.54) was detected in S2 and S3 for 0.01 M NaNO₃ and 0.1 M EDTA tests, respectively. Nevertheless, Figures 4.38 and 4.39 also show that the use of approaching anode technique demonstrates advantageous in 0.1 M citric acid test through reducing the power consumption by 18.75% while maintaining 50% soil volume reduction, with a slight decrease in removal efficiency in S1-S2 region from 71% to 64%.

4.5.3.6 Summary of remediation of single contaminated soil

The study revealed that Pb ion migrated and concentrated to a smaller soil volume under electrokinetic process with the aid of wetting agent. In this study, citric acid showed better Pb migration than NaNO₃ as it provided better soil acidification for Pb desorption while preventing Pb adsorption/ precipitation in cathode region. In contrast, EDTA was not suitable as a wetting agent for Pb eletromigration in this study due to its poor mobility at low pH S1-S2 section. Introduction of approaching anode and

cathode enhanced soil acidification and alkalisation, respectively. However, using the present set up, they did not enhance Pb migration significantly, especially for 0.1 M EDTA tests. Nevertheless, this technique could reduce power wastage by 18.75% while maintaining efficient Pb migration and 50% soil volume reduction when citric acid was used as the wetting agent.

4.5.4 Application of electrokinetic process in treating co-contaminated soil

Sub-section 4.5.3 proves that electrokinetic process can be successfully applied as soil pretreatment method for concentrating Pb into smaller soil volume for the soil that was mainly contaminated by Pb. In order to further justify the feasibility of electrokinetic process in contaminated soil volume reduction, another study was conducted for treating a co-contaminated soil which consisted of 402.2 mg/kg Pb (cation contaminant) and 797.9 mg/kg Cr(VI) (anion contaminant) using similar equipment setup. Through this study, the effect of different types of contaminants with different charge on the effectiveness of electrokinetic process in electromigration and contaminated soil volume reduction could be evaluated.

4.5.4.1 Effect of wetting agents on electric current and soil pH

Figures 4.40a and 4.40b record the electric current passed through the soil during the experiments and final soil pH at different soil sections, respectively. In general, it is found that the magnitude of electric current is in a sequence of 0.1 M EDTA > 0.1 M citric acid > 0.01 M NaNO₃, as shown in Figure 4.40a. This trend is similar to single contaminated soil study, as discussed in Sub-section 4.5.3.1. However, it was worth noting that the initial electric current generated was higher in this study than single contaminated soil as the number of moles of mobile ions in the co-

contaminated soil (\approx 400 mg/kg Pb and \approx 800 mg/kg Cr) was higher. In addition, the electric current for 0.01 M NaNO₃ test is found to increase from 11.2 mA to 18.0 mA at the first 3 h of the experiment, as shown in Figure 4.40a. This was mainly due to the increase in concentration for mobile ions as a result of electrolysis and desorption/solubilisation of Pb and Cr. As the experiments progressed, the electric current decreased even at constant voltage gradient, regardless of wetting agent used. This was also observed in the works of other researchers; it could be attributed to the increase in resistance polarisation in the soil and loss of ionic strength in the system, as a result of the electromigration of mobile ions towards their respective electrodes (Saichek and Reddy, 2003; Shen *et al.*, 2007; Shen *et al.*, 2009; Shrestha *et al.*, 2009; Colacicco *et al.*, 2010; Giannis *et al.*, 2012; Zhang *et al.*, 2014).

Figure 4.40b illustrates soil pH at different soil sections after the experiments. It was found that, in general, low pH of 2.45-3.05 was observed in S1 whilst high pH of 9.98-10.04 was observed in S4 region. When NaNO₃ is utilised as the wetting agent, a stable pH increment from 3.05 to 10.02 is observed, as shown in Figure 4.40b. This is a normal trend for electrokinetic process as electrolysis occurs at both anode and cathode which produce H^+ and OH⁻, respectively (Acar and Alshawabkeh, 1993), as discussed in Sub-section 4.5.3.1. As the migration speed of H^+ is about 1.8 times faster than OH⁻ migration (Acar and Alshawabkeh, 1993; Acar *et al.*, 1995; Chung and Kang, 1999; Gioannis *et al.*, 2008; Park *et al.*, 2009; Kim *et al.*, 2011b), a pH gradient is observed in the S3 region. In contrast, when 0.1 M citric acid is used, the soil is at low pH in S1-S3 whilst S4 shows a much higher pH at 9.98, as shown in Figure 4.40b. Citric acid generally acidified the soil and served as buffer solution for preventing pH change. Thus, the only high pH region observed after the experiment was S4, which had undergone electrolysis via the cathode (Acar and Alshawabkeh, 1993). For 0.1 M EDTA test, high overall soil pH was obtained. Figure 4.40b depicts that a soil pH of 6.81 was achieved

in S2 region and a pH of >10 was obtained for S3-S4 regions whilst the only low pH region was in S1 at 2.45. High overall pH was mainly due to the pH of EDTA solution, which was around 8-9. The presence of OH^- in the wetting agent was responsible for increasing the soil pH before electrokinetic process was applied. This eventually caused higher soil pH in S2-S4 region in comparison to the tests that used 0.01 M NaNO₃ and 0.1 M citric acid.



Figure 4.40: a) Electric current across the soil, b) soil pH at different soil sections after the experiments

It was worth noting that higher pH gradient was observed in these tests in comparison to the single contaminated soil (Sub-section 4.5.3.1). This was mainly due to the difference in nature of soils manifested after spiking with different types of heavy metals. A higher electric current generated in these tests (Figure 4.40a) enhanced the electrolysis process (Tsai *et al.*, 2010). The higher rate in increment of H⁺ and OH⁻ concentrations at both regions for co-contaminated soil study eventually caused a sharper pH gradient in the soil. Besides that, a higher initial soil pH for co-contaminated soil (\approx 6) than single contaminated soil (\approx 4) after spiking could be responsible for higher pH in cathode region. This suggested that the acid buffering capacity for the co-contaminated soil was higher than that of single contaminated soil. The amount of H⁺ provided by the wetting agent in co-contaminated soil was inadequate to maintain low pH condition as in single contaminated soil.



4.5.4.2 Effect of wetting agents on Pb migration

Figure 4.41: Normalised Pb concentration at different soil sections after the experiments

Figure 4.41 illustrates the distribution of normalised Pb concentration at different soil sections after the experiments. The results revealed that different migration trends were observed for Pb when different types of wetting agents were used. The figure illustrates that, in general, significant Pb migration is only observed when 0.1 M EDTA is applied as the wetting agent whilst 0.01 M NaNO₃ and 0.1 M citric acid provides poor migration, even though these tests have significant electric current. For NaNO₃ and citric acid tests, minor migration of Pb was observed from both S1 and S4. The Pb is accumulated in S3 region with enrichment factor of 1.04-1.12 while other regions show normalized concentration of 0.9-1, as shown in Figure 4.41. The migration trend was contradictory to the work reported in Sub-section 4.5.3.2. This was mainly due to the difference in Pb species in Pb/Cr co-contaminated soil. In the presence of excessive Cr, the main Pb species in the co-contaminated soil was PbCrO₄, as a result of precipitation reaction between Pb(NO₃)₂ and K₂Cr₂O₇ during soil spiking (Madan and Prakash, 1987). PbCrO₄ generally had very low solubility in water (Eagleson, 1994) and they dissolve partially in acidic and basic media for electromigration, as shown in Equations (4.17) and (4.18), respectively (BUTE, n.d.). The low overall mobility for PbCrO₄ in these wetting agents could be the main reason for poor electromigration. This observation was also reported by Jensen et al. (2007) and Zhang et al. (2012) on the electromigration of PbCrO₄ using NaNO₃ and deionized water, respectively. Besides that, the enrichment of Cr in S1-S2 region during electromigration could also slowed down Pb migration towards the cathode via possible formation of water insoluble PbCrO₄ (Zhang *et al.*, 2012) during the contact. Moreover, the pH jump in S3-S4 region (Figure 4.40b) which favoured Pb adsorption and Pb(OH)₂ precipitation (Chaari et al., 2008; Mohapatra et al., 2009; Niu et al., 2013) may also contributed to Pb accumulation in S3 region, as discussed in Sub-section 4.5.3.2.

$$2PbCrO_4 + 2H^+ \Leftrightarrow 2Pb^{2+} + Cr_2O_7^{2-} + H_2O$$
(4.17)

$$PbCrO_{4} + 4OH^{-} \Leftrightarrow \left[Pb(OH)_{4}\right]^{2^{-}} + CrO_{4}^{2^{-}}$$

$$(4.18)$$

In contrast, Figure 4.41 shows that the use of 0.1 M EDTA provides Pb migration towards S1 (anode region), even though it had relatively lower concentration than Cr. This was mainly due to the effect of high pH EDTA solution which not only dissolved PbCrO₄ (Koshi and Iwasaki, 1983; Madan and Prakash, 1987; Tuli and R.L. Madan, 1999; Trishna Knowledge Systems, 2012) but also served as chelating agent for water soluble Pb-EDTA complexes formation (Niinae *et al.*, 2008; Zhang *et al.*, 2014). As the complexes formed are anionic, Pb electromigrates towards S1 under the electric field and accumulates with an enrichment factor of 1.55, as shown in Figure 4.41. Unlike the migration in single contaminated soil, Pb accumulation was not observed in S2 region. This was plausibly due to relatively high soil pH in S2 region (6.81 vs 3.63) which improved the stability of both EDTA and Pb-EDTA complex while minimizing the adsorption process, as discussed in Sub-section 4.5.3.2. In addition, Pb concentration in S3-S4 region also reduced to 0.34-0.37 of its initial concentration.

4.5.4.3 Effect of wetting agents on Cr migration

Figure 4.42 depicts the normalised Cr concentration at different soil sections using different types of wetting agents. It was found that the migration trend was generally different from Pb whereby Cr migration from S4 to S1 regions was achieved in all types of wetting agents. In this study, majority of Cr was more mobile than Pb as they were initially in hexavalent form after spiking and they were mainly adsorbed on the soil surface rather than in water insoluble form. The results, as shown in Figure 4.42, suggest that the migration follows the sequence of 0.1 M EDTA > 0.01 M NaNO₃ > 0.1 M citric acid. This trend is also observed from the physical appearance of the soil after the experiment, as shown in Figure 4.43 where the yellow and purple textures in the figure indicate high concentration of Cr ions and Cr(III)-EDTA (Hedrick, 1965), respectively, in the soil region. A stronger yellow texture in 0.01 M NaNO₃ test (Figure 4.43a) and purple texture in 0.1 M EDTA test (Figure 4.43c) in S1 region indicate Cr is electromigrated and concentrated in S1 region whilst a mild yellow texture in S1-S3 region, as shown in Figure 4.43b, shows poor Cr migration in 0.1 M citric acid test.



Figure 4.42: Normalised Cr concentration at different soil sections after the experiments

Figure 4.42 shows that unlike Pb, a smooth migration from S4 to S1 is observed for both 0.01 M NaNO₃ and 0.1 M EDTA even though the former does not enhance Cr solubility in aqueous phase. This could be mainly due to the higher mobility for Cr(VI) than PbCrO₄ in this study. Among the wetting agents, EDTA provided lower Cr concentration in S3-S4 region, which could be attributed to higher current for electromigration and relatively high soil pH provided by 0.1 M EDTA, which favoured Cr(VI) desorption (Hu *et al.*, 2005; Reddy, 2013; Troy, 2013). In addition, EDTA also showed higher Cr migration in S2 region as it could enhance Cr(VI) desorption via

relatively higher soil pH provided in S2 region. Moreover, it can complex with H⁺ ions or adsorbed on the soil surface, releasing Cr(VI) from the soil (Saeedi et al., 2013). EDTA also served as a chelating agent for the formation of Cr(III)-EDTA anionic complexes (Jung et al., 1997; Cao et al., 2011; Saeedi et al., 2013), which ensured one way electromigration towards S1. This migration trend was also in line with the work of Reddy and Chinthamreddy (2004) and Saeedi et al. (2013). In contrast, Cr migration slows down at low pH S1-S2 region when 0.01 M NaNO₃ is applied, as shown in Figure 4.42. Low soil pH condition provided more positively charged binding sites for the adsorption of negatively charged Cr(VI) compounds such as $Cr_2O_7^{2-}$ and CrO_4^{2-} (Hawley et al., 2005). This was also in line with the works of Hu et al. (2005) and Troy (2013) which reported that acidic condition favoured Cr(VI) adsorption on the soil surface's iron oxides. This in turn reduced the desorption process and Cr(VI) mobility in S1-S2 region and thus the electromigration was slowed down. In addition, low pH condition may also cause Cr(VI) reduction into Cr(III), especially in the presence of iron species as electron donor (Huang et al., 1995; Hawley et al., 2005; Weng and Tsai, 2009; Barrera-Díaz et al., 2012). These conditions were achieved in S1-S2 region in this test, as the soil had significant amount of iron content (3719 mg/kg). The formation of positively charged Cr(III) species and the inability of NaNO₃ to form anionic complexes with Cr(III) may lead to transport of Cr(III) towards the cathode region (S4) via electromigration. Consequently, the net electromigration rate for Cr towards anode in S1-S2 region decreased.



Figure 4.43: Physical observation of soil after the experiments under: a) 0.01 M NaNO₃;b) 0.1 M citric acid; c) 0.1 M EDTA (Left section: anode region (S1); Right section: cathode region (S4))

For the case of 0.1 M citric acid, Figures 4.42 and 4.43b show that significant Cr migration takes place only in S4 region whilst S1-S3 region shows weak migration. The migration in S4 was mainly attributed to high pH condition that favoured Cr(VI) desorption (Hu et al., 2005; Troy, 2013). In addition, citric acid was also reported to cause Cr(VI) reduction into Cr(III) (Meichtry et al., 2007), which could further form negatively charged complexes with citrate ion at a pH of >9 (Cao et al., 2011). Thus, Cr migrated away from S4. However, as S1-S3 was in low pH, the migration declined. Low soil pH condition not only favoured Cr(VI) adsorption (Hu et al., 2005; Troy, 2013) but also Cr(VI) reduction (Barrera-Díaz et al., 2012), especially in the presence of citric acid (Meichtry *et al.*, 2007). This is observed in the present study, as shown in Figure 4.42, where most of the Cr ion in the soil is in the form of Cr(III). Moreover, the formation of Cr(III)-citrate complex in the presence of citric acid could be another reason for poor Cr electromigration in S1-S3. As majority of Cr(III)-citrate complexes formed were in neutral or positive charged (Cao et al., 2011) at the resulted pH condition, it may have halted Cr electromigration towards S1. Furthermore, Cr(III)citrate is also reported to show high affinity towards adsorption on soil at low pH condition (Cao et al., 2011) and this could reduce the migration in S1-S3 region.

4.5.4.4 Effect of approaching electrode on electric current and soil pH

The results as shown in Figures 4.41 reveal that Pb is generally immobilized in the co-contaminated soil when 0.01 M NaNO₃ and 0.1 M citric acid are applied as wetting agents. On the other hand, 0.1 M EDTA provides one way electromigration for both Pb and Cr from S4 to S1 region, as shown in Figures 4.41 and 4.42. Thus, the feasibility of approaching electrode in enhancing Pb and Cr migration was evaluated for 0.1 M EDTA test only as it showed positive electromigration for both Pb and Cr in comparison to the other two. As the electromigration headed towards the anode, approaching cathode was applied.



Figure 4.44: Electrokinetic properties for fixed electrode test and approaching cathode test using 0.1 M EDTA as wetting agent: a) electric current across the soil; b) soil pH at different soil sections

Figure 4.44a shows that, unlike the works of Shen *et al.* (2009) and Zhou *et al.* (2014) which reported that approaching cathode enhanced electric current in the remediation process, the electric current generated is found to decrease from 17.1 mA to 8.7 mA when the cathodes are switched to the middle electrodes at 12th hour. The results follow similar trend as the single contaminated soil, as shown in Figure 4.37a.

This could be mainly due to the present experimental setup, which was a closed system in which the wash solution chamber was absent for continuous supply of wetting agent/ions into the system. The switching of electrode at 12^{th} hour of the experiment reduced the amount of mobile ions available under the electric field, as the ions in S3-S4 were no longer involved in electrokinetic process, as discussed in Sub-section 4.5.3.3. Thus, the electric current was reduced. Nevertheless, it is worth noting that despite the current reduction in approaching cathode test, Figure 4.44b depicts that soil pH in S2 increases from 6.81 to 8.18, indicating progressive soil alkalisation. Progressive alkalisation was also observed in the works of Shen *et al.* (2009), Zhou *et al.* (2014) and the treatment for single contaminated soil in the present work (Sub-section 4.5.3.3). This could be attributed to the OH⁻ production in S2 region for nullifying the available H⁺ in S2 and reduction in OH⁻ migration distance for better base front after the cathode was switched to the middle of the soil.

4.5.4.5 Effect of approaching electrode on Pb and Cr migrations

Figure 4.45a illustrates that, in general, the improvement in Pb migration from S2 to S1 via approaching cathode is insignificant under the present experimental conditions. Unlike the work of Shen *et al.* (2009), the increment in soil pH in the S2 region as well as the reduction in migration distance did not show significant improvement in Pb migration. In comparison with other publications which reported that approaching electrode enhanced the electromigration and removal efficiency (Shen *et al.*, 2007; Shen *et al.*, 2009; Li *et al.*, 2012; Zhang *et al.*, 2014; Zhou *et al.*, 2014), electromigration for Pb-EDTA was not enhanced in the present study as majority of Pb ions had already electromigrated close to the anode (S1 region). Thus, the mobility was not significantly enhanced when electrode distance was reduced. Moreover, possible

adsorption of negatively charged Pb-EDTA complexes on the soil (Reddy *et al.*, 2010) in low pH S1 region may be another reason for poor electromigration enhancement. Furthermore, Figure 4.45a also depicts that the switching process results in higher Pb concentration in S3-S4 region in comparison to fixed electrode test. This is in line with the observation for single contaminated soil, as shown in Figure 4.38, as the absence of electric field in S3-S4 region after cathode switching may cause incomplete Pb-EDTA migration in this region, as discussed in Sub-section 4.5.3.4. However, it was worth noting that the increase in normalised Pb concentration in S3-S4 region in approaching cathode test was relatively low for co-contaminated soil (0.36 to 0.43) in comparison to single contaminated soil (0.30 to 0.60). This can be mainly attributed to higher soil pH achieved in S2 region in co-contaminated soil for better Pb-EDTA complex electomigration in comparison to single contaminated soil, which tended to accumulate in low pH S2 region and reduced Pb electromigration.

Similar observation is also made for Cr, as shown in Figure 4.45b. The use of approaching cathode did not enhance Cr accumulation in S1 region significantly. Instead, it showed higher Cr concentration in S3-S4 region as a result of incomplete Cr migration due to cathode switching. Thus, it was suggested that the present soil pretreatment set-up and experimental condition is not appropriate for approaching cathode techniques to enhance electromigrations of Pb and Cr. Nevertheless, it is suggested that the effect of approaching cathode on heavy metals migration could be more significant when larger size of soil sandbox is used, whereby the distance between anode and cathode is longer so that multiple electrode switching can be used, as reported by Shen *et al.* (2009).



Figure 4.45: Normalised concentration for a) Pb, and b) Cr after the experiments at different soil sections under fixed electrode and approaching electrode tests using 0.1 M EDTA as the wetting agent

4.5.4.6 Removal efficiency and power consumption



Figure 4.46: Power consumption and removal efficiency in pretreated soil regions (S3-S4) under different types of wetting agents (AC: approaching cathode)

As shown in Figures 4.41 and 4.42, electrokinetic process concentrates Pb and Cr into S1 and S2 regions when EDTA is applied as the wetting agent. This generally provided soil pretreatment and contaminated soil volume reduction indirectly in S3-S4 region as the concentrations of Pb and Cr were reduced. Therefore, Pb and Cr removal was achieved in these regions. Figure 4.46 reveals that electrokinetic process yields high removal efficiency for both Pb and Cr at 64-70% in S3-S4 regions via electromigration when 0.1 M EDTA is used as the wetting agent in comparison to the other two. On the other hand, 0.01 M NaNO₃ and 0.1 M citric acid showed poor Pb removal efficiency in S3-S4 at -6% and 1%, respectively. This was particularly true for Pb due to poor mobility of PbCrO₄ for electromigration. Moreover, the removal efficiency for Cr in S3-S4 regions using these agents is also found to be relatively lower than 0.1 M EDTA, as shown in Figure 4.46.

The power consumption for electrokinetic soil pretreatment process was calculated based on Equation (4.16). Despite 0.1 M EDTA showing higher removal efficiency for Pb and Cr, Figure 4.46 shows that 0.1 M EDTA causes an electric power consumption of \approx 2.8 and \approx 2.4 times higher than 0.01 M NaNO₃ and 0.1 M citric acid tests, respectively. As an effort to improve Pb/Cr electromigrations at lower power consumption, the study on approaching cathode using 0.1 M EDTA revealed that enhancement of electromigrations in S1-S2 region was not achievable using the present setup. Nevertheless, a high saving in power consumption of 22.5% from 3.65x10⁻³ kWh to 2.83x10⁻³ kWh was obtained with a minor decrease in removal efficiency for both Pb and Cr by 4% to 8%.

4.5.4.7 Summary of remediation of co-contaminated soil

The study revealed that electrokinetic process can electromigrate and concentrate heavy metals into smaller soil volume even for the metal ions carrying different charges, as long as suitable wetting agent was used. In this study, EDTA emerged as the best wetting agent among the three. It not only provided high pH condition for enhancing Cr(VI) desorption but also performed as a chelant for negatively charged water soluble complexes formation for sparingly soluble Pb and Cr(III). Unlike the test for single contaminated soil, EDTA did not show Pb and Cr accumulation in the S2 region as a result of relatively high initial soil pH for co-contaminated soil. In contrast, NaNO₃ and citric acid showed poorer performance, particularly in providing Pb mobility for electromigration. From the aspect of approaching electrode technique, the use of approaching cathode in 0.1 M EDTA test provided better soil alkalisation in S2 region. However, this did not enhance the electromigration significantly for both Pb and Cr in S1-S2 region. Nevertheless, this

technique could save 22.5% of power consumption while maintaining efficient migration in S3-S4 when EDTA was used as wetting agent.

4.5.5 Summary

The study reveals that electrokinetic process is effective in electromigrating and concentrating Pb into smaller soil volume. This generally serves as a good choice as soil pretreatment method before the application of other remediation techniques such as stabilisation and solidification as the volume of contaminated soil can be reduced. The present study showed that this technique was in potential to reduce the contaminated soil volume for both single Pb contaminated soil and Pb/Cr co-contaminated soil. The performance of electrokinetic process was highly dependent on the types of wetting agents used, the nature of contaminates, and the soil condition. As discussed in Subsections 4.5.3.2, citric acid emerged as the best wetting agent in providing high Pb electromigration in single contaminated soil, in which the Pb was mainly adsorbed/ bounded on the iron/mineral oxides in the soil. However, citric acid showed poor Pb electromigration in co-contaminated soil which mainly consisted of water insoluble PbCrO₄, as discussed in Sub-section 4.5.4.2. Instead, EDTA emerged as a superior wetting agent in providing one way electromigration for both Pb and Cr towards the anode region in co-contaminated soil study.

The application of approaching electrodes in this study did not enhance electromigration for Pb and Cr significantly for both contaminated soils. Nevertheless, this technique reduced the power consumption for electrokinetic process while it only slightly reduced the removal efficiency in the pretreated soil sections when suitable wetting agent was used, as discussed in Sub-sections 4.5.3.5 and 4.5.4.6. Power wastage on the pretreated soil sections could be prevented by using approaching electrode.

152

CHAPTER 5: CONCLUSIONS AND RECOMMENDATIONS

This chapter summarises the findings on the technical feasibility study of the incorporation of electrokinetic process with other soil remediation methods based on the research objectives established. This chapter also proposes recommendations on future works.

5.1 Conclusions

The suitability of electrokinetic process to incorporate with other soil remediation methods was investigated from two angles: i) to enhance removal efficiency in soil washing at low chemical solution consumption, and ii) to serve as soil pretreatment method to concentrate heavy metals into smaller soil volume such that chemical usage for further remediation could be reduced.

5.1.1 Application of two-stage electrokinetic washing

Electrokinetic process had been incorporated with soil washing as a two-stage electrokinetic washing for remediating Pb contaminated soil in this study. Two-stage electrokinetic washing is a novel soil remediation technique in which the soil washing and electrokinetic process were operated at two different stages in a single equipment. This process consists of: i) initial soil washing to provide desorption condition to the soil, and ii) electrokinetic process to transport the desorbed Pb away from the soil. Soil washing was initiated by filling up anode chamber while the cathode chamber was left empty. Due to the hydraulic gradient difference between the chambers, the wash solution entered the soil column and provided Pb desorption. The transport of wash solution into cathode chamber provided early Pb removal until the cathode chamber was

fully filled up. Once the soil washing process was completed, electrokinetic process was induced as the second driving force for transporting the desorbed Pb in the soil towards the cathode chamber in the absence of hydraulic flow.

The study on the removal of Pb from high iron content sandy soil was initiated by screening the chemical agents via shake flask desorption tests. Based on the tests, 0.1 M HNO₃, 0.1 M citric acid and 0.01 M EDTA solution were used for two-stage electrokinetic washing as they not only showed high Pb removal efficiency but also had negligible undesirable effects such as odour and possible release of harmful gas during electrokinetic washing compared to H_2SO_4 , HCl and acetic acid.

The feasibility of two-stage electrokinetic washing in remediating 1000mg/kg Pb contaminated soil was investigated using 0.01 M NaNO₃, 0.1 M HNO₃, 0.1 M citric acid as well as 0.01 M EDTA as the wash solutions at a constant electric potential difference of 15 V for 24 h. The study revealed that two-stage electrokinetic washing could enhance Pb removal efficiency by 4.98-20.45% in comparison to normal soil washing via electromigration and electroosmosis. As for the wash solution, it was found that the removal efficiency followed a trend of 0.1 M HNO₃ > 0.1 M citric acid > 0.1 M NaNO₃ > 0.01 M EDTA. The removal efficiency was found higher when the wash solution provided low soil pH condition and high current density. This could be attributed to the fact that low soil pH condition enhanced Pb desorption while high electric current improved Pb electromigration rate after soil washing.

Among the wash solutions, it was suggested that NaNO₃ was unsuitable as it induced high degree of electroosmosis. High generation of effluent at 492.9mL (\approx 5.6 pore volume) was contradictory to the objective of two-stage electrokinetic washing. EDTA solution was also screened out as it only showed low removal efficiency enhancement of 7.84%, as a result of unfavorable counter-migration of Pb-EDTA in the two-stage electrokinetic washing. In contrast, the acidic based agents 0.1 M HNO₃ and 0.1 M citric acid achieved >90% removal efficiency. However, 0.1 M HNO₃ not only caused high amount of effluent generation (288.5 mL) with high electric power consumption but also fluctuation in cathode chamber's pH and corrosion of graphite anode, making the system unstable. Among the wash solutions, 0.1 M citric acid emerged as the best wash solution in this study as it not only yielded high removal efficiency enhancement by 20.45% in comparison to soil washing (from 70.39% to 90.84%) at relatively low effluent generation (65.9 mL) and power consumption in comparison to NaNO₃, HNO₃, and EDTA, but also maintained a stable system conditions for the electrodes and the pH values in wash solution chambers.

The effect of operating conditions such as electric potential difference, wash solution concentration and initial Pb concentration was further investigated using citric acid as the wash solution. The process was evaluated from the aspects of i) Pb removal efficiency, ii) effluent generation, and iii) power consumption such that the feasibility of two-stage electrokinetic washing in enhancing Pb removal efficiency while minimising effluent generation and power consumption could be justified. The results revealed that the increase in electric potential difference and wash solution concentration enhanced the removal efficiency and the interaction between these two parameters was significantly positive. Low pH and high current density were the most important criteria in the removal process. However, unfavourable high effluent generation and power consumption were also observed under these conditions. The optimisation study showed that optimum removal efficiency of 84.14% with negligible extra effluent generation and low power consumption of 2.27 kWh/kg Pb removed could be achieved under 7.58 V and 0.057 M citric acid concentration. The optimisation study proved that two-stage electrokinetic washing process was capable in enhancing the soil remediation efficiency by $\approx 16\%$ in comparison to normal soil washing (84% vs 68%) under similar low

solution: soil ratio of <0.8 mL: 1 g and operating conditions. In addition, this process also provided *in situ* Pb recovery as solid in cathode chamber by electrodepositing 49% of Pb from the system at the optimum condition. Based on the experimental results, it can be recommended that the incorporation of electrokinetic process into soil washing could enhance Pb removal via one way electromigration to cathode chamber at low wash solution consumption when a proper wash solution was used.

5.1.2 Application of electrokinetic soil pretreatment process

Application of electrokinetic process as soil pretreatment method to reduce contaminated soil volume via concentrating Pb into smaller soil section by electromigration was investigated for two types of contaminated soils, namely i) single Pb contaminated soil, and ii) Pb and Cr co-contaminated soil. The wetting agents investigated in this study were 0.01 M NaNO₃, 0.1 M citric acid and 0.1 M EDTA solution.

In general, it was observed that there was variation in the transport of Pb under different types of soils/contaminant species. For single contaminated soil, a potential contaminated soil volume reduction of \approx 50% was achieved for all wetting agents studied. Electrokinetic process provided good electromigration for Pb, which was mainly adsorbed/ bounded on the iron/mineral oxides. The migration trend was highly dependent on the types of wetting agent used, with a sequence of 0.1 M citric acid > 0.1 M EDTA > 0.01 M NaNO₃. For NaNO₃ and citric acid, Pb electromigrated towards cathode; the main mechanism for this process could be attributed to Pb²⁺ desorption at low soil pH condition, which was generated by either acid front (electrolysis) or the wetting agent (citric acid). On the other hand, the use of 0.1 M EDTA provided Pb electromigration towards anode region as a result of the formation of negatively charged water soluble Pb-EDTA complexes. It was worth noting that electromigration process was highly influenced by the soil pH. For NaNO₃, the increase in soil pH in S3-S4 caused cessation in Pb electromigration and accumulation of Pb in S3 was detected. On the other hand, Pb accumulated in low pH S2 region with a normalised concentration of 1.83 in EDTA test due to attenuation of Pb-EDTA electromigration, as a result of instability of EDTA and adsorption of Pb-EDTA on soil surface at low pH condition.

On the other hand, contaminated soil volume reduction for co-contaminated soil was only observed when 0.1 M EDTA solution was used as the wetting agent. NaNO₃ and citric acid showed poor Pb electromigration, as the Pb species in the cocontaminated soil was primarily water insoluble PbCrO₄. In this study, low pH condition generated by acid front and citric acid did not enhance Pb dissolution into ions significantly. In addition, low pH condition also slowed down Cr electromigration towards the anode region, especially in citric acid test. In contrast, EDTA solution emerged as a better wetting agent for this soil due to its high pH for Cr desorption and its chelating ability for water soluble Pb-EDTA and Cr(III)-EDTA complexes formation. One way electromigration towards anode was achieved for both Pb and Cr as the complexes formed were negatively charged, and this favoured soil volume reduction. The study proved that electrokinetic process is a technically feasible soil pretreatment method as it has the potential to reduce soil volume to about 50% of its initial amount in the present setup for both ions as well as both types of soils via electromigration when proper wetting agent was used, which was highly dependent on the properties of contaminants and soils.

The approaching electrode technique was further applied to improve electromigration via progressive soil conditioning. However, the results obtained were contradicted other publications as the use of approaching electrode did not enhance Pb electromigration significantly. Nevertheless, the study suggested that approaching electrodes reduced the power consumption for electrokinetic process. For single contaminated soil, approaching anode reduced power consumption by 18.75% while maintaining Pb electromigration when 0.1 M citric acid was applied as the wetting agent. On the other hand, the application of approaching cathode in co-contaminated soil using 0.1 M EDTA also showed a reduction in power consumption by 22.5%. This suggested that power wastage on the pretreated soil sections could be prevented using the approaching electrode technique. However, it was worth noting that the use of approaching electrodes caused a slight reduction in the removal efficiency for Pb and Cr in the pretreated regions (7-8% for single Pb contaminated soil and 4-8% for Pb/Cr co-contaminated soil), as a result of the absence of electrical driving force in these regions after electrode switching.

5.2 Recommendations for future studies

The study reveals that two-stage electrokinetic washing provides enhanced performance in Pb removal from iron rich sandy soil in Malaysia in comparison to normal soil washing at similar low wash solution: soil ratio in a laboratory scale set-up. However, it is to be noted that the current set-up restricts data collection, especially for continuous checking of heavy metal concentration in the soil. There is a need for further investigation on the feasibility of this process on heavy metal removal at a larger scale such that the kinetics and detailed transport mechanisms can be evaluated more effectively. The current small scale design and closed column is not suitable for soil sampling at different time interval as the sampling may cause drastic variation in the results.
For electrokinetic soil pretreatment process, a larger equipment is proposed to be utilised in future works instead of current small scale sandbox so that the effect of approaching electrode in electrokinetic process can be investigated more effectively. In addition to that, studies from the viewpoint of other hardware such as the materials, shapes and number of electrodes used, the arrangement as well as the distance between the electrodes in the soil are proposed as future works for evaluating/maximizing the effectiveness and uniformity of electric field/current. This is important for the improvement of the electromigration of heavy metals in the soil, especially when the process is applied at a larger scale such that the ability of electrokinetic process as soil pretreatment method in reducing contaminated soil volume can be enhanced at minimum power consumption.

Moreover, the feasibility studies on the applications of two-stage electrokinetic washing and electrokinetic soil pretreatment process in treating other types of soils, in particular, heavy metals contaminated sites, is urgently required. It should also be noted that the wash solutions/wetting agents used in the present study are conventional chemical agents, which are frequently used for soil remediation. However, these solutions are normally less environmental benign (e.g. HNO₃, EDTA) and often cause damage to the soil after the remediation process. Thus, there is an urgent need to study the feasibility of other types of chemicals that are least harmful such as ionic liquids, deep eutectic solvents and natural surfactants as the wash solution/ wetting agent in the two-stage electrokinetic washing as well as electrokinetic soil pretreatment process.

REFERENCES

- Abumaizar, R. J., & Smith, E. H. (1999). Heavy metal contaminants removal by soil washing. *Journal of Hazardous Materials*, 70(1–2), 71-86.
- Acar, Y. B., & Alshawabkeh, A. N. (1993). Principles of electrokinetic remediation. Environmental Science and Technology, 27(13), 2638-2647.
- Acar, Y. B., Gale, R. J., Alshawabkeh, A. N., Marks, R. E., Puppala, S., Bricka, M., & Parker, R. (1995). Electrokinetic remediation: Basics and technology status. *Journal of Hazardous Materials*, 40(2), 117-137.
- Adejumo, S. A., Togun, A. O., Adediran, J. A., & Ogundiran, M. B. (2010). Effects of compost application on remediation and the growth of maize planted on lead contaminated soil. Paper presented at the 19th World Congress of Soil Science Symposium 3.5.1: Heavy Metal Contaminated Soils, Brisbane, Australia.
- Ahmedzeki, N. S. (2013). Adsorption filtration technology using iron-coated sand for the removal of lead and cadmium ions from aquatic solutions. *Desalination and Water Treatment*, *51*(28-30), 1-7.
- Alcántara, M. T., Gómez, J., Pazos, M., & Sanromán, M. A. (2012). Electrokinetic remediation of lead and phenanthrene polluted soils. *Geoderma*, 173–174, 128-133.
- Almeira O, J., Peng, C.-S., & Abou-Shady, A. (2012). Simultaneous removal of cadmium from kaolin and catholyte during soil electrokinetic remediation. *Desalination*, 300, 1-11.
- Amrate, S., Akretche, D. E., Innocent, C., & Seta, P. (2005). Removal of Pb from a calcareous soil during EDTA-enhanced electrokinetic extraction. *Science of the Total Environment*, *349*(1–3), 56-66.
- Anonymous. (2013). *Lead*. Retrieved 3rd June 2013, from http://en.wikipedia.org/wiki/Lead
- Asadi, A., Huat, B. B. K., Nahazanan, H., & Keykhah, H. A. (2013). Theory of electroosmosis in soil. *International Journal of Electrochemical Science*, *8*, 1016-1025.

- Avram, A. M., Radoi, A., Schiopu, V., Avram, M., & Gavrila, H. (2011). Synthesis and Characterization of γ-Fe2O3 Nanoparticles for Applications in Magnetic Hyperthermia. Paper presented at the 2011 Soft Magnetic Materials Conference, Kos, Greece.
- Baek, K., Kim, D.-H., Park, S.-W., Ryu, B.-G., Bajargal, T., & Yang, J.-S. (2009). Electrolyte conditioning-enhanced electrokinetic remediation of arseniccontaminated mine tailing. *Journal of Hazardous Materials*, 161(1), 457-462.
- Baraud, F., Fourcade, M. C., Tellier, S., & Astruc, M. (1997). Modelling of decontamination rate in an electrokinetic soil processing. *International Journal* of Environmental Analytical Chemistry, 68(2), 105-121.
- Barrera-Díaz, C. E., Lugo-Lugo, V., & Bilyeu, B. (2012). A review of chemical, electrochemical and biological methods for aqueous Cr(VI) reduction. *Journal of Hazardous Materials*, 223–224(0), 1-12.
- Beyer, W. N. (1990). *Evaluating soil contamination*. U.S. Fish and Wildlife Service, Biological Report. Retrieved on 15th May 2015 from http://www.nwrc.usgs.gov/wdb/pub/others/FWS_Bio_Rep_90-2.pdf
- Boujelben, N., Bouzid, J., & Elouear, Z. (2009). Studies of lead retention from aqueous solutions using iron-oxide-coated sorbents. *Environmental Technology*, 30(7), 737-746.
- BUTE. (n.d.). The Group Ia Elements (Li, Na, K, Rb, Cs) and Their Principle Ions (Me+). Retrieved 22nd July 2015, from http://www.inc.bme.hu/en/subjects/inchem/sillabus/7-35.pdf
- Cao, X., Guo, J., Mao, J., & Lan, Y. (2011). Adsorption and mobility of Cr(III)-organic acid complexes in soils. *Journal of Hazardous Materials*, 192(3), 1533-1538.
- Carlon, C., Dalla Valle, M., & Marcomini, A. (2004). Regression models to predict water-soil heavy metals partition coefficients in risk assessment studies. *Environmental Pollution*, 127(1), 109-115.
- Chaari, I., Fakhfakh, E., Chakroun, S., Bouzid, J., Boujelben, N., Feki, M., Rocha, F., & Jamoussi, F. (2008). Lead removal from aqueous solutions by a Tunisian smectitic clay. *Journal of Hazardous Materials*, *156*(1–3), 545-551.

- Cheng, Z., Tan, A. L. K., Tao, Y., Shan, D., Ting, K. E., & Yin, X. J. (2012). Synthesis and characterization of iron oxide nanoparticles and applications in the removal of heavy metals from industrial wastewater. *International Journal of Photoenergy*, 2012, 1-5.
- Chung, H. I., & Kang, B. H. (1999). Lead removal from contaminated marine clay by electrokinetic soil decontamination. *Engineering Geology*, *53*(2), 139-150.
- Cline, S., & Reed, B. (1995). Lead removal from soils via bench-scale soil washing techniques. *Journal of Environmental Engineering*, 121(10), 700-705.
- Colacicco, A., De Gioannis, G., Muntoni, A., Pettinao, E., Polettini, A., & Pomi, R.
 (2010). Enhanced electrokinetic treatment of marine sediments contaminated by heavy metals and PAHs. *Chemosphere*, 81(1), 46-56.
- Cong, Y., Ye, Q., & Wu, Z. (2005). Electrokinetic behaviour of chlorinated phenols in soil and their electrochemical degradation. *Process Safety and Environmental Protection*, 83(2), 178-183.
- Dada, A. O., Olalekan, A. P., Olatunya, A. M., & Dada, O. (2012). Langmuir, Freundlich, Temkin and Dubinin–Radushkevich isotherms studies of equilibrium sorption of Zn²⁺ unto phosphoric acid modified rice husk. *IOSR Journal of Applied Chemistry*, 3(1), 38-45.
- Dermont, G., Bergeron, M., Mercier, G., & Richer-Laflèche, M. (2008). Soil washing for metal removal: A review of physical/chemical technologies and field applications. *Journal of Hazardous Materials*, 152(1), 1-31.
- Desogus, P., Manca, P. P., Orrù, G., & Zucca, A. (2013). Stabilization–solidification treatment of mine tailings using Portland cement, potassium dihydrogen phosphate and ferric chloride hexahydrate. *Minerals Engineering*, 45, 47-54.
- DoE-Malaysia. (2009). Contaminated land management and control guidelines No. 1: Malaysian recommended site screening levels for contaminated land. Retrieved 30 January 2012, from http://www.doe.gov.my/portal/wpcontent/uploads/Contaminated-Land-Management-and-Control-Guidelines-No-1_Malaysian-Recommended-Site-Screening-Levels-for-Contaminated-Land.pdf
- Doering, F., Doering, N., Iovenitti, J. L., & Hill, D. G. (2001). *Electrochemical remediation technologies for soil, sediment and ground water*. Paper presented at the Innovative strategies for the remediation of chlorinated solvents and dnapls in the subsurface.

Duruibe, J. O., Ogwuegbu, M. O. C., & Egwurugwu, J. N. (2007). Heavy metal pollution and human biotoxic effects. *International Journal of Physical Sciences*, 2(5), 112-118.

Eagleson, M. (1994). Concise encyclopedia chemistry. Berlin: Walter De Gruyter & Co.

- Erdem, M., & Özverdi, A. (2011). Environmental risk assessment and stabilization/solidification of zinc extraction residue: II. Stabilization/solidification. *Hydrometallurgy*, 105(3–4), 270-276.
- Falciglia, P. P., Cannata, S., Romano, S., & Vagliasindi, F. G. A. (2014). Stabilisation/solidification of radionuclide polluted soils — Part I: Assessment of setting time, mechanical resistance, γ-radiation shielding and leachate γradiation. *Journal of Geochemical Exploration*, 142, 104-111.
- Finzgar, N., Kos, B., & Lestan, D. (2006). Bioavailability and mobility of Pb after soil treatment with different remediation methods. *Plant, Soil and Environment*, 52(1), 25-34.
- FRTR. (n.d). 2.2.1 Data Requirements for Soil, Sediment, Bedrock and Sludge. Retrieved 16th April 2014, from http://www.frtr.gov/matrix2/section2/2 2 1.html
- Giannis, A., & Gidarakos, E. (2005). Washing enhanced electrokinetic remediation for removal cadmium from real contaminated soil. *Journal of Hazardous Materials*, *123*(1–3), 165-175.
- Giannis, A., Nikolaou, A., Pentari, D., & Gidarakos, E. (2009). Chelating agent-assisted electrokinetic removal of cadmium, lead and copper from contaminated soils. *Environmental Pollution*, 157(12), 3379-3386.
- Giannis, A., Tay, E., Kao, J., & Wang, J.-Y. (2012). Impact of vertical electrokineticflushing technology to remove heavy metals and polycyclic aromatic hydrocarbons from contaminated soil. *Electrochimica Acta, 86*, 72-79.
- Gioannis, G. D., Muntoni, A., Polettini, A., & Pomi, R. (2008). Enhanced electrokinetic treatment of different marine sediments contaminated by heavy metals. *Journal of Environmental Science and Health, Part A*, 43(8), 852.
- Global Info Mine. (n.d). *Global Info Mine Lead Mining*. Retrieved 3rd June 2013, from http://www.infomine.com/commodities/lead.asp

- Gómez, J., Alcántara, M. T., Pazos, M., & Sanromán, M. A. (2009). A two-stage process using electrokinetic remediation and electrochemical degradation for treating benzo[a]pyrene spiked kaolin. *Chemosphere*, 74(11), 1516-1521.
- Gu, Y.-Y., & Yeung, A. T. (2011). Desorption of cadmium from a natural Shanghai clay using citric acid industrial wastewater. *Journal of Hazardous Materials*, 191(1–3), 144-149.
- Gworek, B., & Mocek, A. (2003). Comparison of sequential extraction methods with reference to zinc fraction in contaminated soil. *Polish Journal of Environmental Studies*, 12(1), 41-48.
- Hamed, J. T., & Bhadra, A. (1997). Influence of current density and pH on electrokinetics. *Journal of Hazardous Materials*, 55(1–3), 279-294.
- Han, H., Lee, Y.-J., Kim, S.-H., & Yang, J.-W. (2009). Electrokinetic remediation of soil contaminated with diesel oil Using EDTA–cosolvent solutions. *Separation Science and Technology*, 44(10), 2437-2454.
- Han, J.-G., Hong, K.-K., Kim, Y.-W., & Lee, J.-Y. (2010). Enhanced electrokinetic (E/K) remediation on copper contaminated soil by CFW (carbonized foods waste). *Journal of Hazardous Materials*, 177(1–3), 530-538.
- Hansen, H. K., Rojo, A., & Ottosen, L. M. (2007). Electrokinetic remediation of copper mine tailings: Implementing bipolar electrodes. *Electrochimica Acta*, 52(10), 3355-3359.
- Harris, D. C. (2013). *Exploring chemical analysis* (5 ed.). New York: W. H. Freeman and Company.
- Hashimoto, Y., Tajima, K., Yamada, H., & Sato, T. (2010). *Rhizosphere bacterial* community PCR5DGGE profiles and metal speciation in shooting range soils. Paper presented at the 19th World Congress of Soil Science Symposium 3.5.1: Heavy Metal Contaminated Soils, Brisbane, Australia.
- Hawley, E. L., Deeb, R. A., Kavanaugh, M. C., & Jacobs, J. (2005). Treatment technologies for chromium(VI). In J. Guertin, C. P. Avakian & J. A. Jacobs (Eds.), *Chromium(VI) handbook* (pp. 273-308). Florida: CRC Press.
- Hedrick, C. E. (1965). Formation of the chromium-EDTA complex: An undergraduate kinetics experiment. *Journal of Chemical Education, 42*(9), 479.

- Hu, J., Chen, G., & Lo, I. M. C. (2005). Removal and recovery of Cr(VI) from wastewater by maghemite nanoparticles. *Water Research*, *39*(18), 4528-4536.
- Hu, J., Chen, G., & Lo, I. M. C. (2006). Selective Removal of heavy metals from industrial wastewater using maghemite nanoparticle: Performance and mchanisms. *Journal of Environmental Engineering*, 132, 709-715.
- Huang, C. P., Shin, H.-M., Allen, H. E., & Cheng, A. H.-D. (1995). Final Report on: Effect of specific chemical reactions on the transformation and the transport of chromium in the soil-water system. New Jersey: New Jersey Department of Environment Protection and Energym (Division of Science & Research).
- Hummers, W. S., & Offeman, R. E. (1958). Preparation of Graphitic Oxide. *Journal of the American Chemical Society*, 80(6), 1339-1339.
- IDEM. (2012). Appendix A: Screening levels. In IDEM (Ed.), *Remediation closure guide* (pp. 159-189).
- ILZSG. (2013). *Ilzfg Statistics*. Retrieved 3rd June 2013, from http://www.ilzsg.org/static/statistics.aspx
- Inchem. (1994). *Lead, inorganic (PIM 301)*. Retrieved 3rd June 2013, from http://www.inchem.org/documents/pims/chemical/inorglea.htm
- Isoyama, M., & Wada, S.-I. (2007). Remediation of Pb-contaminated soils by washing with hydrochloric acid and subsequent immobilization with calcite and allophanic soil. *Journal of Hazardous Materials*, *143*(3), 636-642.
- Jensen, P. E., Ottosen, L. M., & Harmon, T. C. (2007). The effect of soil type on the electrodialytic remediation of lead-contaminated soil. *Environmental Engineering Science*, 24(2), 234-244.
- Jung, G. Y., Kim, Y. S., & Lim, H. B. (1997). Simultaneous determination of chromium(III) and chromium(VI) in aqueous solution by capillary electrophoresis with on-column UV-VIS detection. *Analytical Sciences*, 13(3), 463-467.
- Kim, B.-K., Baek, K., Ko, S.-H., & Yang, J.-W. (2011a). Research and field experiences on electrokinetic remediation in South Korea. *Separation and Purification Technology*, 79(2), 116-123.

- Kim, D.-H., Jeon, C.-S., Baek, K., Ko, S.-H., & Yang, J.-S. (2009a). Electrokinetic remediation of fluorine-contaminated soil: Conditioning of anolyte. *Journal of Hazardous Materials*, 161(1), 565-569.
- Kim, D.-H., Ryu, B.-G., Park, S.-W., Seo, C.-I., & Baek, K. (2009b). Electrokinetic remediation of Zn and Ni-contaminated soil. *Journal of Hazardous Materials*, 165(1–3), 501-505.
- Kim, D.-H., Jo, S.-U., Choi, J.-H., Yang, J.-S., & Baek, K. (2012a). Hexagonal two dimensional electrokinetic systems for restoration of saline agricultural lands: A pilot study. *Chemical Engineering Journal*, 198–199(0), 110-121.
- Kim, G.-N., Jung, Y.-H., Lee, J.-J., Moon, J.-K., & Jung, C.-H. (2008a). Development of electrokinetic-flushing technology for the remediation of contaminated soil around nuclear facilities. *Journal of Industrial and Engineering Chemistry*, 14(6), 732-738.
- Kim, G.-N., Jung, Y.-H., Lee, J.-J., Moon, J.-K., & Jung, C.-H. (2008b). An analysis of a flushing effect on the electrokinetic-flushing removal of cobalt and cesium from a soil around decommissioning site. *Separation and Purification Technology*, 63(1), 116-121.
- Kim, G.-N., Yang, B.-I., Moon, J.-K., & Lee, K.-W. (2009c). Vertical electrokineticflushing remediation. Separation Science and Technology, 44(10), 2354-2370.
- Kim, G.-N., Kim, S.-S., Park, H.-M., Kim, W.-S., Park, U.-R., & Moon, J.-K. (2013). Cs-137 and Cs-134 removal from radioactive ash using washing–electrokinetic equipment. *Annals of Nuclear Energy*, 57, 311-317.
- Kim, G. N., Yang, B. I., Choi, W. K., Lee, K. W., & Hyeon, J. H. (2009d). Washingelectrokinetic decontamination for concrete contaminated with cobalt and cesium. *Nuclear Engineering and Technology*, *41*(8), 1079-1086.
- Kim, G. N., Kim, S. S., Kim, G. H., Park, H. M., Kim, W. S., Park, U. R., Kwon, H. J., Ryu, O. H., & Moon J. K. (2012b). *The removal of Cs-137 from soil using washing-electrokinetic decontamination equipment*. Paper presented at the Transaction of the Korean Nuclear Society Autumn Meeting, Gyeongju, South Korea.
- Kim, J. H., Han, S. J., Kim, S. S., & Yang, J. W. (2006). Effect of soil chemical properties on the remediation of phenanthrene-contaminated soil by electrokinetic-Fenton process. *Chemosphere*, 63(10), 1667-1676.

- Kim, K.-J., Kim, D.-H., Yoo, J.-C., & Baek, K. (2011b). Electrokinetic extraction of heavy metals from dredged marine sediment. *Separation and Purification Technology*, 79(2), 164-169.
- Kim, R.-Y., Sung, J.-K., Kim, S.-C., Jang, B.-C., & Ok, Y.-S. (2010). Effect of calcined eggshell on fractional distribution and plant uptake of Cd, Pb and Zn in contaminated soils near mine. Paper presented at the 19th World Congress of Soil Science Symposium 3.5.1: Heavy Metal Contaminated Soils, Brisbane, Australia.
- Kim, S.-S., Kim, J.-H., & Han, S.-J. (2005a). Application of the electrokinetic-Fenton process for the remediation of kaolinite contaminated with phenanthrene. *Journal of Hazardous Materials*, 118(1–3), 121-131.
- Kim, W.-S., Kim, S.-O., & Kim, K.-W. (2005b). Enhanced electrokinetic extraction of heavy metals from soils assisted by ion exchange membranes. *Journal of Hazardous Materials*, 118(1–3), 93-102.
- Kim, W.-S., Park, G.-Y., Kim, D.-H., Jung, H.-B., Ko, S.-H., & Baek, K. (2012c). In situ field scale electrokinetic remediation of multi-metals contaminated paddy soil: Influence of electrode configuration. *Electrochimica Acta*, 86, 89-95.
- Kogbara, R. B., Al-Tabbaa, A., Yi, Y., & Stegemann, J. A. (2013). Cement–fly ash stabilisation/solidification of contaminated soil: Performance properties and initiation of operating envelopes. *Applied Geochemistry*, 33, 64-75.
- Komárek, M., Tlustoš, P., Száková, J., Chrastný, V., & Balík, J. (2007). The role of Feand Mn-oxides during EDTA-enhanced phytoextraction of heavy metals. *Plant, Soil and Environment,* 53(5), 216-224.
- Korolev, V. A. (2006). *Electrokinetic remediation of a contaminated land in cities*. Retrieved 14th January 2012, from http://iaeg2006.geolsoc.org.uk/cd/PAPERS/IAEG_134.PDF
- Koshi, K., & Iwasaki, K. (1983). Solubility of low-solubility chromates and their clastogenic activity in cultured cells. *Industrial Health*, 21(2), 57-65.
- Lai, C. H., & Chen, C. Y. (2001). Removal of metal ions and humic acid from water by iron-coated filter media. *Chemosphere*, 44(5), 1177-1184.

- LDAI. (n.d). *LDAI Information*. Retrieved 3rd June 2013, from http://web.archive.org/web/20070827030846/http://www.ldaint.org/information. htm
- Lee, H.-H., & Yang, J.-W. (2000). A new method to control electrolytes pH by circulation system in electrokinetic soil remediation. *Journal of Hazardous Materials*, 77(1–3), 227-240.
- Lee, Y.-J., Han, H., Kim, S.-H., Yang, J.-W., & Aaron, J. J. (2009). Combination of electrokinetic separation and electrochemical oxidation for acid dye removal from soil. *Separation Science and Technology*, 44(10), 2455-2469.
- Lenntech. (n.d.). *Lead (Pb) Chemical properties, Health and Environmental effects*. Retrieved 19th May 2015, from http://www.lenntech.com/periodic/elements/pb.htm
- Li, G., Guo, S., Li, S., Zhang, L., & Wang, S. (2012). Comparison of approaching and fixed anodes for avoiding the 'focusing' effect during electrokinetic remediation of chromium-contaminated soil. *Chemical Engineering Journal, 203*, 231-238.
- Li, T., Yuan, S., Wan, J., Lin, L., Long, H., Wu, X., & Lu, X. (2009). Pilot-scale electrokinetic movement of HCB and Zn in real contaminated sediments enhanced with hydroxypropyl-β-cyclodextrin. *Chemosphere*, *76*(9), 1226-1232.
- Li, T., Yuan, S., Wan, J., & Lu, X. (2010). Hydroxypropyl-β-cyclodextrin enhanced electrokinetic remediation of sediment contaminated with HCB and heavy metals. *Journal of Hazardous Materials*, *176*(1–3), 306-312.
- Li, Z., Yuan, S., Wan, J., Long, H., & Tong, M. (2011). A combination of electrokinetics and Pd/Fe PRB for the remediation of pentachlorophenol-contaminated soil. *Journal of Contaminant Hydrology*, 124(1–4), 99-107.
- Liao, W.-P., Yang, R., Zhou, Z.-X., & Huang, J.-Y. (2014). Electrokinetic stabilization of heavy metals in MSWI fly ash after water washing. *Environmental Progress & Sustainable Energy*, 33(4), 1235-1241.
- Lima, A. T., Kleingeld, P. J., Heister, K., & Loch, J. P. G. (2011). Removal of PAHs from contaminated clayey soil by means of electro-osmosis. *Separation and Purification Technology*, *79*(2), 221-229.

- Liu, H., Probst, A., & Liao, B. (2005). Metal contamination of soils and crops affected by the Chenzhou lead/zinc mine spill (Hunan, China). *Science of the Total Environment*, 339(1–3), 153-166.
- Long, A., Zhang, H., & Lei, Y. (2013). Surfactant flushing remediation of toluene contaminated soil: Optimization with response surface methodology and surfactant recovery by selective oxidation with sulfate radicals. *Separation and Purification Technology*, 118, 612-619.
- Lou, H. H., & Huang, Y. (2007). Electroplating. In S. Lee (Ed), *Encyclopedia of chemical processing* (pp. 839-848). New York: Taylor and Francis.
- Ma, J. W., Wang, F. Y., Huang, Z. H., & Wang, H. (2010). Simultaneous removal of 2,4-dichlorophenol and Cd from soils by electrokinetic remediation combined with activated bamboo charcoal. *Journal of Hazardous Materials*, 176(1–3), 715-720.
- Madan, R. D., & Prakash, S. (1987). *Modern inorganic chemistry*. New Delhi: S. Chand & Company Ltd.
- Markus, J., & McBratney, A. B. (2001). A review of the contamination of soil with lead: II. Spatial distribution and risk assessment of soil lead. *Environment International*, 27(5), 399-411.
- Maturi, K., & Reddy, K. R. (2006). Simultaneous removal of organic compounds and heavy metals from soils by electrokinetic remediation with a modified cyclodextrin. *Chemosphere*, *63*(6), 1022-1031.
- McCartor, A., & Becker, D. (2010). *Blacksmith institute's world worst pollution* probelm report 2010. New York: Blacksmith Institute.
- Meichtry, J. M., Brusa, M., Mailhot, G., Grela, M. A., & Litter, M. I. (2007).
 Heterogeneous photocatalysis of Cr(VI) in the presence of citric acid over TiO₂ particles: Relevance of Cr(V)–citrate complexes. *Applied Catalysis B: Environmental*, 71(1–2), 101-107.
- Mellis, E. V., Casagrande, J. C., Soares, M. R., Cruz, M. C. P. d., & Camargo, O. A. d. (2012). Sorption of heavy metals in tropical soils. In H. M. Selim (Ed.), *Competitive sorption and transport of heavy metals in soils and geological media* (pp. 171-214). Boca Raton: CRC Press.

- MOE-Korea. (n.d). *Soil Pollutants and Soil Environment Standard*. Retrieved 3rd June 2013, from http://eng.me.go.kr/content.do?method=moveContent&menuCode=pol_wss_soi_standard
- Mohanty, B., & Mahindrakar, A. B. (2011). Removal of heavy metal by screening followed by soil washing from contaminated soil. *International Journal of Technology and Engineering System*, 2(3), 290-293.
- Mohapatra, M., Khatun, S., & Anand, S. (2009). Adsorption behaviour of Pb(II), Cd(II) and Zn(II) on NALCO plant sand. *Indian Journal of Chemical Technology*, *16*(4), 291-300.
- Montgomery, C. D. (2001). *Design and analysis of experiments* (5th ed.). New York: John Wiley & Sons, Inc.
- Moutsatsou, A., Gregou, M., Matsas, D., & Protonotarios, V. (2006). Washing as a remediation technology applicable in soils heavily polluted by mining-metallurgical activities. *Chemosphere*, *63*(10), 1632-1640.
- Murillo-Rivera, B., Gonzalez, I., Oropeza-Guzman, M. T., Escobar-Guerrero, V., Sumbarda-Ramos, E. G., & Teutli-Leon, M. M. (2010). Evaluation of lead removal from sandy soils using different electrolytes in electrokinetic experiments: prospective for remediation of a real site contaminated with mining wastes. *Journal of Applied Electrochemistry*, 40, 1145-1152.
- Ng, Y. S., Sen Gupta, B., & Hashim, M. A. (2016). Performance evaluation of natural iron-rich sandy soil as a low-cost adsorbent for removal of lead from water. *Desalination and Water Treatment*, *57*(11), 5013-5024.
- Niinae, M., Nishigaki, K., & Aoki, K. (2008). Removal of lead from contaminated soils with chelating agents. *Materials Transactions*, 49(10), 2377-2382.
- Niu, Y., Qu, R., Sun, C., Wang, C., Chen, H., Ji, C., Zhang, Y., Shao, X. & Bu, F. (2013). Adsorption of Pb(II) from aqueous solution by silica-gel supported hyperbranched polyamidoamine dendrimers. *Journal of Hazardous Materials*, 244-245, 276-286.
- Nor Wahidatul, A. Z. N., Syakirah Afiza, M., Saffaatul Husna, I., Amiza, W., & Amneera, W. A. (2012). Assessment of heavy metal in soil due to human activities in Kangar, Perlis, Malaysia. *International Journal of Civil and Environmental Engineering*, 12(6), 28-33.

- Nosrati, S., Jayakumar, N. S., & Hashim, M. A. (2011). Extraction performance of chromium (VI) with emulsion liquid membrane by Cyanex 923 as carrier using response surface methodology. *Desalination*, 266(1–3), 286-290.
- Park, J.-Y., Lee, H.-H., Kim, S.-J., Lee, Y.-J., & Yang, J.-W. (2007). Surfactantenhanced electrokinetic removal of phenanthrene from kaolinite. *Journal of Hazardous Materials*, 140(1–2), 230-236.
- Park, S.-W., Lee, J.-Y., Yang, J.-S., Kim, K.-J., & Baek, K. (2009). Electrokinetic remediation of contaminated soil with waste-lubricant oils and zinc. *Journal of Hazardous Materials*, 169(1–3), 1168-1172.
- Pazos, M., Sanromán, M. A., & Cameselle, C. (2006). Improvement in electrokinetic remediation of heavy metal spiked kaolin with the polarity exchange technique. *Chemosphere*, 62(5), 817-822.
- Pazos, M., Ricart, M. T., Sanromán, M. A., & Cameselle, C. (2007). Enhanced electrokinetic remediation of polluted kaolinite with an azo dye. *Electrochimica Acta*, 52(10), 3393-3398.
- Pazos, M., Plaza, A., Martín, M., & Lobo, M. C. (2012). The impact of electrokinetic treatment on a loamy-sand soil properties. *Chemical Engineering Journal*, 183, 231-237.
- Pensaert, S., De Groeve, S., Staveley, C., Menge, P., & De Puydt, S. (2008). Immobilisation, stabilisation, solidification: a new approach for the treatment of contaminated soils. Case studies: London Olympics & Total Ertvelde. Paper presented at the 15th Innovatieforum Geotechniek, Antwerpen, Belgium.
- Pham, T. D., Shrestha, R. A., Virkutyte, J., & Sillanpää, M. (2009). Combined ultrasonication and electrokinetic remediation for persistent organic removal from contaminated kaolin. *Electrochimica Acta*, *54*(5), 1403-1407.
- Qiu, R., Zou, Z., Zhao, Z., Zhang, W., Zhang, T., Dong, H., & We, X. (2010). Removal of trace and major metals by soil washing with Na₂EDTA and oxalate. *Journal of Soils and Sediments*, 10(1), 45-53.
- Rajić, L., Dalmacija, B., Perović, S. U., Pešić, V., Dalmacija, M., Klašnja, M., Tomin, M. B., & Watson, M. (2012). Improving the electrokinetic remediation of nickel-, cadmium-, and lead-contaminated sediment. *Soil and Sediment Contamination: An International Journal*, 22(2), 199-207.

- Reddy, K., & Chinthamreddy, S. (2004). Enhanced electrokinetic remediation of heavy metals in glacial till soils using different electrolyte solutions. *Journal of Environmental Engineering* 130(4), 442-455.
- Reddy, K. R., & Saichek, R. E. (2003). Effect of soil type on electrokinetic removal of phenanthrene using surfactants and cosolvents. *Journal of Environmental Engineering*, 129(4), 336-346.
- Reddy, K. R., & Saichek, R. E. (2004). Enhanced electrokinetic removal of phenanthrene from clay soil by periodic electric potential application. *Journal of Environmental Science and Health, Part A, A39*(5), 1189-1212.
- Reddy, K. R., & Ala, P. R. (2005). Electrokinetic remediation of metal-contaminated field soil. *Separation Science and Technology*, 40(8), 1701-1720.
- Reddy, K. R., & Karri, R. M. (2006a). Effect of voltage gradient on integrated electrochemical remediation of contaminant mixtures. *Land Contamination & Reclamation*, 14(3), 685-698.
- Reddy, K. R., Ala, P. R., Sharma, S., & Kumar, S. N. (2006b). Enhanced electrokinetic remediation of contaminated manufactured gas plant soil. *Engineering Geology*, 85(1–2), 132-146.
- Reddy, K. R., & Cameselle, C. (2009). *Electrochemical remediation technologies for polluted soils, sediments and groundwater*. New Jersey: Wiley.
- Reddy, K. R., Cameselle, C., & Ala, P. (2010). Integrated electrokinetic-soil flushing to remove mixed organic and metal contaminants. *Journal of Applied Electrochemistry*, 40(6), 1269-1279.
- Reddy, K. R. (2013). Electrokinetic remediation of soils at complex contaminated sites. In M. Manassero, A. Dominijanni, S. Foti & G. Musso (Eds.), *Coupled phenomena in environmental geotechnics* (pp. 131-147). Croydon: CRC Press.
- Reger, D., Goode, S., & Ball, D. (2010). *Chemistry: Principles and Practice*. Bedford: Cengage Learning.
- Ribeiro, A. B., Rodríguez-Maroto, J. M., Mateus, E. P., & Gomes, H. (2005). Removal of organic contaminants from soils by an electrokinetic process: the case of atrazine.: Experimental and modeling. *Chemosphere*, *59*(9), 1229-1239.

- Roy, A., & Bhattacharya, J. (2012). Removal of Cu(II), Zn(II) and Pb(II) from water using microwave-assisted synthesized maghemite nanotubes. *Chemcial Engineering Journal*, 211-212, 493-500.
- Rubim, J. C., Sousa, M. H., Silva, J. C. O., & Tourinho, F. A. (2001). Raman spectroscopy as a powerful technique in the characterization of ferrofluids. *Brazilian Journal of Physics*, 31(3), 402-408.
- Saeedi, M., Li, L. Y., & Gharehtapeh, A. M. (2013). Effect of alternative electrolytes on enhanced electrokinetic remediation of hexavalent chromium in clayey soil. *International Journal of Environmental Research*, 7, 39-50.
- Sahoo, S. K., Agarwal, K., Singh, A. K., Polke, B. G., & Raha, K. C. (2010).
 Characterization of γ- and α-Fe₂O₃ nano powders synthesized by emulsion precipitation-calcination route and rheological behaviour of α-Fe₂O₃.
 International Journal of Engineering, Science and Technology, 2(8), 118-126.
- Saichek, R. E., & Reddy, K. R. (2003). Effect of pH control at the anode for the electrokinetic removal of phenanthrene from kaolin soil. *Chemosphere*, 51(4), 273-287.
- Sanderson, P., Bolan, N., Bowman, M., & Naidu, R. (2010). Distribution and availability of metal contaminants in shooting range soils around Australia.
 Paper presented at the 19th World Congress of Soil Science Symposium 3.5.1: Heavy Metal Contaminated Soils, Brisbane, Australia.
- Senese, F. (2010). General Chemistry Online: FAQ: Redox reactions: What factors affect the amount of copper plated from a CuSO_4_solution?. Retrieved 20th May 2015, from http://antoine.frostburg.edu/chem/senese/101/redox/faq/cu-deposition.shtml
- She, P., Liu, Z., Ding, F. X., Yang, J. G., & Liu, X. (2003). Surfactant enhanced electroremediation of phenanthrene. *Chinese Journal of Chemical Engineering*, *11*(1), 73-78.
- Shen, Z., Chen, X., Jia, J., Qu, L., & Wang, W. (2007). Comparison of electrokinetic soil remediation methods using one fixed anode and approaching anodes. *Environmental Pollution*, 150(2), 193-199.
- Shen, Z., Zhang, J., Qu, L., Dong, Z., Zheng, S., & Wang, W. (2009). A modified EK method with an I–/I2 lixiviant assisted and approaching cathodes to remedy mercury contaminated field soils. *Environmental Geology*, 57(6), 1399-1407.

- Shenbagavalli, S., & Mahimairaja, S. (2010). Electro kinetic remediation of contaminated habitats. *African Journal of Environmental Science and Technology*, 4(13), 930-935.
- Shrestha, R. A., Pham, T.D., & Sillanpää, M. (2009). Remediation of chrysene from contaminated soil by enhanced electrokinetics. *International Journal of Electrochemical Science*, 4, 1387-1394.
- Silva, A. M. N., Kong, X., Parkin, M. C., Cammack, R., & Hider, R. C. (2009). Iron(III) citrate speciation in aqueous solution. *Dalton Transactions*, 40, 8616-8625.
- Sumbarda-Ramos, E. G., Guerrero-Gutierrez, O. X., Murillo-Rivera, B., Gonzaalez, I., & Oropeza-Guzman, M. T. (2010). Electrokinetic treatment for clayed and sandy soils. *Journal of Applied Electrochemistry*, 40, 1255-1261.
- Suzuki, T., Oyama, Y., Moribe, M., & Niinae, M. (2012). An electrokinetic/Fe0 permeable reactive barrier system for the treatment of nitrate-contaminated subsurface soils. *Water Research*, *46*(3), 772-778.
- Suzuki, T., Niinae, M., Koga, T., Akita, T., Ohta, M., & Choso, T. (2014). EDDSenhanced electrokinetic remediation of heavy metal-contaminated clay soils under neutral pH conditions. *Colloids and Surfaces A: Physicochemical and Engineering Aspects*, 440, 145-150.
- Tawinteung, N., Parkpian, P., DeLaune, R. D., & Jugsujinda, A. (2005). Evaluation of extraction procedure for removing lead from contaminated soil. *Journal of Environmental Science and Health Part A: Toxic/Hazardous Substances and Environmental Engineering*, 40(2), 385-407.
- Tejowulan, R. S., & Hendershot, W. H. (1998). Removal of trace metals from contaminated soils using EDTA incorporating resin trapping techniques. *Environmental Pollution*, 103(1), 135-142.
- Tessier, A., Cambell, P. G. C., & Bisson, M. (1979). Sequential extraction procedure for the speciation of particulate trace metals. *Analytical Chemistry*, *51*(7), 844-851.
- Torres, L. G., Lopez, R. B., & Beltran, M. (2012). Removal of As, Cd, Cu, Ni, Pb, and Zn from a highly contaminated industrial soil using surfactant enhanced soil washing. *Physics and Chemistry of the Earth*, *37-39*, 30-36.
- Trishna Knowledge Systems. (2012). Super course in chemistry for the IIT-JEE: Inorganic chemistry. New Delhi: Dorling Kindersley Pvt. Ltd.

- Troy, A. (2013). Co and Cr adsorption on maghemite, quartz, and maghemite-quartz mixtures. (Honor Theses). Western Michigan University, Michigan.
- Tsai, T.-T., Sah, J., & Kao, C.-M. (2010). Application of iron electrode corrosion enhanced electrokinetic-Fenton oxidation to remediate diesel contaminated soils: A laboratory feasibility study. *Journal of Hydrology*, 380(1–2), 4-13.
- Tuli, G. D., & R.L. Madan. (1999). S. Chand success guide inorganic chemistry. New Delhi: S. Chand & Company Ltd.
- Turer, D., & Genc, A. (2005). Assessing effect of electrode configuration on the efficiency of electrokinetic remediation by sequential extraction analysis. *Journal of Hazardous Materials*, 119(1–3), 167-174.
- USEPA. (1991). Superfund Enginering Issue: Treatment of lead-Contaminated Soils. Retrieved 3rd June 2013, from http://www.epa.gov/superfund/remedytech/tsp/download/leadcont.pdf
- USEPA. (1996). *A citizen's guide to soil washing*. Retrieved 19th May 2015, from http://www.epa.gov/tio/download/remed/soilwash.pdf

USEPA. (2012). Summary of Maximum Allowable Concentrations of Chemical Constituents In Uncontaminated Soil Used as Fill Material At Regulated Fill Operations. Retrieved 3rd June 2013, from http://www.epa.state.il.us/land/ccdd/new-max-allowable-concentrationstable.pdf

- Venkateswaran, P., Vellaichamy, S., & Palanivelu, K. (2007). Speciation of heavy metals in electroplating industry sludge and wastewater residue using inductively coupled plasma. *International Journal of Environmental Science and Technology*, 4(4), 497-504.
- Villen-Guzman, M., Garcia-Rubio, A., Paz-Garcia, J. M., Rodriguez-Maroto, J. M., Garcia-Herruzo, F., Vereda-Alonso, C., & Gomez-Lahoz, C. (2015). The use of ethylenediaminetetraacetic acid as enhancing agent for the remediation of a lead polluted soil. *Electrochimica Acta*, 181, 82-89.
- Violante, A., Krishnamurti, G. S. R., & Pigna, M. (2007). Factors affecting the sorption–desorption of trace elements in soil environments. In A. Violante, P. M. Huang & G. M. Gadd (Eds.), *Biophysico-chemical processes of heavy metals* and metalloids in soil environments (pp. 169-213). Chichester: John Wiley & Sons, Inc.

- Voglar, D., & Lestan, D. (2013). Pilot-scale washing of Pb, Zn and Cd contaminated soil using EDTA and process water recycling. *Chemosphere*, *91*(1), 76-82.
- Voglar, G. E., & Leštan, D. (2011). Efficiency modeling of solidification/stabilization of multi-metal contaminated industrial soil using cement and additives. *Journal* of Hazardous Materials, 192(2), 753-762.
- Wan, J., Yuan, S., Chen, J., Li, T., Lin, L., & Lu, X. (2009). Solubility-enhanced electrokinetic movement of hexachlorobenzene in sediments: A comparison of cosolvent and cyclodextrin. *Journal of Hazardous Materials*, 166(1), 221-226.
- Wan, J., Li, Z., Lu, X., & Yuan, S. (2010). Remediation of a hexachlorobenzenecontaminated soil by surfactant-enhanced electrokinetics coupled with microscale Pd/Fe PRB. *Journal of Hazardous Materials*, 184(1–3), 184-190.
- Wan Zuhairi, W. Y. (2011). Lecture 5: Standard and Guidelines in Contaminated Land. Retrieved 3rd June 2013, from http://pkukmweb.ukm.my/zuhairi/Pengajaran/intranet/STAG3112/WePapers.co m/STAG3342_L5_environmental%20standards.pdf
- Wang, J.-Y., Huang, X.-J., Kao, J. C. M., & Stabnikova, O. (2006). Removal of heavy metals from kaolin using an upward electrokinetic soil remedial (UESR) technology. *Journal of Hazardous Materials*, 136(3), 532-541.
- Wang, J.-Y., Huang, X.-J., Kao, J. C. M., & Stabnikova, O. (2007). Simultaneous removal of organic contaminants and heavy metals from kaolin using an upward electrokinetic soil remediation process. *Journal of Hazardous Materials*, 144(1– 2), 292-299.
- Weng, C.-H., & Tsai, H.-W. (2009). A pilot-scale test of electrokinetic remediation of Cr(VI) contaminated kaolinite incorporated with zero-valen iron. *Journal of Environmental Engineering & Management*, 19(6), 379-387.
- Wuana, R. A., & Okieimen, F. E. (2011). Heavy metals in contaminated soils: A review of sources, chemistry, risks and best available strategies for remediation. *International Scholarly Research Network Ecology, 2011*, 1-20.
- Yang, G. C. C., & Lin, S.-L. (1998). Removal of lead from a silt loam soil by electrokinetic remediation. *Journal of Hazardous Materials*, 58(1–3), 285-299.

- Yang, G. C. C., & Long, Y.-W. (1999). Removal and degradation of phenol in a saturated flow by in-situ electrokinetic remediation and Fenton-like process. *Journal of Hazardous Materials*, 69(3), 259-271.
- Yang, G. C. C., & Liu, C.-Y. (2001). Remediation of TCE contaminated soils by in situ EK-Fenton process. *Journal of Hazardous Materials*, 85(3), 317-331.
- Yang, J. Y., Yang, X. E., He, Z. L., Li, T. Q., Shentu, J. L., & Stoffella, P. J. (2006). Effects of pH, organic acids, and inorganic ions on lead desorption from soils. *Environmental Pollution*, 143(1), 9-15.
- Yang, Y. C., Shaaban, M. G., & Mahmud, H. (2004). Remediation of heavy metal contaminated soil by using chemical stabilization. *Journal Institution of Engineer Malaysia 65*(12), 29-33.
- Yang, Z., Zhang, S., Liao, Y., Li, Q., Wu, B., & Wu, R. (2012). Remediation of Heavy Metal Contamination in Calcareous Soil by Washing with Reagents: A Column Washing. *Procedia Environmental Sciences*, 16, 778-785.
- Yeung, A. T. (2011). Milestone developments, myths, and future directions of electrokinetic remediation. Separation and Purification Technology, 79(2), 124-132.
- Yeung, A. T., & Gu, Y.-Y. (2011). A review on techniques to enhance electrochemical remediation of contaminated soils. *Journal of Hazardous Materials*, 195, 11-29.
- Yuan, C., & Chiang, T.-S. (2008). Enhancement of electrokinetic remediation of arsenic spiked soil by chemical reagents. *Journal of Hazardous Materials*, 152(1), 309-315.
- Yuan, C., Hung, C.-H., & Chen, K.-C. (2009). Electrokinetic remediation of arsenate spiked soil assisted by CNT-Co barrier—The effect of barrier position and processing fluid. *Journal of Hazardous Materials*, *171*(1–3), 563-570.
- Yuan, S., Tian, M., & Lu, X. (2006). Electrokinetic movement of hexachlorobenzene in clayed soils enhanced by Tween 80 and β-cyclodextrin. *Journal of Hazardous Materials*, 137(2), 1218-1225.
- Yuan, S., Tian, M., Li, T., & Lu, X. (2007). Effect of the composition of anodic purging solutions on electroosmosis. *Journal of Environmental Engineering*, 133(6), 627-632.

- Yuan, S., Long, H., Xie, W., Liao, P., & Tong, M. (2012). Electrokinetic transport of CMC-stabilized Pd/Fe nanoparticles for the remediation of PCP-contaminated soil. *Geoderma*, 185–186, 18-25.
- Zhang, P., Jin, C., Zhao, Z., & Tian, G. (2010). 2D crossed electric field for electrokinetic remediation of chromium contaminated soil. *Journal of Hazardous Materials*, 177(1–3), 1126-1133.
- Zhang, T., Zou, H., Ji, M., Li, X., Li, L., & Tang, T. (2014). Enhanced electrokinetic remediation of lead-contaminated soil by complexing agents and approaching anodes. *Environmental Science and Pollution Research*, *21*(4), 3126-3133.
- Zhang, W., Zhuang, L., Tong, L., Lo, I. M. C., & Qiu, R. (2012). Electro-migration of heavy metals in an aged electroplating contaminated soil affected by the coexisting hexavalent chromium. *Chemosphere*, 86(8), 809-816.
- Zhou, D.-M., Deng, C.-F., & Cang, L. (2004). Electrokinetic remediation of a Cu contaminated red soil by conditioning catholyte pH with different enhancing chemical reagents. *Chemosphere*, 56(3), 265-273.
- Zhou, D.-M., Deng, C.-F., Cang, L., & Alshawabkeh, A. N. (2005). Electrokinetic remediation of a Cu–Zn contaminated red soil by controlling the voltage and conditioning catholyte pH. *Chemosphere*, 61(4), 519-527.
- Zhou, D.-M., Cang, L., Alshawabkeh, A. N., Wang, Y.-J., & Hao, X.-Z. (2006). Pilotscale electrokinetic treatment of a Cu contaminated red soil. *Chemosphere*, 63(6), 964-971.
- Zhou, M., Zhu, S., Liu, Y., & Wang, H. (2014). Electrokinetic Remediation of Fluorine-Contaminated Soil Using Approaching Cathodes. *CLEAN – Soil, Air, Water, 42*(12), 1771-1775.
- Zou, Z., Qiu, R., Zhang, W., Dong, H., Zhao, Z., Zhang, T., Wei, X., & Cai, X. (2009). The study of operating variables in soil washing with EDTA. *Environmental Pollution*, 157(1), 229-236.

LIST OF PUBLICATIONS AND PAPER PRESENTED

Article

Ng, Y.S., Sen Gupta, B., & Hashim, M.A. (2014). Stability and performance enhancements of Electrokinetic-Fenton soil remediation, *Reviews in Environmental Science and Bio/Technology 13* (3), 251-263.

Ng, Y.S., Sen Gupta, B., & Hashim, M.A. (2014). Performance evaluation of two-stage electrokinetic washing as soil remediation method for lead removal using different wash solutions, *Electrochimica Acta 147*, 9-18.

Ng, Y. S., Sen Gupta, B., & Hashim, M.A. (2015). Effects of operating parameters on the performance of washing–electrokinetic two stage process as soil remediation method for lead removal, *Separation and Purification Technology 156(2)*, 403-413.

Ng, Y.S., Sen Gupta, B., & Hashim, M.A. (2016). Performance evaluation of natural iron-rich sandy soil as a low-cost adsorbent for removal of lead from water, *Desalination and Water Treatment 57(11)*, 5013-5024.

Ng, Y. S., Sen Gupta, B., & Hashim, M.A. (2016). Remediation of Pb/Cr cocontaminated soil using electrokinetic process and approaching electrode technique, *Environmental Science and Pollution Research 23(1)*, 546-555.

Conference Proceeding

Ng, Y.S., Hashim, M.A., & Sen Gupta, B. (2014). "Effects of Wetting Agents and Approaching Anode on Lead Migration in Electrokinetic Soil Remediation". In T. Gong (Ed). *International Proceeding of Chemical, Biological & Environmental Engineering: Chemical Engineering and Applications V: Proceeding of the* 5th *International Conference on Chemical Engineering and Applications, Taipei* (pp. 44-48). Singapore, IACSIT Press.

MINI-REVIEW

Stability and performance enhancements of Electrokinetic-Fenton soil remediation

Y. S. Ng · B. Sen Gupta · M. A. Hashim

Published online: 23 February 2014 © Springer Science+Business Media Dordrecht 2014

Abstract Electrokinetic process is a potential in situ soil remediation process which transports the contaminants via electromigration and electroosmosis. For organic compounds contaminated soil, Fenton's reagent is utilized as a flushing agent in electrokinetic process (Electrokinetic-Fenton) so that removal of organic contaminants could be achieved by in situ oxidation/destruction. However, this process is not applied widely in industries as the stability issue for Fenton's reagent is the main drawback. The aim of this mini review is to summarize the developments of Electrokinetic-Fenton process on enhancing the stability of Fenton's reagent and process efficiency in past decades. Generally, the enhancements are conducted via four paths: (1) chemical stabilization to delay H₂O₂ decomposition, (2) increase of oxidant availability by monitoring injection method for Fenton's reagent, (3) electrodes operation and iron catalysts and (4) operating conditions such as voltage gradient, electrolytes and H2O2 concentration. In addition, the types of soils and contaminants are also

Y. S. Ng · M. A. Hashim (⊡) Department of Chemical Engineering, Faculty of Engineering, University of Malaya, 50603 Kuala Lumpur, Malaysia e-mail: alihashim@um.edu.my

B. Sen Gupta

School of Planning, Architecture and Civil Engineering, Queen's University Belfast, David Keir Building, Belfast BT9 5AO, UK showing significant effect as the soil with low acid buffering capacity, adequate iron concentration, low organic matter content and low aromatic ring organic contaminants generally gives better efficiency.

Keywords Electrokinetic-Fenton · H₂O₂ stability enhancement · Chemical stabilization · Oxidant delivery · Electrodes operation and iron catalysts

1 Introduction

Electrokinetic soil treatment is a remediation process that operates under a low direct current gradient in soil, as opposed to a hydraulic pressure gradient, which promotes the migration of water and various contaminants under coulombic forces (Shenbagavalli and Mahimairaja 2010). This process produces an electrical gradient that acts as a driving force for the transport of various pollutants in soil, which makes it suitable for both saturated and unsaturated soil (Acar et al. 1995; Yang and Liu 2001; Shenbagavalli and Mahimairaja 2010; Tsai et al. 2010). The main feature of the electrokinetic process is in situ soil contaminant removal, and it may be considered as a preferable treatment option compared to some conventional exsitu treatment processes, such as solidification/stabilization and soil washing. Electrokinetic soil treatment has been studied for the removal of a wide range of contaminants, such as heavy metals, phenols,

D Springer

Electrochimica Acta 147 (2014) 9-18

Contents lists available at ScienceDirect



Electrochimica Acta

journal homepage: www.elsevier.com/locate/electacta

Performance Evaluation of Two-Stage Electrokinetic Washing as Soil Remediation Method for Lead Removal using Different Wash Solutions



Y.S. Ng^a, B. Sen Gupta^{b,1}, M.A. Hashim^{a,*}

^a Department of Chemical Engineering, University of Malaya, 50603 Kuala Lumpur, Malaysia
^b School of Planning, Architecture and Civil Engineering, Queen's University Belfast, David Keir Building, Belfast BT9 3AG, UK

ARTICLE INFO

Article history: Received 28 May 2014 Received in revised form 31 July 2014 Accepted 28 August 2014 Available online 18 September 2014

Reywords: Two-stage electrokinetic washing soil treatment lead wash solutions transport mechanisms

ABSTRACT

The study explores the application of a two-stage electrokinetic washing system on remediation of lead (Pb) contaminated soil. The process involved an initial soil washing, followed by an electrokinetic process. The use of electrokinetic process in soil washing not only provided additional driving force for transporting the desorbed Pb away from the soil but also reduced the high usage of wash solution. In this study, the effect of NaNO₃, HNO₃, citric acid and EDTA as wash solutions on two-stage electrokinetic washing system were evaluated. The results revealed that a two-stage electrokinetic washing process enhanced Pb removal efficiency by 2.52-9.08% and 4.98-20.45% in comparison to a normal electrokinetic process and normal washing process, respectively, Low pH and adequate current were the most important criteria in the removal process as they provided superior desorption and transport properties. The effect of healting by EDTA was less dominant as it delayed the removal process by forming a transport loop in anode region between Pb ion and complexes. HNO₃ was not suitable as wash solution in electrokinetic washing in spite of offering highest removal efficiency as it caused pH fluctuation in the cathode chamber, corroded graphite anode and showed high power consumption. In contrast, citric acid not only yielded high Pb removal efficiency with low power consumption but also maintained a low soil: solution ratio of 1 g: <1 mL, stable pH and electrode integrity. Possible transport mechanisms for Pb under each wash solution are also discussed in this work.

© 2014 Elsevier Ltd. All rights reserved.

1. Introduction

Lead (Pb) has been identified as one of the top six toxic threats in the world [1] as it can cause acute and chronic illnesses by damaging central and peripheral nervous systems, cardiovascular and reproductive systems as well as gastrointestinal and urinary tracts when it is inhaled and ingested [2,3]. As a result of its extensive use, humans are exposed to Pb via several pathways. One of the pathways is soil contamination. Pb concentration as high as 751.98-138,000 mg/kg was reported in shooting range soil, dumpsite for lead-acid battery as well as manufacturing and mining regions over the world [1,4-6]. Therefore, a proper treatment must be conducted for these soils.

http://dx.doi.org/10.1016/j.electacta.2014.08.124 0013-4686/© 2014 Elsevier Ltd. All rights reserved. According to Dermont et al. [7], soil washing is one of the *ex situ* remediation methods for treating Pb and heavy metals contaminated soil especially for higher permeability soil that has lower silt and clay content [8]. There has been several studies on the treatment of Pb contaminated soil using different washing agents such as surfactants [9] acids [10] and chelating agents [11–13] and positive results have been reported. However, it is worth noting that one of the disadvantages of soil washing is the occurrence of spent chemicals in the wash solution and this increases the cost for treatment [7]. Moreover, a high ratio for soil: solution in the system of 1 g:3.33-20 mL [9,11,12] is another disadvantage for soil washing as the amount of spent wash solution to be treated is large.

In contrast, electrokinetic soil remediation could reduce the amount of wash solution needed in remediating heavy metals contaminated soil. Electrokinetic process is a potential soil treatment method that uses electricity as the driving force for ion transport. The induction of low intensity direct current through the soil augments the transport of water and contaminants via two major mechanisms, namely electromigration for ion transport and electroosmosis for pore water/neutral compounds transport [14–16]. Electromigration transports the desorbed heavy metals in the wash

Corresponding author. Department of Chemical Engineering, Faculty of Engineering, University of Malaya, 50603 Kuala Lumpur, Malaysia. Tel.: +603-79675296; fax: +603-79675319.

E-mail address: alihashim@um.edu.my (M.A. Hashim).

¹ School of the Built Environment, Heriot-Watt University, Edinburgh Campus, EH14 4AS Scotland

Separation and Purification Technology 156 (2015) 403-413

Contents lists available at ScienceDirect



Separation and Purification Technology



Effects of operating parameters on the performance of washing-electrokinetic two stage process as soil remediation method for lead removal



Yee-Sern Ng^a, Bhaskar Sen Gupta^{b,1}, Mohd Ali Hashim^{a,a}

Department of Chemical Engineering, University of Malaya, 50603 Kuala Lumpur, Malaysia

*Water Academy, School of the Built Environment, Heriot-Watt University, Edinburgh Campus, EH14 4AS Scotland, UK

ARTICLE INFO

Article history: Received 20 March 2013 Received in ravired form 12 Oc

Received in revised form 12 October 2015 Accepted 13 October 2015 Available online 17 October 2015

Reywords: Soil remediation Lead Washing-electrokinetic Response surface methodology

ABSTRACT

The application of washing-electrokinetic two stage process on remediation of lead (Pb) contaminated soil was investigated. The process consisted of an initial soil washing, followed by an electrokinetic process. The use of electrokinetic process in soil washing could provide additional driving force for transporting the desorbed Pb away from the soil even in the absence of pore flow. Thus, high usage of wash solution may be mitigated. In this study, the effect of operating conditions such as electric potential dif-ference, wash solution concentration and initial Pb concentration on the feasibility of washing-electrokinetic two stage process on Pb removal was investigated using response surface methodology based on Box-Behnken design. The wash solution used was citric acid and three main aspects were examined, namely: (i) removal efficiency, (ii) effluent generated, and (iii) power consumption. The results revealed that the increase in electric potential difference and wash solution concentration generally enhanced Pb removal efficiency and the interactions of these parameters were significantly positive. However, undesirable high effluent generation and power consumption were also caused by these increments. Optimisation study revealed that 84.14% removal efficiency with zero effluent generation and a power consumption of 2.27 kW h/kg Pb removed could be achieved at 7.58 V and 0.057 M citric acid concentration. In comparison with normal soil washing, washing-electrokinetic two stage process showed an enhancement in removal efficiency by \$10% via electromigration under optimum conditions using similar solution: soil ratio of <0.8 mL:1 g soil. The study reveals that incorporation of electrokinetic process in soil washing is feasible as it not only enhances Pb removal efficiency at minimum wash solution usage with respect to normal soil washing, but also provides in situ Pb recovery in cathode chamber via electrodeposition.

© 2015 Elsevier B.V. All rights reserved.

1. Introduction

Lead (Pb) is a highly toxic substance to living organisms. Its ability to bio-accumulate often causes acute and chronic illnesses which damage human body systems when it is inhaled and ingested [1,2]. Pb has been extensively used as a raw material in manufacturing processes such as ammunition, batteries, bearings, plumbing, ceramic glazes, weights, caulks, dyes, pigments and pesticides [2,3]. In order to support huge demand for Pb in the world, global Pb production as high as 10,654,000 tonnes was reported in

http://dx.doi.org/10.1016/j.seppur.2015.10.029 1383-5806/@ 2015 Elsevier B.V. All rights reserved. 2012. As a result of its huge production and usage, humans can be easily exposed to Pb via variety pathways. Soil contamination is one of the pathways for Pb exposure especially from industrial lands with the activities like battery manufacturing, gould casing, scrap Pb handling [4] as well as Pb smelting and mining [1,2]. High Pb concentration in the range of 751.98–138,000 mg/kg was reported in the soil from these industries [3,5–8]. Thus, a proper treatment to these soils is necessary.

Soil washing has been reported as one of the effective soil remediation methods for removing Pb and heavy metals [9]. However, it is worth noting that soil washing requires high amount of wash solution for effective treatment. A high solution: soil ratio of 3.33–20 mL:1 g [10–12] shows the disadvantage of applying soil washing as the amount of spent wash solution that requires post treatment is large. In contrast, electrokinetic process could reduce the amount of wash solution needed in soil remediation.

^{*} Corresponding author at: Department of Chemical Engineering, Faculty of Engineering, University of Malaya, 30003 Kuala Lumpur, Malaysia.

E-mail address: althashim@um.edu.my (M.A. Hashim). ¹ Previous affiliation: School of Planning, Architecture and Civil Engineering, Queen's University Belfast, David Keir Building, Belfast BT9 5AG, UK.

Desalination and Water Treatment www.deswater.com doi: 10.1080/19443994.2014.999711 57 (2016) 5013–5024 March

Taylor & Francis

Performance evaluation of natural iron-rich sandy soil as a low-cost adsorbent for removal of lead from water

Yee Sern Ng^a, Bhaskar Sen Gupta^{b,1}, Mohd Ali Hashim^{a,*}

^aFaculty of Engineering, Department of Chemical Engineering, University of Malaya, 50603 Kuala Lumpur, Malaysia, Tel. +60 12 6453952; email: yeesern86@yahoo.com (Y.S. Ng), Tel. +60 3 79675296; Fax: +60 3 79675319; email: alihashim@un.edu.my (M.A. Hashim)

^bSchool of the Built Environment, Heriot-Watt University, Edinburgh Campus, EH14 4AS Edinburgh, Scotland, UK, Tel. +44(0)131 451 8171; email: b.sengupta.hw.ac.uk (B.S. Gupta)

Received 6 May 2014; Accepted 13 December 2014

ABSTRACT

Technical feasibility of natural iron-rich sandy soil as a low-cost adsorbent for removal of lead from water was investigated . The soil, which had an iron content of 3,719 mg/kg, was collected from Hulu Langat, Malaysia, and was used for adsorption studies without any surface modification through chemical treatment. The effects of pH, solution: soil ratio and initial lead concentration on the adsorption efficiency were studied using response surface methodology based on Box-Behnken experimental design. The results showed that pH of the solution had the highest impact on the adsorption efficiency whereby adsorption efficiency of 97% could be achieved at pH 3,5–5. The experimental data were also checked for compliance with different kinetic models and adsorption isotherms. The adsorption process was found to be rapid monolayer chemisorption with adsorption capacity of 0.9–1.0 mg/g, as it fitted Langmuir isotherm and followed pseudo-second-order kinetic model.

Keywords: Natural iron-rich sandy soil; Lead; Adsorption; Adsorption kinetics; Adsorption isotherm

1. Introduction

Lead is reported as a highly toxic material whereby the maximum allowable concentration in water is 0.1 ppm for standard A (effluent that is discharged upstream of a water supply intake) and 0.5 ppm for standard B (effluent that is discharged downstream of a water supply intake) in Malaysia [1]. Lead often contaminates water bodies as well as soil environment via different sources such as fertilisers, pesticides, metal mining, milling and smelting processes [2,3]. Inhalation and ingestion of lead can cause poisoning and even death to human as it accumulates in the organs. Lead can cause serious health effects such as loss of memory/brain damage, nausea, insomnia, anorexia and weakness of joints. Moreover, it may also inhibit the synthesis of haemoglobin, causing dysfunction on kidneys, joints and systems such as reproductive system, cardiovascular system, central

^{*}Corresponding author.

¹Former affiliation: School of Planning, Architecture and Civil Engineering, Queen's University Belfast, David Keir Building, Belfast BT9 5AG, UK

^{1944-3994/1944-3986 © 2014} Balaban Desalination Publications. All rights reserved.

RESEARCH ARTICLE



Remediation of Pb/Cr co-contaminated soil using electrokinetic process and approaching electrode technique

Yee-Sern Ng1 · Bhaskar Sen Gupta2.3 · Mohd Ali Hashim1

Received: 16 May 2015 / Accepted: 20 August 2015 / Published online: 2 September 2015 © Springer-Verlag Berlin Heidelberg 2015

Abstract Electrokinetic process has emerged as an important tool for remediating heavy metal-contaminated soil. The process can concentrate heavy metals into smaller soil volume even in the absence of hydraulic flow. This makes it an attractive soil pre-treatment method before other remediation techniques are applied such that the chemical consumption in the latter stage can be reduced. The present study evaluates the feasibility of electrokinetic process in concentrating lead (Pb) and chromium (Cr) in a co-contaminated soil using different types of wetting agents, namely 0.01 M NaNO3, 0.1 M citric acid and 0.1 M EDTA. The data obtained showed that NaNO3 and citric acid resulted in poor Pb electromigration in this study. As for Cr migration, these agents were also found to give lower electromigration rate especially at low pH region as a result of Cr(VI) adsorption and possible reduction into Cr(III). In contrast, EDTA emerged as the best wetting agent in this study as it formed water-soluble anionic complexes with both Pb and Cr. This provided effective one-way electromigration towards the anode for both ions, and they were accumulated into smaller soil volume with an enrichment ratio of 1.55-1.82. A further study on the application of approaching cathode in EDTA test showed that soil alkalisation was achieved, but this did not provide significant

Responsible editor: Philippe Garrigues

Mohd Ali Hashim alihashim@um.edu.my

- Department of Chemical Engineering, Faculty of Engineering, University of Malaya, 50603 Kuala Lumpur, Malaysia
- ² Water Academy, School of the Built Environment, Heriot-Watt University, Edinburgh Campus, EH14 4AS, Scotland, UK
- ³ School of Planning, Architecture and Civil Engineering, Queen's University Belfast, David Keir Building, Belfast BT9 5AG, UK

enhancement on electromigration for Pb and Cr. Nevertheless, the power consumption for electrokinetic process was decreased by 22.5 %.

Keywords Electrokinetic soil remediation · Approaching cathode · Lead · Chromium · EDTA

Introduction

Electrokinetic process is a potential soil remediation method. This process is carried out by the introduction of low magnitude direct current across the soil via electrodes that are in contact with the soil. During the process, electrolysis occurs in both anode and cathode chambers, and H+ and OH are generated, respectively. The potential difference between the electrodes causes the contaminants/ions in the soil to migrate towards their respective chambers via two mechanisms, namely (i) electromigration for ions transport and (ii) electroosmosis for neutral compound transport (Acar and Alshawabkeh 1993; Acar et al. 1995). Among these mechanisms, electromigration not only transports charged ions in the soil but also transports H⁺ and OH⁻ generated from electrolysis through the soil. As the migration speed of H+ is about 1.8 times faster than of OH" (Acar and Alshawabkeh 1993; Acar et al. 1995; Kim et al. 2011), soil acidification may occur, and this generally enhances heavy metals desorption from the soil. It makes electrokinetic process an effective method in remediating heavy metals polluted soil.

Electromigration is generally free from limitations of hydraulic gradient and pore flow (Acar et al. 1995). Thus, electrokinetic process can concentrate heavy metals into a smaller soil volume even in the absence of hydraulic flow. This ability provides an attractive soil pre-treatment option before other remediation methods are applied such that the reduction in

Effects of Wetting Agents and Approaching Anode on Lead Migration in Electrokinetic Soil Remediation

Yee-Sern Ng 1+, Mohd Ali Hashim 1 and Bhaskar Sen Gupta 2

¹ Department of Chemical Engineering, Faculty of Engineering, University of Malaya, Malaysia ² School of Planning, Architecture and Civil Engineering, Queen's University Belfast, United Kingdom

Abstract. Approaching anode is one of the enhancement techniques in electrokinetic soil remediation. This technique is reported to give promising migration for heavy metals under shorter treatment time and at lower cost in comparison to normal fixed anode system. In the present study, the effectiveness of fixed anode and approaching anode techniques in electrokinetic soil remediation for lead migration under different types of wetting agents (0.01M NaNO₃ and 0.1M citric acid) was investigated. The study showed that the use of citric acid enhanced lead migration in comparison to NaNO₃. For NaNO₃ tests, lead was found to accumulate in the middle of soil section due to high soil pH, which favoured lead adsorption and precipitation. Approaching anode reduced this effect by compressed high soil pH region and enhanced lead migration at cathode region. The approaching anode only showed technical advantages when NaNO₃ was used whereas the enhancement in citric acid test was insignificant. Nevertheless, this technique reduced electricity usage by 18-20% for both wetting agents.

Keywords: Lead, electrokinetic soil remediation, wetting agent, approaching anode

1. Introduction

McCartor and Becker [1] and Wuana and Okieimen [2] reported that lead (Pb) has been extensively used in manufacturing processes such as ammunition, batteries, plumbing, dyes, pigments and pesticides and this causes potential environmental pollution and human exposure. Soil contamination is one of the pathways for Pb exposure to humans especially from Pb smelting and mining areas [2], [3] as well as from industrial land used for battery and scrap Pb handlings [4]. As Pb is highly toxic and can cause health hazard to living organisms, a proper treatment to these soils is required.

Electrokinetic process is a soil remediation method which uses electricity as driving force for contaminant transport in the soil. This process is initiated by applying low magnitude direct current from the electrodes injected into the soil. During electrokinetic process, electrolysis occurs in both anode and cathode chambers which produce H⁺ and OH, respectively. Due to the potential difference between the electrodes, H⁺ and OH transport to the respective electrode, and these are named as acid front and base front. These mechanisms create two transport forces in the soil: i) electromigration for ions transport, and ii) electroosmosis for water/neutral compounds transport [5], [6]. The forces are responsible for contaminants removal from the soil, whereby the contaminants are concentrated in the electrolyte chambers or enriched in a smaller soil volume. To date, electrokinetic process has been proven as successful method in treating Pb contaminated soil using different types of desorption/flushing agents to enhance Pb mobility via better desorption condition [7], [8] or forming stable water soluble complexes [7]-[12].

However, it is worth noting that the lengthy remediation time and additional cost in power consumption in using electrokinetic process often restricts its wide application [13]-[15]. Besides that, simultaneous acid and base fronts create a region where H⁺ and OH⁻ meet and this causes accumulation/focusing of heavy

^{*} Corresponding author. Tel.: + 603-79676867; fax: +603-79675319.

E-mail address: yeesem86@yahoo.com.

APPENDICES

APPENDIX A: Chemicals used in the analytical study

Acetic acid, HOAc

Ammonium acetate, NH₄OAc

Ammonium chloride, NH₄Cl

Ammonium hydroxide, NH₄OH

Ammonium Oxalate, (NH₄)₂C₂O₄

Bromocresol green-methyl red mixed indicator

Dipotassium phosphate, K₂HPO₄

Hydrochloric acid, HCl

Hydrogen peroxide, H₂O₂

Hydroxylamine hydrochloride, NH₂OH.HCl

Isopropyl alcohol, C₃H₇OH

Magnesium chloride, MgCl₂

Magnesium oxide, MgO

Monopotassium phosphate, KH₂PO₄

Nitric acid, HNO₃

Perboric acid, H₃BO₄

Silver nitrate, AgNO3

Sodium acetate, NaOAc

Sodium carbonate, Na₂CO₃

Sodium chloride, NaCl

Sodium hydroxide, NaOH

Sulfuric acid, H₂SO₄

APPENDIX B: Procedures for soil characterisation study

Appendix B1: Procedure for cation exchange capacity (CEC) determination in soil

Cation exchange capacity (CEC) for the soil sample was determined using ammonium acetate method.

- 3 g of air-dried soil sample was placed in a 250 mL Erlenmeyer flask. 100 mL of 1 N NH₄OAc (pH 7.0) solution was added into the flask. The flask was shaked thoroughly by hand and the mixture was allowed to stand overnight (cover the flask mouth with parafilm).
- ii) The soil was filtered with light suction using a Buchner funnel and filter paper into a clean Buchner flask (25 mL at a time).
- iii) The soil was filtered with additional 100 mL of 1 N NH₄OAc (pH 7.0). Check for Ca²⁺: The vacuum was closed and funnel was lifted out of the flask. Three drops of filtrate was transferred into a test tube. Three drops of 1 N NH₄Cl, three drops of 1:1 NH₄OH, and three drops of 10% ammonium oxalate were added. No precipitate indicates the completion of filtering.
- iv) The soil was filtered with light suction using 200 mL 1 N NH₄Cl, followed by 100 mL 0.25 N NH₄Cl.
- v) The soil was washed with 150-200 mL of isopropyl alcohol (25 mL at a time). Check for Cl⁻: Funnel was removed as in step 3, 10 drops of filtrate and 10 drops of 0.1N AgNO₃ was mixed into a clean test tube. A precipitate (AgCl) indicates the presence of chloride. When the chloride is no longer present, the collection flask was emptied and cleaned.
- vi) The soil was further filtered with 300 mL of 10% NaCl in 5-6 portions. The filtrate was saved for CEC analysis.

Microkjeldahl Distillation of Ammonium for CEC Experiment

- i) Rheostat to the heater was turned on and the water was allowed to boil
- ii) Steam distillation apparatus was prepared by opening the lower stopcock on the steam-bypass assembly and closing the upper stopcock, which connected to the distillation head. These should be the positions of these stopcocks while the water heats.
- 5 mL of H₃BO₃ indicator solution was added into 100 mL beaker marked for 40 mL, and the beaker was positioned under the condenser of the distillation apparatus so that the up of the condenser is in contact with the side of the beaker.
- iv) Kjedahl flasks which contained 20 mL NaCl filtrate and a spoon of MgO was attached to the distillation apparatus. The funnel was sealed with the peg stopper, and distillation was commenced immediately by opening the upper stopcock on the steam-bypass assembly and closing the lower stopcock.
- v) When the volume of distillation reached 40 mL, the tip of the condenser was rinsed and the distillation was stopped.

vi) 10 drops of bromocresol green-methyl red mixed indicator was added and NH_4^+ -N content in the distillate was determined by titration using 0.001 N H_2SO_4 . At the end point, the colour changed from green to a faint pink.

CEC was calculated based on the equation as shown below:

CEC (meq/100g soil) = V x 0.001 N x 300 mL/20 mL x 100 g/m_s

Where:

V = volume of 0.001 N H₂SO₄ spent for titration in mL $m_s =$ weight of the soil sample used, g

university chalad

Appendix B2: Procedures for specific gravity determination in soil

ASTM D 854 was used to determine the specific gravity for the soil sample.

- i) The weight of empty clean and dry pycnometer was recorded, W_p.
- ii) 10 g of dried soil sample was placed in a pycnometer. The weight of the dry soil and pycnometer was recorded, W_{sp}.
- iii) Distilled water was filled to about 50-75% of the pycnometer. The sample was soaked for 10 minutes.
- iv) Partial vacuum was applied for 10 minutes to remove the entrapped air.
- v) The vacuum was stopped and the pycnometer was removed from the vacuum line.
- vi) The pycnometer was filled with distilled water to the mark. The exterior surface of the pycnometer was cleaned with a clean and dry cloth. The weight of the pycnometer and contents, W_b was recorded.
- vii) The pycnometer was emptied and cleaned. The, distilled water was filled to the mark. The exterior surface of the pycnometer was cleaned with a dry and clean cloth. The weight of the pycnometer and distilled water, W_a was recorded.

Specific gravity for the soil was calculated using the equation as shown below:

Specific Gravity, Gs =
$$\frac{W_0}{W_0 + (W_a - W_b)}$$

where W_0 is the weight of dry soil sample, W_{sp} -W_{\text{p}}

Appendix B3: Procedures for organic matter determination in soil

Soil organic matter was estimated using Loss of Weight on Ignition Method (LOI).

- i. A crucible was weighed to 3 decimal places (W1).
- ii. A mass of 5-10 g of dried and ground soil sample was weighed into crucible.
- iii. The crucible with soil samples was dried and heated for 2 h at 150 °C.
- iv. The crucible with samples was removed and weighed to 3 decimal places (W2).
- v. The crucible with samples was further heated at 360 °C for 2 h (After temperature reaches 360 °C).
- vi. The crucible and samples was removed and was cooled to 150 °C.
- vii. The sample was weighed to 3 decimal places (W3).

Loss of weight on ignition (LOI) was determined using the equation as shown below:

LOI % (OM) = $[(W2-W3)/(W2-W1)] \times 100$

Appendix B4: Procedures for Tessier's method

The study on Pb speciation after the adsorption was conducted using Tessier's method. The general steps involved in this method are as summarized in Figure B4.1. The detailed procedures are elaborated by Tessier et al. (1979), Gworek and Mocek (2003) and Venkateswaran et al. (2007).



Figure B4.1: Schematic procedure for Pb speciation study using Tessier's Method

Appendix B5: Procedures for soil voidage

The voidage in the soil column in two-stage electrokinetic washing after soil packing was calculated based on the following equation:

Soil voidage = $\frac{A - B}{A}$

where:

A = Volume of the column, $\frac{\pi}{4} (D_{column})^2 L_{column}$, and

 $B = Volume occupied by soil particles, \frac{Mass of soil used}{Density of the soil particle}$

Source	Sum of Squares	DF	Mean Square	F value	Prob > F	
Model	20853.96	11	1895.81	666.81	< 0.0001	significant
А	6304.36	1	6304.36	2217.42	< 0.0001	
В	748.02	1	748.02	263.10	< 0.0001	
С	0.12	1	0.12	0.04	0.8438	
A^2	2465.09	1	2465.09	867.04	< 0.0001	
B^2	14.08	1	14.08	4.95	0.0766	
C^2	609.73	1	609.73	214.46	< 0.0001	
AB	2670.31	1	2670.31	939.22	< 0.0001	
AC	1288.81	1	1288.81	453.31	< 0.0001	
A^2B	295.85	1	295.85	104.06	0.0002	
A ² C	657.03	1	657.03	231.10	< 0.0001	
AB^2	384.34	1	384.34	135.18	< 0.0001	
Residual	14.22	5	2.84			
Lack of Fit	0.12	1	0.12	0.04	0.8612	not significan
Pure Error	14.09	4	3.52			
Cor Total	20868.17	16				
Std. Dev.	1.69		R-Squared	0.9993		
Mean	19.66		Adj R-Squared	0.9978		
C.V.	8.58		Pred R-Squared			
PRESS			Adeq Precision	81.51		

Appendix C1: ANOVA for effluent generation

Source	Sum of Squares	DF	Mean Square	F value	Prob > F	
Model	25.478	9	2.831	2958.52	< 0.0001	significant
А	18.050	1	18.050	18863.82	< 0.0001	
В	3.372	1	3.372	3524.30	< 0.0001	
С	0.710	1	0.710	741.87	< 0.0001	
A^2	1.553	1	1.553	1623.47	< 0.0001	
B^2	0.650	1	0.650	678.93	< 0.0001	
C^2	0.002	1	0.002	1.99	0.2015	
AB	0.002	1	0.002	2.39	0.1663	
AC	0.067	1	0.067	69.87	< 0.0001	
$A^{2}C$	0.009	1	0.009	9.59	0.0174	
Residual	0.007	7	0.001			
Lack of Fit	0.004	3	0.0011	1.88	0.2734	not significant
Pure Error	0.003	4	0.001			
Cor Total	25.485	16				×
Std. Dev.	0.03		R-Squared	0.9997		
Mean	2.78		Adj R-Squared	0.9994		
C.V.	1.11		Pred R-Squared	0.9973		
PRESS	0.07		Adeq Precision	187.61		

Appendix C2: ANOVA for power consumption

194